DESIGN CONTRACT

THIS CONTRACT is made and entered into by and between the CITY OF FRESNO, a California municipal corporation (City), and INTEGRATED COMMUNITY DEVELOPMENT, a limited liability company (Design-Build Entity), as follows:

RECITALS

WHEREAS, the City desires to obtain professional design services for the design of plans and general construction contract documents for the Senior Activity Center at 4343 North Blackstone Avenue (Project); and

WHEREAS, the City desires to utilize a Progressive Design Build delivery method to design and build the Project; and

WHEREAS, the selected design-build team composed of [DESIGN-BUILDER NAME] and it's sub-consultants, including architects and engineers from [SUB-CONSULTANTS NAME], shall provide design services and hereby represents that it desires to and is professionally and legally capable of performing the services call for by this Agreement; and

WHEREAS, this Agreement will be administered for the City by its Department of Public Works Director (Director) or designee.

- <u>Contract Documents</u>. The "Request for Qualifications", "Specifications", the "DBIA Progressive Design Build Agreement", as modified, and the "DBIA Standard Form of General Conditions of Contract Between Owner and Design-Builder", as modified, for design services related to a Senior Activity Center at 4343 North Blackstone Avenue, copies of which are annexed hereto as part of Exhibit 1, together with all the documents specifically referred to in said annexed documents, are hereby incorporated into and made part of this Contract, and shall be known as the Contract Documents.
- 2. <u>Price</u>. For the monetary consideration of \$634,117, as set forth in **Exhibit 1**, Design-Builder promises and agrees to perform or cause to be performed, in a good and workmanlike manner, and to the satisfaction of the City, and in strict accordance with the Specifications, all of the work as set forth in the Contract Documents.
- 3. <u>Payment</u>. The City accepts the Design-Builder's Proposal as stated and agrees to pay the consideration stated, at the times, in the amounts, and under the conditions specified in the Contract Documents. The Design-Builder agrees to accept electronic payment from the City.
- Scope of Services. Design-Builder shall perform the services described herein and in Exhibit 1 to complete the design of the Project more fully described in Exhibit 1, and this shall include all work incidental to, or necessary to perform, such design services even though not specifically described in Exhibit 1. By entry into this Agreement and upon City's issuance of a written "Notice to Proceed", City

contracts for the services in Phase One for design. Design-Builder shall not perform any other Phase of the Agreement, and this Agreement shall not be a contract for any other Phase, until further performance is authorized by City's issuance of a written "Notice to Proceed" for such additional phase(s). It shall, however, remain Design-Builder's offer to perform all remaining phases described herein. In the event Design-Builder performs without City's prior written authorization, Design-Builder will not be entitled to compensation for such services.

The DBIA Design Build Agreement (Agreement), as modified, contains additional terms and conditions related to this Contract. In the event of a conflict between this Contract and the Agreement, this Contract shall take precedence. However, if this Contract does not address a legal issue for this Project, the parties shall be governed by the provisions of the Agreement.

[SIGNATURES FOLLOW ON THE NEXT PAGE]

IN WITNESS WHEREOF, the parties have executed this Contract on the day and year here below written, of which the date of execution by the City shall be subsequent to that of the Design-Builder's, and this Contract shall be binding and effective upon execution by both parties.

CITY OF FRESNO, A California municipal corporation

By: 6 for Scott Mozier

Public Works Director Public Works Department

Dated: 07/20/2023

APPROVED AS TO FORM: ANDREW JANZ

City Attorney Bv: Brandon M. Collet

Supervising Deputy City Attorney City Attorney's Office

Dated:

ATTEST: TODD STERMER, CMC City Clerk

By: _____ Deputy

Dated:

Addresses: CITY: City of Fresno Attention: Mike Mooneyham, PE, Professional Engineer 2600 Fresno Street Fresno, CA 93721 Phone: (559) 621-8623 E-mail: mike.mooneyham@fresno.gov

20-2023

INTEGRATED COMMU DEVELOPMEN A limited liability company By Name: Title: (If corporation or LLC., Board Chair, Pres. or Vice Pres.) By: Name: Title:

(If corporation of LC., CFO, Treasurer, Secretary or Assistant Secretary)

REVIEWED BY: Mike Mooneyham, PE Professional Engineer Public Works Department

DESIGN-BUILDER: INTEGRATED COMMUNITY DEVELOMPENT Attention: Jake Lingo, Senior Vice President 20750 Ventura Boulevard, Suite 155 Woodland Hills, CA 91364 Phone: 818-974-2966 E-mail: jakelingo@icdemail.com

Attachment:

Exhibit 1 – Contract Documents including Scope of Services

Exhibit A – Owner's Project Criteria

Exhibit B – Scope of Services

Exhibit C – Federal Requirements

Exhibit D – (Intentionally Omitted)

Federal Wage Rates

Exhibit E – Disclosure of Conflict of Interest

General Conditions

Progressive Design-Build Agreement

EXHIBIT 1

Document No. 544 Second Edition, 2022

© Design-Build Institute of America Washington, D.C.





Progressive Design-Build Agreement

This document has important legal consequences. Consultation with an attorney is recommended with respect to its completion or modification.

This AGREEMENT is made as of the the year of <u>2023</u> , by identified below:	20 th and between the following parties,	day of for services in co	<u>July</u> nnection with the	in e Project
OWNER:				

City of Fresno 2600 Fresno Street Fresno, CA 93721

DESIGN-BUILDER: Integrated Community Development 20750 Ventura Boulevard, Suite 155 Woodland Hills, CA 91364

PROJECT: Senior Activity Center 4343 North Blackstone Avenue

In consideration of the mutual covenants and obligations contained herein, Owner and Design-Builder agree as set forth herein.

Article 1

General

- 1.1 Duty to Cooperate. Owner and Design-Builder commit at all times to cooperate fully with each other, and proceed on the basis of trust and good faith to permit each party to realize the benefits afforded under this Agreement
- 1.2 Definitions. Terms, words, and phrases used in this Agreement shall have the meanings given them in DBIA Document No. 535, Standard Form of General Conditions of Contract Between Owner and Design-Builder (2022 Edition) ("General Conditions of Contract")
- 1.3 Design Services. Design-Builder shall, consistent with applicable state licensing laws, provide design services, including architectural, engineering, and other design professional services required by this Agreement. Such design service es shall be provided through qualified, licensed design professionals who are either (i) employed by Design-Builder, or (ii) procured by Design-Builder from independent sources. Nothing in this Agreement is intended to create any legal or contractual relationship between Owner and any independent design professional.

<u>Article 2</u>

Design-Builder's Services and Responsibilities

2.1 General Services.

2.1.1 Owner shall provide Design-Builder with Owner's Project Criteria describing Owner's program requirements and objectives for the Project as set forth in Exhibit A. Owner's Project Criteria shall include Owner's use, space, price, time, site, performance, and expandability requirements. Owner's Project Criteria may include conceptual documents, design specifications, design performance specifications, and other technical materials and requirements prepared by or for Owner.

2.1.2 If Owner's Project Criteria have not been developed prior to the execution of this Agreement, Design-Builder will assist Owner in developing Owner's Project Criteria, with such service deemed to be an additional service for which additional compensation shall be paid by Owner to Design-Builder. If Owner has developed Owner's Project Criteria prior to executing this Agreement, Design-Builder shall review and prepare a written evaluation of such criteria, including recommendations to Owner for different and innovative approaches to the design and construction of the Project. The parties shall meet to discuss Design-Builder's written evaluation of Owner's Project Criteria and agree upon what revisions, if any, should be made to such criteria.

2.2 Phased Services.

2.2.1 Phase 1 Services. Design-Builder shall perform the services of design, pricing, and other services for the Project based on Owner's Project Criteria, as may be revised in accordance with Section 2.1 hereof, as set forth in Exhibit B, Scope of Services. Design-Builder shall perform such services to the level of completion required for Design-Builder and Owner to establish the Contract Price for Phase 2, as set forth in Section 2.3 below. The Contract Price for Phase 2 shall be developed during Phase 1 on an "open-book" basis. Design-Builder's Compensation for Phase 1 Services is set forth in Section 7.1.1 herein. The level of completion required for Phase 1 Services is defined in Exhibit B, Scope of Services (either as a percentage of design completion or by defined deliverables).

2.2.2 Phase 2 Services. Design-Builder's Phase 2 services shall consist of the completion of design services for the Project, the procurement of all materials and equipment for the Project, the

performance of construction services for the Project, the start-up, testing, and commissioning of the Project, and the provision of warranty services, all as further described in the Contract Price Amendment. Upon receipt of Design-Builder's proposed Contract Price for Phase 2, Owner may proceed as set forth in Section 2.3.

2.3 Proposal. Upon completion of the Phase 1 Services and any other Basis of Design Documents upon which the parties may agree, Design-Builder shall submit a proposal to Owner (the "Proposal") for the completion of the design and construction for the Project for the Contract Price, which may be based on Lump Sum or Design-Builder's Fee and Cost of the Work with an option for a Guaranteed Maximum Price (GMP).

2.3.1 The Proposal shall include the following unless the parties mutually agree otherwise:

2.3.1.1 The Contract Price shall be in accordance with Section 7.1

2.3.1.2 The Basis of Design Documents, which may include, by way of example, Owner's Project Criteria, which are set forth in detail and are attached to the Proposal;

2.3.1.3 A list of the assumptions and clarifications made by Design-Builder in the preparation of the Proposal, which list is intended to supplement the information contained in the drawings and specifications and is specifically included as part of the Basis of Design Documents;

2.3.1.4 The Scheduled Substantial Completion Date upon which the Proposal is based, to the extent said date has not already been established under Section 6.2.1 hereof, and a schedule upon which the Scheduled Substantial Completion Date is based and a Project Schedule for the Work;

2.3.1.5 If applicable, a list of Allowance Items, Allowance Values, and a statement of their basis;

- 2.3.1.6 If applicable, a schedule of alternate prices;
- 2.3.1.7 If applicable, a schedule of unit prices;

2.3.1.8 If applicable, a statement of Additional Services which may be performed but which are not included in the Proposal, and which, if performed, shall be the basis for an increase in the Contract Price and/or Contract Time(s);

- 2.3.1.9 If applicable, a Savings provision;
- 2.3.1.10 If applicable, Performance Incentives;
- 2.3.1.11 The time limit for acceptance of the Proposal; and

2.3.1.12 An Owner's permit list, a list detailing the permits and governmental approvals that Owner will bear responsibility to obtain.

2.3.2 Review and Adjustment to Proposal.

2.3.2.1 After submission of the Proposal, Design-Builder and Owner shall meet to discuss and review the Proposal. If Owner has any comments regarding the Proposal, or finds any inconsistencies or inaccuracies in the information presented, it shall promptly give written notice to Design-Builder of such comments or findings. If appropriate, Design-Builder shall, upon receipt of Owner's notice, make appropriate adjustments to the Proposal.

2.3.2.2 Acceptance of Proposal. If Owner accepts the Proposal, as may be amended by Design-Builder, the Contract Price and its basis shall be set forth in an amendment to this Agreement, when mutually agreed between the parties (Contract Price Amendment). Once the parties have agreed upon the Contract Price and Owner has issued a Notice to Proceed with Phase 2, Design-Builder shall perform the Phase 2 Services, all as further described in the Contract Price Amendment, as it may be revised.

2.3.2.3 Failure to Accept the Proposal. If Owner rejects the Proposal, or fails to notify Design-Builder in writing on or before the date specified in the Proposal that it accepts the Proposal, the Proposal shall be deemed withdrawn and of no effect. In such event, Owner and Design-Builder shall meet and confer as to how the Project will proceed, with Owner having the following options:

- i Owner may suggest modifications to the Proposal, whereupon, if such modifications are accepted in writing by Design-Builder, the Proposal shall be deemed accepted and the parties shall proceed in accordance with Section 2.3.2.2 above;
- ii Owner may authorize Design-Builder to continue to proceed with the Work on the basis of reimbursement as provided in Section 7.1.2 hereof without a Contract Price, in which case all references in this Agreement to the Contract Price shall not be applicable; or
- iii Owner may terminate this Agreement for convenience in accordance with Article 9 hereof; provided, however, in this event, Design-Builder shall not be entitled to the payment provided for in Section 9.2 hereof.

2.3.2.4 If Owner fails to exercise any of the above options, Design-Builder shall have the right to (a) continue with the Work as if Owner had elected to proceed in accordance with Section 2.3.2.3 ii. above, and be paid by Owner accordingly, unless and until Owner notifies it in writing to stop the Work; (b) suspend performance of Work in accordance with Section 11.3.1 of the General Conditions of Contract, provided, however, that in such event Design-Builder shall not be entitled to the payment provided for in Section 9.2 hereof; or (c) may give written notice to Owner that it considers this Agreement completed. If Owner fails to exercise any of the options under Section 2.3.2.3 within ten (10) days of receipt of Design-Builder's notice, then this Agreement shall be deemed completed. If Owner terminates the relationship with Design-Builder under Section 2.3.2.3(iii), or if this Agreement is deemed completed under this paragraph, then Design-Builder shall have no further liability or obligations to Owner under this Agreement.

Article 3

Contract Documents

3.1 The Contract Documents are comprised of the following:

3.1.1 All written modifications, amendments, minor changes, and Change Orders to this Agreement issued in accordance with DBIA Document No. 535, *Standard Form of General Conditions of Contract Agreement Between Owner and Design-Builder* (2022 Edition) ("General Conditions of Contract");

3.1.2 The Contract Price Amendment referenced in Section 2.3.2.2 herein or the Proposal accepted by Owner in accordance with Section 2.3 herein.

3.1.3 This Agreement, including all exhibits (list for example, performance standard requirements, performance incentive arrangements, markup exhibits, allowances, unit prices, or exhibit detailing offsite reimbursable personnel) but excluding, if applicable, the Contract Price Amendment;

3.1.4 The General Conditions of Contract;

3.1.5 Construction Documents prepared and approved in accordance with Section 2.4 of the General Conditions of Contract;

- 3.1.6 Exhibit B, Scope of Services; and
- 3.1.7 The following other documents, if any:

Article 4

Interpretation and Intent

4.1 Design-Builder and Owner, at the time of acceptance of the Proposal by Owner in accordance with Section 2.3 hereof, shall carefully review all the Contract Documents, including the various documents comprising the Basis of Design Documents for any conflicts or ambiguities. Design-Builder and Owner will discuss and resolve any identified conflicts or ambiguities prior to execution of the Agreement, or if applicable, prior to Owner's acceptance of the Proposal.

4.2 The Contract Documents are intended to permit the parties to complete the Work and all obligations required by the Contract Documents within the Contract Time(s) for the Contract Price. The Contract Documents are intended to be complementary and interpreted in harmony so as to avoid conflict, with words and phrases interpreted in a manner consistent with construction and design industry standards. In the event inconsistencies, conflicts, or ambiguities between or among the Contract Documents are discovered after Owner's acceptance of the Proposal, Design-Builder and Owner shall attempt to resolve any ambiguity, conflict, or inconsistency informally, recognizing that the Contract Documents shall take precedence in the order in which they are listed in Section 3.1 hereof. (Note, the parties are strongly encouraged to establish in the Contract Price Amendment or Proposal (as applicable) the priority of the various documents comprising such exhibit or proposal.)

4.3 Terms, words, and phrases used in the Contract Documents, including this Agreement, shall have the meanings given them in the General Conditions of Contract.

4.4 If Owner's Project Criteria contain design specifications: (a) Design-Builder is entitled to reasonably rely on the accuracy of the information represented in the design specifications and their compatibility with other information set forth in Owner's Project Criteria, including any design performance specifications; and (b) Design-Builder shall be entitled to an adjustment in its Contract Price and/or Contract Time(s) to the extent Design-Builder's cost and/or time of performance have been adversely impacted by such inaccurate design specification.

4.5 The Contract Documents form the entire agreement between Owner and Design-Builder and by incorporation herein are as fully binding on the parties as if repeated herein. No oral representations or other agreements have been made by the parties except as specifically stated in the Contract Documents.

Article 5

Ownership of Work Product

5.1 Any reports, information, or other data prepared or assembled by the Design Builder pursuant to this Agreement shall not be made available to any individual or organization by the Design Builder without the prior written approval of the City. During the term of this Agreement, and thereafter, the Design Builder shall not, without the prior written consent of the City, disclose to anyone any Confidential Information. The term Confidential Information for the purposes of this Agreement shall include all proprietary and confidential information of the City, including but not limited to business plans, marketing plans, financial information, designs, drawings, specifications, materials, compilations, documents, instruments, models,

source or object codes and other information disclosed or submitted, orally, in writing, or by any other medium or media. All Confidential Information shall be and remain confidential and proprietary in the City.

5.2 Any and all original sketches, pencil tracings of working drawings, plans, computations, specifications, computer disk files, writings and other documents prepared or provided by the Design Builder pursuant to this Agreement are the property of the City at the time of preparation and shall be turned over to the City upon expiration or termination of the Agreement or default by the Design Builder. The Design Builder grants the City a copyright license to use such drawings and writings. The Design Builder shall not permit the reproduction or use thereof by any other person except as otherwise expressly provided herein. The City may modify the design including any drawings or writings. Any use by the City of the aforesaid sketches, tracings, plans, computations, specifications, computer disk files, writings, and other documents in completed form as to other projects or extensions of this Project, or in uncompleted form, without specific written verification by the Design Builder will be at the City's sole risk and without liability or legal exposure to the Design Builder. The Design Builder may keep a copy of all drawings and specifications for its sole and exclusive use.

5.3 If the Design Builder should subcontract all or any portion of the services to be performed under this Agreement, the Design Builder shall cause each subcontractor to also comply with the requirements of this Section 5.

5.4 This Section 5 shall survive expiration or termination of this Agreement.

Article 6

Contract Time

6.1 Date of Commencement. The Phase 1 Services shall commence within five (5) days of Design-Builder's receipt of Owner's Notice to Proceed unless the parties mutually agree otherwise in writing. The Work shall commence within five (5) days of Design-Builder's receipt of Owner's Notice to Proceed for Phase 2 Services ("Date of Commencement") if the Proposal is accepted and the Contract Price Amendment is amended to this Agreement unless the parties mutually agree otherwise in writing.

6.2 Substantial Completion and Final Completion.

6.2.2 Interim milestones and/or Substantial Completion of identified portions of the Work shall be achieved as follows: (Insert any interim milestones ("Scheduled Interim Milestone Dates") for portions of the Work with different scheduled dates for Substantial Completion.)

6.2.3 Final Completion of the Work or identified portions of the Work shall be achieved as expeditiously as reasonably practicable. Final Completion is the date when all Work is complete pursuant to the definition of Final Completion set forth in Section 1.2.14 of the General Conditions of Contract.

6.2.4 All of the dates set forth in this Article 6 ("Contract Time(s)") shall be subject to adjustment in accordance with the General Conditions of Contract.

6.3 Time is of the Essence. Owner and Design-Builder mutually agree that time is of the essence with respect to the dates and times set forth in the Contract Documents.

6.4 Liquidated Damages. Design-Builder understands that if Substantial Completion is not attained by the Scheduled Substantial Completion Date, Owner will suffer damages which are difficult to determine and accurately specify. Design-Builder agrees that if Substantial Completion is not attained by ______

(_____) days after the Scheduled Substantial Completion Date (the "LD Date"), Design-Builder shall pay Owner ______ Dollars (\$______) as liquidated damages for each day that Substantial Completion extends beyond the LD Date. (If a GMP is not established upon execution of this Agreement, the parties should consider setting liquidated damages after GMP negotiations.)

6.5 Any liquidated damages assessed pursuant to this Agreement shall be in lieu of all liability for any and all extra costs, losses, expenses, claims, penalties, and any other damages, whether special or consequential, and of whatsoever nature, incurred by Owner which are occasioned by any delay in achieving Substantial Completion, Interim Milestone Dates (if any), or Final Completion.

Article 7

Contract Price

7.1 Contract Price.

7.1.1 Owner shall pay Design-Builder in accordance with Article 6 of the General Conditions of Contract the sum of <u>Six Hundred Thirty-Four Thousand One Hundred Seventeen</u> Dollars (\$634,117) for the Phase 1 Services, subject to adjustments made in accordance with the General Conditions of Contract. Unless otherwise provided in the Contract Documents, the Phase 1 Services compensation is deemed to include all sales, use, consumer and other taxes mandated by applicable Legal Requirements.

7.1.2 For Phase 2 Services, Owner shall pay Design-Builder in accordance with Article 6 of the General Conditions of Contract a contract price ("Contract Price") equal to the Lump Sum amount set forth in Section 7.2 hereof or in the Contract Price Amendment, or equal to the Design-Builder's Fee (as defined in Section 7.4 hereof) plus the Cost of the Work (as defined in Section 7.5 hereof), subject to any GMP established in Section 7.6 hereof or as set forth in the Contract Price Amendment and any adjustments made in accordance with the General Conditions of Contract.

7.1.3 For the specific Work set forth below, Owner agrees to pay Design-Builder, as part of the Contract Price, on the following basis: (*This is an optional section intended to provide the parties with flexibility to identify and price limited services.*)

7.2 Lump Sum. Owner shall pay Design-Builder in accordance with Article 6 of the General Conditions of Contract the sum of ______ Dollars (\$_____) ("Contract Price") for the Work for Phase 2 Services, subject to adjustments made in accordance with the General Conditions of Contract. Unless otherwise provided in the Contract Documents, the Contract Price is deemed to include all sales, use, consumer and other taxes mandated by applicable Legal Requirements.

7.3 Markups for Changes. If the Contract Price requires an adjustment due to changes in the Work, and the cost of such changes is determined under Sections 9.4.1.3 or 9.4.1.4 of the General Conditions of Contract, the following markups shall be allowed on such changes:

7.3.1 For additive Change Orders, including additive Change Orders arising from both additive and deductive items, it is agreed that Design-Builder shall receive a Fee of TEN percent (10%) of the additional costs incurred for that Change Order, plus any other markups set forth at Exhibit hereto.

7.3.2 For deductive Change Orders, including deductive Change Orders arising from both additive and deductive items, the deductive amounts shall include:

[Check one box only.]

No additional reduction to account for Design-Builder's Fee or any other markup.

or

An amount equal to the sum of: (a) ______ percent (____%) applied to the direct costs of the net reduction (which amount will account for a reduction associated with Design-Builder's Fee); plus (b) any other markups set forth at Exhibit _____ hereto applied to the direct costs of the net reduction.

- 7.4 Design-Builder's Fee: Not used
- 7.5 Cost of the Work.

7.5.1 The term Cost of the Work shall mean costs reasonably incurred by Design-Builder in the proper performance of the Work. The Cost of the Work shall include only the following:

7.5.1.1 Wages of direct employees of Design-Builder performing the Work at the Site or, with Owner's agreement, at locations off the Site, provided, however, that the costs for those employees of Design-Builder performing design services shall be calculated on the basis of prevailing market rates for design professionals performing such services or, if applicable, those rates set forth in an exhibit to this Agreement.

7.5.1.2 Wages or salaries of Design-Builder's supervisory and administrative personnel engaged in the performance of the Work and who are located at the Site or working off-Site to assist in the production or transportation of material and equipment necessary for the Work.

7.5.1.3 Wages or salaries of Design-Builder's personnel stationed at Design-Builder's principal or branch offices, but only to the extent said personnel are identified in the Design-Builder's Proposal and performing the function set forth in said Proposal. The reimbursable costs of personnel stationed at Design-Builder's principal or branch offices shall include a TEN percent (10%) markup to compensate Design-Builder for the Project-related overhead associated with such personnel.

7.5.1.4 Costs incurred by Design-Builder for employee benefits, premiums, taxes, insurance, contributions and assessments required by law, collective bargaining agreements, or which are customarily paid by Design-Builder, to the extent such costs are based on wages and salaries paid to employees of Design-Builder covered under Sections 7.5.1.1 through 7.5.1.3 hereof.

7.5.1.5 The reasonable portion of the cost of travel, accommodations and meals for Design-Builder's personnel necessarily and directly incurred in connection with the performance of the Work.

7.5.1.6 Payments properly made by Design-Builder to Subcontractors and Design Consultants for performance of portions of the Work, including any insurance and bond premiums incurred by Subcontractors and Design Consultants.

7.5.1.7 Costs incurred by Design-Builder in repairing or correcting defective, damaged or nonconforming Work (including any warranty or corrective Work performed after Substantial Completion), provided that such Work was beyond the reasonable control of Design-Builder, or caused by the ordinary mistakes or inadvertence, and not the negligence, of Design-Builder or those working by or through Design-Builder. If the costs associated with such Work are recoverable from insurance, Subcontractors or Design Consultants, Design-Builder shall exercise best efforts to obtain recovery from the appropriate source and provide a credit to Owner if recovery is obtained.

7.5.1.8 Costs, including transportation, inspection, testing, storage, and handling of materials, equipment, and supplies incorporated or reasonably used in completing the Work.

7.5.1.9 Costs less salvage value of materials, supplies, temporary facilities, machinery, equipment and hand tools not customarily owned by the workers that are not fully consumed in the performance of the Work and which remain the property of Design-Builder, including the costs of transporting, inspecting, testing, handling, installing, maintaining, dismantling, and removing such items.

7.5.1.10 Costs of removal of debris and waste from the Site.

7.5.1.11 The reasonable costs and expenses incurred in establishing, operating and demobilizing the Site office, including the cost of facsimile transmissions, long-distance telephone calls, postage and express delivery charges, telephone service, photocopying, and reasonable petty cash expenses.

7.5.1.12 Rental charges and the costs of transportation, installation, minor repairs and replacements, dismantling and removal of temporary facilities, machinery, equipment and hand tools not customarily owned by the workers, which are provided by Design-Builder at the Site, whether rented from Design-Builder or others, and incurred in the performance of the Work.

7.5.1.13 Premiums for insurance and bonds required by this Agreement or the performance of the Work.

7.5.1.14 All fuel and utility costs incurred in the performance of the Work.

7.5.1.15 Sales, use, or similar taxes, tariffs, or duties incurred in the performance of the Work.

7.5.1.16 Legal costs, court costs, and costs of mediation and arbitration reasonably arising from Design-Builder's performance of the Work, provided such costs do not arise from disputes between Owner and Design-Builder.

7.5.1.17 Costs for permits, royalties, licenses, tests and inspections incurred by Design-Builder as a requirement of the Contract Documents.

7.5.1.18 The cost of defending suits or claims for infringement of patent rights arising from the use of a particular design, process, or product required by Owner, paying legal judgments against Design-Builder resulting from such suits or claims, and paying settlements made with Owner's consent.

7.5.1.19 Deposits which are lost, except to the extent caused by Design-Builder's negligence.

7.5.1.20 Costs incurred in preventing damage, injury, or loss in case of an emergency affecting the safety of persons and property.

7.5.1.21 Accounting and data processing costs related to the Work.

7.5.1.22 Other costs reasonably and properly incurred in the performance of the Work to the extent approved in writing by Owner.

[Design-Builder and Owner may want to consider adding the following Section 7.5.1.23 to address the payment of warranty work.]

7.5.1.23 Owner and Design-Builder agree that an escrow account in the amount of <u>Six Hundred Thirty-Four Thousand One Hundred Seventeen</u> Dollars (<u>\$634,117</u>) shall be established prior to Final Completion, which escrow shall be used to reimburse Design-Builder for the Costs of the Work incurred after Final Completion to perform warranty Work. The escrow agreement will provide that any sums not used at the expiration of the warranty period shall be returned to Owner, subject to any savings Design-Builder may be entitled to under this Agreement. In the event the warranty escrow account is exhausted, but funds remain under the GMP, Owner shall be obligated to pay Design-Builder the Costs of the Work incurred after Final Completion to perform warranty Work up to the GMP.

7.5.2 Non-Reimbursable Costs. The following shall be excluded from the Cost of the Work:

7.5.2.1 Compensation for Design-Builder's personnel stationed at Design-Builder's principal or branch offices, except as provided for in Sections 7.5.1.1, 7.5.1.2, and 7.5.1.3 hereof.

7.5.2.2 Overhead and general expenses, except as provided for in Section 7.5.1 hereof, or which may be recoverable for changes to the Work.

7.5.2.3 The cost of Design-Builder's capital used in the performance of the Work.

7.5.2.4 If the parties have agreed on a GMP, costs that would cause the GMP, as adjusted in accordance with the Contract Documents, to be exceeded.

- 7.6 The Guaranteed Maximum Price.
 - 7.6.1 Design-Builder guarantees that it shall not exceed the GMP of

Dollars (\$______). Documents used as a basis for the GMP shall be identified in the Contract Price Amendment to this Agreement. Design-Builder does not guarantee any specific line item provided as part of the GMP and has the sole discretion to apply payment due to overruns in one line item to savings due to underruns in any other line item. Design-Builder agrees, however, that it will be responsible for paying all costs of completing the Work which exceed the GMP, as adjusted in accordance with the Contract Documents. (While the Contract Price Amendment will be developed in advance or concurrently with the execution of this Agreement, it is recommended that such exhibit include the items set forth in Section 2.3 above, to ensure that the basis for the GMP is well understood).

7.6.2 The GMP includes a Contingency in the amount of

) which is available for Design-Builder's exclusive use for unanticipated Dollars (\$ costs it has incurred that are not the basis for a Change Order under the Contract Documents. By way of example, and not as a limitation, such costs may include: (a) trade buy-out differentials; (b) overtime or acceleration; (c) escalation of materials; (d) correction of defective, damaged or nonconforming Work, design errors or omissions, however caused; (e) Subcontractor defaults; or (f) those events under Section 8.2.2 of the General Conditions of Contract that result in an extension of the Contract Time but do not result in an increase in the Contract Price. The Contingency is not available to Owner for any reason, including changes in scope or any other item which would enable Design-Builder to increase the GMP under the Contract Documents. Design-Builder shall provide Owner notice of all anticipated charges against the Contingency, and shall provide Owner as part of the monthly status report required by Section 2.1.2 of the General Conditions of Contract an accounting of the Contingency, including all reasonably foreseen uses or potential uses of the Contingency in the upcoming three (3) months. Design-Builder agrees that with respect to any expenditure from the Contingency relating to a Subcontractor default or an event for which insurance or bond may provide reimbursement, Design-Builder will in good faith exercise reasonable steps to obtain performance from the Subcontractor and/or recovery from any surety or insurance company. Design-Builder agrees that if Design-Builder is subsequently reimbursed for said costs, then said recovery will be credited back to the Contingency.

7.6.3 Savings.

7.6.3.1 If the sum of the actual Cost of the Work and Design-Builder's Fee (and, if applicable, any prices established under Section 7.1.3 hereof) is less than the GMP, as such GMP may have been adjusted over the course of the Project, the difference ("Savings") shall be shared as follows:

[Choose one of the following.]



or

The first	Dollars (\$) of	Savings
shall be provided to (choose either Design-	Builder or Owner)		,
with the balance of Savings, if any, shared	, <u> </u>	percent (%)
to Design-Builder and	percent (%) to C	Owner.

7.6.3.2 Savings shall be calculated and paid as part of Final Payment under Section 8.4 hereof, with the understanding that to the extent Design-Builder incurs costs after Final Completion which would have been payable to Design-Builder as a Cost of the Work, the parties shall recalculate the Savings in light of the costs so incurred, and Design-Builder shall be paid by Owner accordingly.

7.7 Allowance Items and Allowance Values.

7.7.1 Any and all Allowance Items, as well as their corresponding Allowance Values, are set forth in the Contract Price Amendment or the Proposal.

7.7.2 Design-Builder and Owner have worked together to review the Allowance Items and Allowance Values based on design information then available to determine that the Allowance Values constitute reasonable estimates for the Allowance Items. Design-Builder and Owner will continue working closely together during the preparation of the design to develop Construction Documents consistent with the Allowance Values. Nothing herein is intended in any way to constitute a guarantee by Design-Builder that the Allowance Item in question can be performed for the Allowance Value.

7.7.3 No work shall be performed on any Allowance Item without Design-Builder first obtaining in writing advanced authorization to proceed from Owner. Owner agrees that if Design-Builder is not provided written authorization to proceed by the date set forth in the Project schedule, due to no fault of Design-Builder, Design-Builder may be entitled to an adjustment of the Contract Time(s) and Contract Price.

7.7.4 The Allowance Value includes the direct cost of labor, materials, equipment, transportation, taxes, and insurance associated with the applicable Allowance Item. All other costs, including design fees, Design-Builder's overall project management and general conditions costs, overhead and Fee, are deemed to be included in the original Contract Price, and are not subject to adjustment notwithstanding the actual amount of the Allowance Item.

7.7.5 Whenever the actual costs for an Allowance Item is more than or less than the stated Allowance Value, the Contract Price shall be adjusted accordingly by Change Order, subject to Section 7.7.4. The amount of the Change Order shall reflect the difference between actual costs incurred by Design-Builder for the particular Allowance Item and the Allowance Value.

7.8 Performance Incentives.

7.8.1 Owner and Design-Builder have agreed to the performance incentive arrangements set forth in the Design-Builder's Proposal.

Article 8

Procedure for Payment

8.1 Payment for Preliminary Services. Design-Builder and Owner agree upon the following method for partial and final payment to Design-Builder for the services hereunder: (Insert terms.)

8.2 Contract Price Progress Payments.

8.2.1 Design-Builder shall submit to Owner on the <u>FIFTEENTH</u> (15TH) day of each month, beginning with the first month after the Date of Commencement, Design-Builder's Application for Payment in accordance with Article 6 of the General Conditions of Contract.

8.2.2 Owner shall make payment within ten (10) days after Owner's receipt of each properly submitted and accurate Application for Payment in accordance with Article 6 of the General Conditions of Contract, but in each case less the total of payments previously made, and less amounts properly withheld under Section 6.3 of the General Conditions of Contract.

8.2.3 If Design-Builder's Fee under Section 7.4 hereof is a fixed amount, the amount of Design-Builder's Fee to be included in Design-Builder's monthly Application for Payment and paid by Owner shall be proportional to the percentage of the Work completed, less payments previously made on account of Design-Builder's Fee.

8.3 Retainage on Progress Payments.

8.3.1 Owner will retain <u>FIVE</u> percent (5%) of each Application for Payment provided, however, that when fifty percent (50%) of the Work has been satisfactorily completed by Design-Builder and Design-Builder is otherwise in compliance with its contractual obligations, Owner will not retain any additional retention amounts from Design-Builder's subsequent Applications for Payment. Owner will also reasonably consider reducing retainage for Work completed early in the Project.

8.3.2 Within fifteen (15) days after Substantial Completion of the entire Work or, if applicable, any portion of the Work, pursuant to Section 6.6 of the General Conditions of Contract, Owner shall release to Design-Builder all retained amounts relating, as applicable, to the entire Work or completed portion of the Work, less an amount equal to: (a) the reasonable value of all remaining or incomplete items of Work as noted in the Certificate of Substantial Completion; and (b) all other amounts Owner is entitled to withhold pursuant to Section 6.3 of the General Conditions of Contract.

[If Owner and Design-Builder have established a warranty reserve pursuant to Section 7.5.1.23 above, the following provision should be included.]

If a warranty reserve has been established pursuant to Section 7.5.1.23 above, Owner shall at the time of Substantial Completion retain the agreed-upon amounts and establish an escrow account as contemplated by Section 7.5.1.23 above.

8.4 Final Payment. Design-Builder shall submit its Final Application for Payment to Owner in accordance with Section 6.7 of the General Conditions of Contract. Owner shall make payment on Design-Builder's properly submitted and accurate Final Application for Payment (less any amount the parties may have agreed to set aside for warranty work) within ten (10) days after Owner's receipt of the Final

Application for Payment, provided that Design-Builder has satisfied the requirements for final payment set forth in Section 6.7.2 of the General Conditions of Contract.

8.5 Interest. Payments due and unpaid by Owner to Design-Builder, whether progress payments or final payment, shall bear interest commencing five (5) days after payment is due at the rate of ONE percent (1%) per month until paid.

8.6 Record Keeping and Finance Controls. Design-Builder acknowledges that this Agreement is to be administered on an "open book" arrangement relative to Costs of the Work. Design-Builder shall keep full and detailed accounts and exercise such controls as may be necessary for proper financial management, using accounting and control systems in accordance with generally accepted accounting principles and as may be provided in the Contract Documents. During the performance of the Work and for a period of three (3) years after Final Payment, Owner and Owner's accountants shall be afforded access to, and the right to audit from time to time, upon reasonable notice, Design-Builder's records, books, correspondence, receipts, subcontracts, purchase orders, vouchers, memoranda, and other data relating to the Work, all of which Design-Builder shall preserve for a period of three (3) years after Final Payment. Such inspection shall take place at Design-Builder's offices during normal business hours unless another location and time is agreed to by the parties. Any multipliers or markups agreed to by Owner and Design-Builder as part of this Agreement are only subject to audit to confirm that such multiplier or markup has been charged in accordance with this Agreement, but the composition of such multiplier or markup is not subject to audit. Any lump sum agreed to by Owner and Design-Builder as part of this Agreement is not subject to audit.

8.7 Prompt Progress Payment to Subcontractors.

All Contracts and Subcontracts (all tiers), shall contain the following provisions:

8.7.1 Prompt Payment To Subcontractors A Design-Builder, prime contractor or subcontractor shall pay a subcontractor not later than 10 days of receipt of each progress payment in accordance with the provision in Section 7108.5 of the California Business and Professions Code concerning prompt payment to subcontractors. The 10 days is applicable unless, a longer period is agreed to in writing. Any violation of Section 7108.5 shall subject the violating contractor or subcontractor to the penalties, sanction and other remedies of that section. Federal regulation (49 CFR 26.29) requires that any delay or postponement of payment over 30 days of receipt of each payment may take place only for good cause and with the Owner's prior written approval. These requirements shall not be construed to limit or impair any contractor in the event of a dispute involving late payment, or nonpayment by the Design-Builder, deficient subcontract performance, or noncompliance by a subcontractor. This provision applies to both low-and very low-income prime contractors and subcontractors as well as those that are not.

Prompt Payment Of Funds Withheld To Subcontractors. The Owner shall hold retainage 8.7.2 from the prime contractor and shall make prompt and regular incremental acceptances of portions, as determined by the Owner of the contract work and pay retainage to the Design-Builder based on these acceptances. The Design-Builder, prime contractor or subcontractor shall return all monies withheld in retention from a subcontractor within 30 days after receiving payment for work satisfactorily completed and accepted including incremental acceptances of portions of the contract work by the Owner. Federal regulations (49 CFR 26.29) require that any delay or postponement of payment over 30 days may take place only for good cause and with the agency's prior written approval. Any violation of this provision shall subject the violating prime contractor or subcontractor to the penalties, sanctions, and other remedies specified in Section 7108.5 of the California Business and Professions Code. These requirements shall not be construed to limit or impair any contractual, administrative, or judicial remedies otherwise, available to the prime contractor or subcontractor in the event of a dispute involving late payment, or nonpayment by the prime contractor, deficient subcontract performance, or noncompliance by a subcontractor. This provision applies to both low-and very low-income prime contractors and subcontractors as well as those that are not.

Article 9

Termination for Convenience

9.1 Upon ten (10 days' written notice to Design-Builder, Owner may, for its convenience and without cause, elect to terminate this Agreement. In such event, Owner shall pay Design-Builder for the following:

9.1.1 All services performed and Work executed and for proven loss, cost, or expense in connection with the services and Work;

9.1.2 The reasonable costs and expenses attributable to such termination, including demobilization costs and amounts due in settlement of terminated contracts with Subcontractors and Design Consultants; and

[Choose one of the following.]

The fair and reasonable sums for overhead and profit on the sum of items 9.1.1 and 9.1.2 above.

or

Overhead and profit in the amount of TEN percent (10%) on the sum of items 9.1.1 and 9.1.2 above.

9.2 In addition to the amounts set forth in Section 9.1 above, Design-Builder shall be entitled to receive one of the following if the parties agree to an additional payment:

9.2.1 If Owner terminates this Agreement prior to commencement of construction, Design-Builder shall be paid ZERO percent (0%) of the remaining balance of the Contract Price or, if a GMP has not been established, the remaining balance of the most recent estimated Contract Price.

9.2.2 If Owner terminates this Agreement after commencement of construction, Design-Builder shall be paid TEN percent (10%) of the remaining balance of the Contract Price or, if a GMP has not been established, the remaining balance of the most recent estimated Contract Price.

9.3 If Owner terminates this Agreement pursuant to Section 9.1 above and proceeds to design and construct the Project through its employees, agents or third parties, Owner's rights to use the Work Product shall be as set forth in Article 5 hereof. Such rights may not be transferred or assigned to others without Design-Builder's express written consent and such third parties' agreement to the terms of Article 5.

Article 10

Representatives of the Parties

10.1 Owner's Representatives.

10.1.1 Owner designates the individual listed below as its Senior Representative ("Owner's Senior Representative"), which individual has the authority and responsibility for avoiding and resolving disputes under Section 10.2.2 of the General Conditions of Contract: (Identify individual's name, title, address, and telephone numbers.)

Scott Mozier, PE Public Works Director 2600 Fresno Street Fresno, CA 93721 559-621-8650

10.1.2 Owner designates the individual listed below as its Owner's Representative, which individual has the authority and responsibility set forth in Section 3.4 of the General Conditions of Contract: (Identify individual's name, title, address, and telephone numbers.)

Mike Mooneyham, PE Professional Engineer 2600 Fresno Street Fresno, CA 93721 559-621-8623

10.2 Design-Builder's Representatives.

10.2.1 Design-Builder designates the individual listed below as its Senior Representative ("Design-Builder's Senior Representative"), which individual has the authority and responsibility for avoiding and resolving disputes under Section 10.2.2 of the General Conditions of Contract: (Identify individual's name, title, address, and telephone numbers.)

Benjamin Lingo Principal 20750 Ventura Boulevard, Suite 155 Woodland Hills, CA 91364 818-974-9246

Charles Brumbaugh Principal 20750 Ventura Boulevard, Suite 155 Woodland Hills, CA 91364 818-974-9395

10.2.2 Design-Builder designates the individual listed below as its Design-Builder's Representative, which individual has the authority and responsibility set forth in Section 2.1.1 of the General Conditions of Contract: (Identify individual's name, title, address, and telephone numbers.)

Jake Lingo Senior Vice President 20750 Ventura Boulevard, Suite 155 Woodland Hills, CA 91364 818-974-2966

Article 11

Bonds and Insurance

11.1 Insurance. Design-Builder and Owner shall procure the insurance coverages set forth in the Insurance Exhibit attached hereto and in accordance with Article 5 of the General Conditions of Contract.

11.2 Bonds and Other Performance Security. Design-Builder shall provide the following performance bond and labor and material payment bond or other performance security:

Performance Bond. (Check one box only.	If no box is checked, then no bond is required.)
🛛 Required	Not Required
Payment Bond. [Check one bo	x only. If no box is checked, then no bond is required.]
🛛 Required	Not Required
Other Performance Sec	urity.

[Check one box only. If no box is checked, then no other performance security is required. If the "Required" box is checked, identify below the specific performance security that is being required and all salient commercial terms associated with that security.]

Required

Not Required

Article 12

Other Provisions

- 12.1 Other provisions, if any, are as follows: (Insert any additional provisions.)
- 12.2 Listing of Exhibits and documents incorporated herein:

Exhibit A – Owner's Project Criteria Exhibit B – Scope of Services Exhibit C – Federal Requirements Exhibit D – (Intentionally Omitted) Exhibit E – Disclosure of Conflict of Interest DBIA Document No. 535, Standard Form of General Conditions of Contract Between Owner and Design-Builder (2022 Edition) ("General Conditions of Contract") Contract Price Amendment, if any. In executing this Agreement, Owner and Design-Builder each individually represents that it has the necessary financial resources to fulfill its obligations under this Agreement, and each has the necessary corporate approvals to execute this Agreement, and perform the services described herein.

OWNER:

DESIGN-BUILDER:

(Name of Design-Builder)

Scott Mozier, PE (Name of Owner)

(Signature) Randall Morrison, PE Assistant Director of Public Works (Printed Name)

for

Signature) BRUM HARLES UGU (Printed Name) MANAGER (Title) Date: 202

NTEGRATED COMMUNITY DEVELOPMENT

Public Works Director (Title)

Date: 07/20/2023

Caution: An original DBIA document has this caution printed in blue. This is a printable copy and an original assures that changes will not be obscured as may occur when documents are reproduced.

ATTEST: TODD STERMER, CMC CITY CLERK By: 20-2023 Deputy usan

The license for use of this document expires 1 year from the date of purchase. To renew your license, visit <u>store.dbia.org</u>.

Questions? We're here to help.

Contact us



Design-Build Institute of America 1001 Pennsylvania Ave. NW, Suite 410 Washington, DC 20004

> (202) 682-0110 dbia@dbia.org

EXHIBIT A

OWNER'S PROJECT CRITERIA

Progressive Design-Build Agreement Between

City of Fresno (City) and Integrated Community Development (Design-Build Entity)

SENIOR ACTIVITY CENTER AND AFFORDABLE HOUSING PROJECT

1. Project Delivery and Objectives

- 1.1. The Senior Activity Center (Center) and Senior Affordable Housing Project (Housing) will be designed and constructed in two phases using the progressive design-build delivery method as follows:
 - 1.1.1. Preconstruction Phase (Phase One): Prepare design to approximately 80 percent complete and establish a Guaranteed Maximum Price (GMP) proposal.
 - 1.1.2. Construction Phase (Phase Two): Upon award of a contract for the GMP or lump-sum price by the Owner, complete design, construction, and post-construction tasks including site facilities testing, as-built recording, warranty, and support.
- 1.2. Construction will consist of a 30,000 square foot Senior Activity Center as well as up to 100 units of Senior Affordable Housing. The City intends to own, manage, and operate the Senior Activity Center; while the Design-Build Entity will own, manage and operate the Senior Affordable Housing.
- 1.3. The site shall be designed to accommodate the Senior Activity Center and affordable housing units, optimizing use of shared spaces (e.g., parking and other amenities) as much as possible. Together Center and Housing shall be referred to as the Facility.
- 1.4. The Senior Activity Center must be located on parcel 426-253-17 and the Senior Affordable Housing must be developed on parcel 426-253-19.
- 1.5. The City's objectives for delivery of the Project are as follows:
 - 1.5.1. Design a high-quality Center, while engaging with the community to include needed features and amenities
 - 1.5.2. Deliver a high-quality Facility in accordance with industry construction standards within the prescribed schedule and budget
 - 1.5.3. Ensure a high degree of cost and schedule certainty
 - 1.5.4. Achieve an optimal balance of risk allocation between the City and the Design-Builder
 - 1.5.5. Develop the two City owned parcels in a coordinated, efficient, and complimentary manner to best serve the targeted senior community
 - 1.5.6. Provide a Center to offer classes, activities, and resources to support, enhance and encourage an active healthy lifestyle for an active senior population
 - 1.5.7. Provide a safe, comfortable, and accessible environment for senior living and socializing

2. Communications and Submittals

2.1. Regular meetings will be held through the life of the project. The Owner and Design-Builder will work together to determine the appropriate frequency, duration, and format (virtual or in-person).

2.2. All document submittals must be addressed to the Project Manager for the City of Fresno. Document delivery may be via email, Microsoft Teams, or file transfer protocol. A transmittal cover sheet must be utilized for all document submittals.

3. Funding and Financing

- 3.1. The work hereunder constitutes a "public work" as defined in Chapter 1, Part 7, Division 2 of the California Labor Code, and Contractor shall cause the work to be performed as a "public work" in accordance with such Chapter of the California Labor Code. The Council of the City of Fresno has adopted Resolution No. 82 297 ascertaining the general prevailing rate of per diem wages and per diem wages for holidays and overtime in the Fresno area for each craft, classification, or type of worker needed in the execution of Contracts for the City. Information specific to the Work to be done under this Contract can be obtained by contacting the Contract Compliance Officer at the City of Fresno office of Construction Management, (559) 621-5600.
- 3.2. This is a federal project funded in full or in part by the Community Development Block Grant (CDBG) Program as administered by the U.S. Department of Housing and Urban Development (HUD). This project is subject to the payment of minimum wages mandated by the Davis-Bacon Act of 1931. A copy of the Federal wage rates is enclosed in the contract Agreement between the Owner and Design-Builder.
- 3.3. This project is subject to the payment of State or Federal wage rates, whichever rate is higher.
- 3.4. The successful Design-Build Entity will be responsible for financing the Senior Affordable Housing Development.
- 3.5. The City may contribute an amount, to be determined through development proposals submitted in response to the City's Affordable Housing Development Notice of Funding Availability, toward the Senior Affordable Housing Development. All funds contributed to this Development may consist of a combination of Federal, State, or local sources. Without this contribution the Housing is infeasible. However, this does not commit the City to provide any funds towards the development of the Housing.
- 3.6. A ground lease and service contract will be negotiated between the selected Design-Build Entity and the City, for a period of 55 years, for the Senior Affordable Housing Development.
- 3.7. The site located at 4323-4333 N. Blackstone Ave, Fresno, California, (APN 426-253-19) is owned by the City and will remain under the City's ownership.
- 3.8. The Design-Builder will enter into a development agreement for a low-cost, long-term ground lease transaction for the Senior Affordable Housing site and agrees to accept the Housing Property in its present state and condition, as-is.
- 3.9. The Property was purchased using a combination of Community Development Block Grant funds and Measure P funds and has been determined by the Fresno City Council and the State's Housing and Community Development Department (HCD) to be surplus exempt property under a qualifying exemption but is still subject to certain conditions of the Surplus Land Act and the requirements of California Government Code §§ 54220-54234.

4. Site Overview

4.1. This site consists of two contiguous parcels: Fresno County Tax Assessor Parcels 426-253-17 and 426-253-19. The site is located west of Blackstone Avenue, between Holland and Swift Avenues, as shown below (Figure 1). The northern parcel has been vacated and will be demolished to allow for the development of a Senior Activity Center. The southern parcel currently has tenants that will be relocated by City to allow for demolition and the development of Senior Affordable Housing.



Figure 1: Site Location

4.2. For information on allowable land uses, please reference the link below:

https://library.municode.com/ca/fresno/codes/code_of_ordinances?nodeId=MUCOFR_CH15CIDECOIN RE_PTIIBAOVDI_ART11MIEDIMX&showChanges=true

- 4.3. APN 426-253-17 (Senior Activity Center):
 - 4.3.1. Zoning Highlights
 - Size: 4.25 Acres
 - Zoning: CMX (Corridor/Center Mixed-Use District)
 - The CMX district is intended to allow for either horizontal or vertical mixed-use development along key circulation corridors in the city where height and density can be easily

accommodated. Ground-floor retail and upper-floor residential or offices are the primary uses, with residential uses, personal and business services, and public and institutional space as supportive uses. Development will facilitate the transformation of existing transportation corridors into vibrant, highly walkable areas with broad, pedestrian-friendly sidewalks, trees, landscaping, and local-serving uses with new buildings that step down in relationship to the scale and character of adjacent neighborhoods.

- Residential Density du/ac: 16 Minimum, No Maximum
- Height (Maximum): 60 Feet
- 4.4. APN 426-253-19 (Senior Affordable Housing):
 - 4.4.1. Zoning Highlights
 - Size: 1.26 Acres
 - Zoning: CMX (Corridor/Center Mixed-Use District)
 - The CMX district is intended to allow for either horizontal or vertical mixed-use development along key circulation corridors in the city where height and density can be easily accommodated. Ground-floor retail and upper-floor residential or offices are the primary uses, with residential uses, personal and business services, and public and institutional space as supportive uses. Development will facilitate the transformation of existing transportation corridors into vibrant, highly walkable areas with broad, pedestrian-friendly sidewalks, trees, landscaping, and local-serving uses with new buildings that step down in relationship to the scale and character of adjacent neighborhoods.
 - Residential Density du/ac: 16 Minimum, No Maximum
 - Height (Maximum): 60 Feet

4.5. Utilities:

- 4.5.1. The site has access to:
 - Water (City of Fresno)
 - Trash (City of Fresno)
 - Sewer (City of Fresno)
 - Storm Water (Fresno Metropolitan Flood Control)
 - Gas (PG&E)
 - Electric (PG&E)

5. Senior Activity Center Scope of Work

5.1. The size of the facility and specific site features and amenities will be determined through a community engagement process and may include features such as community education rooms, aquatic facilities, a commercial kitchen, multi-purpose spaces, and indoor and outdoor recreational and fitness facilities. All aspects of the design should prioritize accessibility. The exact size of the Senior Activity Center will be determined based on the feature selection following community engagement, but 30,000 square feet of indoor space is anticipated. The building may be multi-story or single-story depending on various factors such as community input and budget.

- 5.2. The Design-Builder shall include the following systems in the facility and shall work with the City to define the requirements during the design phase. These systems include, but are not limited to:
 - 5.2.1. Building Management System Metasys (or similar system) for the purpose of integrating and controlling energy efficient lighting and HVAC systems
 - 5.2.2. Emergency standby generator
 - 5.2.3. Access control
 - 5.2.4. Interior and exterior security cameras
 - 5.2.5. Security system
 - 5.2.6. IDF Rooms
 - 5.2.7. Structural cabling for network communication
 - 5.2.8. Low voltage cabling
 - 5.2.9. Fiber connectivity between buildings/future buildings
 - 5.2.10. City standard enterprise networking equipment (switches, access points, voip phones) from Cisco
 - 5.2.11. Install a locking lid Christy P-48 vault and 4" conduit from the outside the Senior Activity Center to the MPOE.
 - 5.2.12. Directional bore to install (2) 1.5" HDPE conduits from the new Senior Activity Center vault to the existing ITS vault on the east side of Blackstone Avenue.
 - 5.2.13. Install, terminate, and splice a 144-count fiber optic cable from the Senior Activity Center MPOE to the existing ITS vault.
 - 5.2.14. Install Conduits for future EV Charging Infrastructure

6. Community Engagement Process

- 6.1. The City, in collaboration with Southwest Strategies, is facilitating the first phase of the community engagement process July 5 August 30, 2023, including the staffing of a Senior Center Outreach Headquarters; a nine (9) meeting workshop series; survey gathering; and door-to-door campaigning.
- 6.2. The Design-Builder shall actively participate in the first phase of the engagement process including:
 - 6.2.1. Attendance of at least one-half (1/2) of the scheduled in-person community meetings.
 - 6.2.2. Provide, at minimum, one (1) staff member for 20 hours per week for purposes of staffing the Senior Center Outreach Headquarters through August 30, 2023.
 - 6.2.3. Provide data-entry support for physical paper surveys.
- 6.3. The Design-Builder shall propose a second phase of the community engagement process to include:
 - 6.3.1. At minimum seven (7) in-person and two (2) hybrid/virtual community meetings for purposes of presenting and collecting feedback on the Design-Builder's schematic design. Including all collateral required for advertising in print, digital, and social media.
 - 6.3.2. At minimum, one (1) in-person and one (1) hybrid/virtual community meetings for purposes of presenting the final Design-Builder's 80%/Pre-construction design to the public. Including all collateral required for advertising in print, digital, and social media. Recommended inperson meeting location is the Senior Center Outreach Headquarters.

- 6.3.3. All materials presented to the public, in any medium, must be translated into Hmong, Spanish, and Punjabi by the Design-Builder or their sub-consultant and submitted to the City for review and approval prior to public release, unless explicitly approved by the City.
- 6.3.4. Provide, at minimum, one (1) staff member for 40 hours per week for purposes of staffing the Senior Center Outreach Headquarters beginning September 1, 2023, until such time the building will be demolished, or Design-Builder has entered the Construction phase, whichever comes first.

7. Senior Affordable Housing Development Scope of Work

- 7.1. Affordable rental housing development for Seniors, consisting of studio, one bedroom, and twobedroom units. Development to also include parking, which may be shared with the neighboring activity center.
- 7.2. The proposed development and affordability restrictions support the findings that the property qualifies as exempt from the California Surplus Land Act as stated in Government Code Section 37364(a) which requires:
 - 7.2.1. Minimum of 80% of the area of any parcel shall be used for development of housing (remaining 20% could be ancillary commercial or park/open space use);
 - 7.2.2. Not less than 40% of the total number of housing units developed on any parcel pursuant to this section shall be affordable to households whose incomes are equal to, or less than, 75 percent of the maximum income of lower income households (80% of area median income) and at least half of which (20% of the units) shall be affordable to very low-income households (50% of area median income); and
 - 7.2.3. Dwelling units shall be restricted by regulatory agreement to remain continually affordable to those persons and families for the longest feasible time, but not less than 30 years and shall be recorded against the property; and
 - 7.2.4. Other income restrictions shall apply depending on funding sources. In addition, the site may be subject to the U.S. Department of Housing and Urban Development (HUD) Community Development Block Grant Program requirements for very low-, low- and moderate-income housing, and any other affordability requirements of proposed funding sources.
- 7.3. HCD requires that a Covenant or Restriction to be recorded on the Property. The affordable covenant/restriction states if 10 or more units of housing is ever built on the property 15% of the units must remain affordable for 45-55 years (depending on whether rental or owner-occupied units.)

8. Sustainability

- 8.1. All development must meet or exceed the requirements as stated in the following:
 - 8.1.1. Fresno Municipal Code Section 4-116
 - 8.1.2. California Building Energy Efficiency Standards Title 24

9. Standard Drawings and Standard Specifications

- 9.1. The development of the Senior Activity Center and Affordable Housing must be in compliance with City of Fresno Standard Drawings and Standard Specifications.
- 9.2. For information on City of Fresno Standard Drawings and Standard Specifications, please reference the link below:

https://www.fresno.gov/publicworks/developer-doorway/#tab-8

10. Environmental Clearance

10.1. The City has completed an Environmental Assessment for the development of both parcels, with a Finding of No Significant Impact [24 CFR 58.40(g)(1); 40 CFR 1508.27] (Attachment D). It will ultimately be the responsibility of the Design-Builder to ensure all environmental requirements/mitigations are completed as part of the Environmental Assessment Findings, as shown below:

10.1.1. Mitigation Measures and Conditions [40 CFR 1505.2(c)]

- Summarize below all mitigation measures adopted by the Responsible Entity to reduce, avoid, or eliminate adverse environmental impacts and to avoid non-compliance or nonconformance with the above-listed authorities and factors. These measures/conditions must be incorporated into project contracts, development agreements, and other relevant documents. The staff responsible for implementing and monitoring mitigation measures should be clearly identified in the mitigation plan.
- 10.1.2. Clean Air Compliance Measures:
 - According to the San Joaquin Valley Air Pollution Control District (District), the Project specific annual criteria pollutants emissions from construction and operation emissions of are not expected to exceed any of the District significance thresholds as identified in the District's Guidance for Assessing and Mitigating Air Quality Impacts (GAMAQI). Other potential air quality impacts related to Toxic Air Contaminants, Ambient Air Quality Standards, Hazards and Odors, may require assessments and mitigation. In addition, the Project may be subject to other District rules, which include but are not limited to: Rule 4102 (Nuisance), Rule 4601 (Architectural Coatings), Rule 4641 (Cutback, Slow Cure, and Emulsified Asphalt, Paving and Maintenance Operations), Rule 4702 (Internal Combustion Engines - Phase 1), Rule 4201 (Internal Combustion Engines - Phase 2), and Rule 4101 (Visible Emissions).
- 10.1.3. Contamination and Toxic Substances [24 CFR Part 50.3(i) & 58.5(i)(2] Mitigation Measures:
 - Mitigation Measures may be required if the following conditions are encountered during redevelopment of the subject site:
 - Former fuel storage tank(s) and/or piping or impacts to soil from former fuel storage tanks and/or piping are identified
 - Impacts to soil in the vicinity of the former dry cleaners or auto repair are identified
 - Asbestos and/or lead based paint is identified in building materials upon completion of the comprehensive surveys
- 10.2. A qualified consultant should be contacted to assess potential subject site impacts and to support compliance with regulatory requirements. Mitigations measures may include a limited subsurface investigation to assess potential soil, soil gas, and/or groundwater impacts.
- 10.3. At the time the Environmental Assessment, there were no plans to renovate or demolish the multitenant commercial buildings along the southern portion of the subject site. However, further environmental analysis is planned to evaluate the impacts of demolishing these buildings to development Senior Affordable Housing.

EXHIBIT B

SCOPE OF SERVICES

Progressive Design-Build Agreement Between

City of Fresno (City) and Integrated Community Development (Design-Build Entity)

Senior Activity Center

Scope of Work

The City seeks to develop a new Senior Activity Center and Affordable Housing on two contiguous Cityowned properties in central Fresno, located at 4323-4343 North Blackstone Avenue (APN 426-253-19 and APD 426-253-17). The new development will expand services to the community by connecting seniors to vital resources such as the senior hot meal program, health and fitness programs, wellness programs, transportation services, volunteer opportunities, educational and arts programs, and technology programs.

The size, specific site features, and amenities of the Senior Activity Center will be determined through a community engagement process and may include features such as community education rooms, aquatic facilities, a commercial kitchen, multi-purpose spaces, and indoor and outdoor recreational and fitness facilities. All aspects of the design should prioritize accessibility. The exact size of the Senior Activity Center will be determined based on the feature selection following community engagement, but a minimum of 30,000 square feet of indoor space is anticipated. The building may be multi-story or single-story depending on various factors such as community input and budget. The site shall be designed to accommodate the Senior Activity Center and up to 100 units of affordable housing, optimizing use of shared spaces (e.g., parking, and other amenities) as much as possible.

Scope of Services – Preconstruction Services Phase 1

- Task 01 Site Assessment
 - Review and verify project site and existing conditions
 - Review zoning regulations
 - Review regulatory requirements and planning and development standards
 - o Review existing technical documentation, surveys, and engineering reports
 - Survey subsurface utility locations
 - Perform additional engineering studies as required to support design and cost estimating
 - o Provide geotechnical testing as required
- Task 02 Preliminary Master Site Plan
 - $\circ\,$ Develop a Master Site Plan concept to be utilized as the initial starting point for public engagement
- Task 03 Community Engagement Campaign
 - Develop a community engagement campaign to engage the community in the design process
 - Review previously gathered community feedback defining needed features and amenities and potentially meet with prior consultant(s) if necessary

- Prepare materials for and lead inclusive community engagement initiatives, including community meetings
- Present design development plans (up to and including 80%) at community meetings, gather additional feedback, and incorporate in subsequent designs
- Provide materials and engagement opportunities in accessible settings in multiple languages to include at minimum English, Spanish, Hmong, Punjabi, and ASL
- Provide thoughtful leadership in innovative or nontraditional means of engaging hard-to-reach individuals and communities
- Social media, print, radio, television, mail, etc.
- Provide one staff member for 40 hours per week at the Senior Center Outreach Headquarters as specified in the Owner's Project Criteria
- Task 04 Plan Development
 - Develop the architectural/engineering design (including preparing and submitting intermediate design review packages) based on active engagement with the community and City
 - o Provide an Opinion of Probable Construction Cost with each plan revision
 - Confirm adequacy of existing NEPA/CEQA documentation as design progresses
- Task 05 Proposed Design and Cost Assessment
 - Produce the basis-of-design report and preliminary cost estimate
 - o Summarize preliminary design criteria
 - Develop preliminary design including preliminary drawings and technical specification outlines
 - Develop proposed schedule
 - o Regulatory requirements and permit acquisition planning
 - Develop project budget estimate
- Task 06 Project Management:
 - o Develop and implement Project Management Plan
 - o Develop and implement Quality Plan
 - Develop and implement Change Management Plan
 - Develop a project execution plan and project schedule
 - o Administer bi-weekly team meetings and updates with City staff
 - Submit progress reports as required
 - Submit progress payment and invoicing documentation
 - Prepare a project cost model and provide detailed cost estimates as the design and design alternatives are advanced
 - Prepare, submit, and negotiate a GMP or lump-sum price to complete the Construction Services Phase 2:
 - GMP proposal development
 - Final construction work plan and schedule development
 - Final regulatory and permitting plan

Phase 2 contract finalization and negotiation meetings

Compensation

Compensation for all services described above. Fees will be billed monthly in accordance with the work completed.

PRECONSTRUCTION SERVICES PHASE 1 \$ 634,117

Additional Service Fees

There would be no additional services without prior written authorization by the City.

Schedule

Time allotted for each phase is described in detail in Agreement Section 1 Scope of Services and is summarized below.

Part 1: Programming and Schematic Design	60 calendar days
Part 2: Design Development Phase	90 calendar days
Part 3: Construction Document Phase	120 calendar days
Part 4: Bidding Phase Assistance	No defined duration
Part 5: Construction Phase Assistance	No defined duration

EXHIBIT C

FEDERAL REQUIREMENTS

Requirements Checklist

	Federal Requirement	Applicability Threshold
	False Information	
	Access to Project Site and Records	
	Equal Employment Opportunity	
	Form: Certification of Compliance of Affirmative	
	Action (only required for non-construction	
	contracts more than \$50,000)	All Contracts
Ι	Suspension and Debarment	All Contracts
	Subcontracting with Small, Minority Firms, and	
	Women's Business Enterprises	
	Form: Good Faith Effort Solicitation List	
	Form: Good Faith Effort Certification	
	Procurement of Recovered Materials	
	Energy Efficiency	

	Federal Requirement	Applicability Threshold
П	Anti-Lobbying	Contracts in excess of \$100,000
III	Clean Air Act	Contracts in excess of \$150,000

	Construction Contracts	Applicability Threshold
IV	Accessibility Requirements	All Construction Contracts
V	Federal Labor Standards Form: Current DB Wage Determination	Construction contracts in excess of \$2,000
VI	Contract Work Hours and Safety Standards	Contracts in excess of \$100,000 that involve the employment of mechanics or laborers
VII	Section 3 Clause Form: Section 3 Plan Form: Section 3 Assurance Form: Section 3 Workforce Breakdown Form: Section 3 Subcontract Breakdown	Contracts that (1) involve public construction, housing rehabilitation, or housing construction and (2) are in excess of \$100,000
VIII	Bonding Requirements	For construction or facility improvement contracts or subcontracts exceeding the Simplified Acquisition Threshold (\$250,000 as of March 2018)

False Information

Bidder is advised that providing false, fictitious or misleading information with respect to CDBG funds may result in criminal, civil or administrative prosecution under 18 U.S.C. § 1001, 18 U.S.C. § 1343, 31 U.S.C. § 3729, 31 U.S.C. § 3801 or another applicable statute. Bidder shall promptly refer to City and HUD's Office of the Inspector General any credible evidence that a principal, employee, agent, subcontractor, or other person has submitted a false claim under the False Claims Act or has committed a criminal or civil violation of laws pertaining to fraud, conflict of interest, bribery, gratuity, or similar misconduct involving CDBG funds. Bidder shall ensure that contractual language in third party contracts enforces these provisions.

Access to Project Site and Records

Bidder will provide access to the City, HUD, the Comptroller General of the United States, or any of their duly authorized representatives to any books, documents, papers, and records of the contractor which are directly pertinent to that specific contract for the purpose of making audit, examination, excerpts, and transcriptions. Contractor will retain all required records for three years after final payments are made and all other pending matters are closed.

Bidder will provide suitable access to the project site at all reasonable times during construction to the City, HUD, the Comptroller General of the United States, or any of their duly authorized representatives. Contractor shall also meet all reporting requirements to allow City to comply with the Federal Funding Accountability and Transparency Act of 2006 (Public Law 109–282).

Equal Employment Opportunity

Bidder shall abide by all Executive Order 11246, "Equal Employment Opportunity" (30 FR 12319, 12935, 3 CFR Part, 1964-1965 Comp., p. 339), as amended by Executive Order 11375, "Amending Executive Order 11246 Relating to Equal Employment Opportunity," and implementing regulations at 41 CFR part 60, "Office of Federal Contract Compliance Programs, Equal Employment Opportunity, Department of Labor. All contracts and subcontracts entered into will contain the following equal opportunity clause:

During the performance of this contract, the contractor agrees as follows:

(1) The contractor will not discriminate against any employee or applicant for employment because of race, color, religion, sex, sexual orientation, gender identity, or national origin. The contractor will take affirmative action to ensure that applicants are employed, and that employees are treated during employment without regard to their race, color, religion, sex, sexual orientation, gender identity, or national origin. Such action shall include, but not be limited to the following:

Employment, upgrading, demotion, or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. The contractor agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided setting forth the provisions of this nondiscrimination clause.

(2) The contractor will, in all solicitations or advertisements for employees placed by or on behalf of the contractor, state that all qualified applicants will receive consideration for employment without regard to race, color, religion, sex, sexual orientation, gender identity, or national origin.

(3) The contractor will not discharge or in any other manner discriminate against any employee or applicant for employment because such employee or applicant has inquired about, discussed, or disclosed the compensation of the employee or applicant or another employee or applicant. This provision shall not apply to instances in which an employee who has access to the compensation information of other employees or applicants as a part of such employee's essential job functions discloses the compensation of such other employees or applicants to individuals who do not otherwise have access to such information, unless such disclosure is in response to a formal complaint or charge, in furtherance of an investigation, proceeding, hearing, or action, including an

CDBG REQUIREMENTS

investigation conducted by the employer, or is consistent with the contractor's legal duty to furnish information.

(4) The contractor will send to each labor union or representative of workers with which he has a collective bargaining agreement or other contract or understanding, a notice to be provided advising the said labor union or workers' representatives of the contractor's commitments under this section, and shall post copies of the notice in conspicuous places available to employees and applicants for employment.

(5) The contractor will comply with all provisions of Executive Order 11246 of September 24, 1965, and of the rules, regulations, and relevant orders of the Secretary of Labor.

(6) The contractor will furnish all information and reports required by Executive Order 11246 of September 24, 1965, and by rules, regulations, and orders of the Secretary of Labor, or pursuant thereto, and will permit access to his books, records, and accounts by the administering agency and the Secretary of Labor for purposes of investigation to ascertain compliance with such rules, regulations, and orders.

(7) In the event of the contractor's noncompliance with the nondiscrimination clauses of this contract or with any of the said rules, regulations, or orders, this contract may be canceled, terminated, or suspended in whole or in part and the contractor may be declared ineligible for further Government contracts or federally assisted construction contracts in accordance with procedures authorized in Executive Order 11246 of September 24, 1965, and such other sanctions may be imposed and remedies invoked as provided in Executive Order 11246 of September 24, 1965, or by rule, regulation, or order of the Secretary of Labor, or as otherwise provided by law.

(8) The contractor will include the portion of the sentence immediately preceding paragraph (1) and the provisions of paragraphs (1) through (8) in every subcontract or purchase order unless exempted by rules, regulations, or orders of the Secretary of Labor issued pursuant to section 204 of Executive Order 11246 of September 24, 1965, so that such provisions will be binding upon each subcontractor or vendor. The contractor will take such action with respect to any subcontract or purchase order as the administering agency may direct as a means of enforcing such provisions, including sanctions for noncompliance.

Reporting Requirements

Bidder and any proposed subcontractor shall comply with the filing requirements of 41 CFR §60-1.7 by filing Standard Form 100 (EEO-1) <u>only if</u> (1) the bidder has 50 or more employees; and (2) the contract value will be greater than \$50,000.

Bidder and any proposed subcontractor shall complete the Affirmative Action Program Certification of Compliance <u>only if</u> (1) the bidder has 50 or more employees; (2) the work is for non-construction supply or service; and (2) the contract value will be greater than \$50,000.

Elimination of Segregated Facilities

Bidder shall ensure that facilities provided for employees are provided in such a manner that segregation on the basis of race, color, religion, sex, sexual orientation, gender identity, or national origin cannot result. The contractor may neither require such segregated use by written or oral policies nor tolerate such use by employee custom. The contractor's obligation extends further to ensuring that its employees are not assigned to perform their services at any location, under the contractor's control, where the facilities are segregated. This obligation extends to all contracts containing the equal opportunity clause regardless of the amount of the contract. The term "facilities," as used in this section, means waiting rooms, work areas, restaurants and other eating areas, time clocks, restrooms, wash rooms, locker rooms, and other storage or dressing areas, parking lots, drinking fountains, recreation or entertainment areas, transportation, and housing provided for employees; Provided, That separate or single-user restrooms and necessary dressing or sleeping areas shall be provided to assure privacy between the sexes.
CDBG REQUIREMENTS

Suspension and Debarment

By submitting a bid/proposal under this solicitation, the Bidder certifies that neither it nor any person or firm who has an interest in the Bidder's firm is a person or firm ineligible to be awarded Government contracts, contracts or participate in programs pursuant to 2 CFR Part 180.

The Bidder agrees that no part of this work shall be subcontracted to any person or parties listed on the government-wide Excluded Parties List System in the System for Award Management (SAM), in accordance with the OMB guidelines at 2 CFR 180 that implement Executive Orders 12549 (3 CFR Part 1986 Comp., p. 189) and 12689 (3 CFR Part 1989 Comp., p. 235). In addition, bidders shall ensure that contractual language in third party contracts enforce this provision.

SUBCONTRACTING WITH SMALL AND MINORITY FIRMS, WOMEN'S BUSINESS ENTERPRISES, AND LABOR SURPLUS AREA FIRMS

The Bidder shall take the following steps to ensure that, whenever possible, subcontracts are awarded to small business firms, minority firms, women's business enterprises, and labor surplus area firms described in Executive Orders 11625, 12432 and 12138, and 2 CFR part 200:

- 1. Placing qualified small and minority businesses and women's business enterprises on solicitation lists;
- 2. Assuring that small and minority businesses, and women's business enterprises are solicited whenever they are potential sources;
- 3. Dividing total requirements, when economically feasible, into smaller tasks or quantities to permit maximum participation by small and minority business, and women's business enterprises;
- 4. Establishing delivery schedules, where the requirement permits, which encourage participation by small and minority business, and women's business enterprises; and
- 5. Using the services and assistance of the U.S. Small Business Administration, the Minority Business Development Agency of the U.S. Department of Commerce, and State and local governmental small business agencies.

Definitions:

Disadvantaged business enterprise (DBE) means an entity owned or controlled by a socially and economically disadvantaged individual as described by Public Law 102-389 (42 U.S.C. 4370d) or an entity owned and controlled by a socially and economically disadvantaged individual as described by Title X of the Clean Air Act Amendments of 1990 (42 U.S.C. 7601 note); a Small Business Enterprise (SBE); a Small Business in a Rural Area (SBRA); or a Labor Surplus Area Firm (LSAF), a Historically Underutilized Business (HUB) Zone Small Business Concern, or a concern under a successor program.

Labor surplus area firm (LSAF) means a concern that together with its first-tier subcontractors will perform substantially in labor surplus areas (as identified by the Department of Labor in accordance with 20 CFR part 654). Performance is substantially in labor surplus areas if the costs incurred under the contract on account of manufacturing, production or performance of appropriate services in labor surplus areas exceed 50 percent of the contract price.

Minority business enterprise (MBE) means a business enterprise that is at least 51 percent owned by a minority group or groups including: a Disadvantaged Business Enterprise (DBE) other than a Small Business Enterprise (SBE), a Labor Surplus Area Firm (LSAF), a Small Business in Rural Areas (SBRA), or a Women's Business Enterprise (WBE).

Small business, small business concern or small business enterprise (SBE) means a concern, including its affiliates, that is independently owned and operated, not dominant in the field of operation in which it is bidding, and qualified as a small business under the criteria and size standards in 13 CFR part 121.

Women's business enterprise (WBE) means a business concern which is at least 51% owned or controlled by women. Determination of ownership by a married woman in a community property jurisdiction will not be affected by her husband's 50 percent interest in her share. Similarly, a business concern which is more than 50 percent owned by a married man will not become a qualified WBE by virtue of his wife's 50 percent interest in his share.

PROCUREMENT OF RECOVERED MATERIALS

Bidder must comply with section 6002 of the Solid Waste Disposal Act, as amended by the Resource Conservation and Recovery Act. The requirements of Section 6002 include procuring only items designated in guidelines of the Environmental Protection Agency (EPA) at 40 CFR part 247 that contain the highest percentage of recovered materials practicable, consistent with maintaining a satisfactory level of competition, where the purchase price of the item exceeds \$10,000 or the value of the quantity acquired during the preceding fiscal year exceeded \$10,000; procuring solid waste management services in a manner that maximizes energy and resource recovery; and establishing an affirmative procurement program for procurement of recovered materials identified in the EPA guidelines.

ENERGY EFFICIENCY

Bidder will comply with mandatory standards and policies relating to energy efficiency which are contained in the state energy conservation plan issued in compliance with the Energy Policy **and Conservation (42 U.S.C. 6201).**

ANTI-LOBBYING

For contracts in excess of \$100,000, the Bidder certifies that it will not and has not used Federal appropriated funds to pay any person or organization for influencing or attempting to influence an officer or employee of any agency, a member of Congress, officer or employee of Congress, or an employee of a member of Congress in connection with obtaining any Federal contract, grant or any other award covered by 31 U.S.C. 1352. Further, the contractor agrees to disclose any lobbying with non-Federal funds that takes place in connection with obtaining any Federal award.

CLEAN AIR ACT

For contracts in excess of \$150,000, the Bidder agrees to comply with all applicable standards, orders or regulations issued pursuant to the Clean Air Act (42 U.S.C. 7401-7671q) and the Federal Water Pollution Control Act as amended (33 U.S.C. 1251-1387). Violations must be reported to the Federal awarding agency and the Regional Office of the Environmental Protection Agency (EPA).

ACCESSIBILITY REQUIREMENTS

All buildings and residential structures designed, constructed, or altered is subject to the requirements of the Architectural Barriers Act of 1968 (42 U.S.C. 4151-4157) and shall comply with the Uniform Federal Accessibility Standards (appendix A to 24 CFR part 40 for residential structures, and appendix A to 41 CFR part 101-19, subpart 101-19.6, for general type buildings).

All facilities designed and constructed shall be readily accessible to and usable by individuals with disabilities. Further, the ADA requires the removal of architectural barriers and communication barriers that are structural in nature in existing facilities, where such removal is readily achievable— that is, easily accomplishable and able to be carried out without much difficulty or expense. (42 U.S.C. 12131; 47 U.S.C. 155, 201, 218, and 225)

FEDERAL LABOR STANDARDS

Davis Bacon

Bidder will comply with all applicable standards, orders, or requirements of the Davis Bacon Act, as amended (40 U.S.C. 3141-3148) as supplemented by Department of Labor regulations (29 CFR part 5). In accordance with the statute, contractors must be required to pay wages to laborers and mechanics at a rate not less than the prevailing wages specified in a wage determination made by

CDBG REQUIREMENTS

the Secretary of Labor. In addition, contractors must be required to pay wages not less than once a week. (State prevailing wage rates required in these Specifications may be higher. Contractor and subcontractors shall pay whichever rate is higher.)

Copeland Anti-Kickback

Bidder will comply with the Copeland "Anti-Kickback" Act (40 U.S.C. 3145) as supplemented in Department of Labor regulations (29 CFR part 3). The Act provides that each contractor or subrecipient must be prohibited from inducing, by any means, any person employed in the construction, completion, or repair of public work, to give up any part of the compensation to which it is otherwise entitled. The Bidder and each Subcontractor must submit a weekly statement on the wages paid to each employee performing on covered work during the prior week.

CONTRACT WORK HOURS AND SAFETY STANDARDS

Bidder will comply with Sections 103 and 107 of the Contract Work Hours and Safety Standards Act (40 U.S.C. 327A 330) as supplemented by Department of Labor regulations (29 CFR part 5). The contractor must compute the wages of every mechanic and laborer on the basis of a standard work week of 40 hours. Work in excess of the standard work week is permissible provided that the worker is compensated at a rate of not less than one and a half times the basic rate of pay for all hours worked in excess of 40 hours in the work week. The requirements of 40 U.S.C. 3704 are applicable to construction work and provide that no laborer or mechanic must be required to work in surroundings or under working conditions which are unsanitary, hazardous, or dangerous. These requirements do not apply to the purchases of supplies or materials or articles ordinarily available on the open market, or contracts for transportation or transmission of intelligence.

SECTION 3

Bidder agrees to comply with HUD's regulations in 24 CFR part 135, which implement Section 3. As evidenced by the submission of their bid, the Bidder certifies that they are under no contractual or other impediment that would prevent them from complying with the part 135 regulations.

Definitions

For purpose of this section, Section 3 residents are defined as: 1) residents of public housing; or 2) individuals that reside in the metropolitan area or non-metropolitan county in which the Section 3 covered assistance is expended and meet the definition of a low- or very low-income person as defined by HUD.

For purpose of this section, Section 3 business concerns are defined as one of the following: 1) businesses that are 51 percent or more owned by Section 3 residents; 2) businesses whose permanent, full-time employees include persons, at least 30 percent of whom are current Section 3 residents or were Section 3 residents within 3 years of the date of first employment with the business concern; or 3) businesses that provide evidence of a commitment to subcontract in excess of 25 percent of the dollar award of all subcontracts to be awarded to business concerns that meet the qualifications set forth in the two previous categories.

CONTRACT CLAUSE

- A. The work to be performed under this contract is subject to the requirements of section 3 of the Housing and Urban Development Act of 1968, as amended, 12 U.S.C. 1701u (section 3). The purpose of section 3 is to ensure that employment and other economic opportunities generated by HUD assistance or HUD-assisted projects covered by section 3, shall, to the greatest extent feasible, be directed to low- and very low-income persons, particularly persons who are recipients of HUD assistance for housing.
- B. The parties to this contract agree to comply with HUD's regulations in 24 CFR part 135, which implement section 3. As evidenced by their execution of this contract, the parties to this contract certify that they are under no contractual or other impediment that would prevent them from complying with the part 135 regulations.

CDBG REQUIREMENTS

- C. The contractor agrees to send to each labor organization or representative of workers with which the contractor has a collective bargaining agreement or other understanding, if any, a notice advising the labor organization or workers' representative of the contractor's commitments under this section 3 clause, and will post copies of the notice in conspicuous places at the work site where both employees and applicants for training and employment positions can see the notice. The notice shall describe the section 3 preference, shall set forth minimum number and job titles subject to hire, availability of apprenticeship and training positions, the qualifications for each; and the name and location of the person(s) taking applications for each of the positions; and the anticipated date the work shall begin.
- D. The contractor agrees to include this section 3 clause in every subcontract subject to compliance with regulations in 24 CFR part 135, and agrees to take appropriate action, as provided in an applicable provision of the subcontract or in this section 3 clause, upon a finding that the subcontractor is in violation of the regulations in 24 CFR part 135. The contractor will not subcontract with any subcontractor where the contractor has notice or knowledge that the subcontractor has been found in violation of the regulations in 24 CFR part 135.
- E. The contractor will certify that any vacant employment positions, including training positions, that are filled (1) after the contractor is selected but before the contract is executed, and (2) with persons other than 3.7
- F. Those to whom the regulations of 24 CFR part 135 require employment opportunities to be directed, were not filled to circumvent the contractor's obligations under 24 CFR part 135.
- G. Noncompliance with HUD's regulations in 24 CFR part 135 may result in sanctions, termination of this contract for default, and debarment or suspension from future HUD assisted contracts.
- H. With respect to work performed in connection with section 3 covered Indian housing assistance, section 7(b) of the Indian Self-Determination and Education Assistance Act (25 U.S.C. 450e) also applies to the work to be performed under this contract. Section 7(b) requires that to the greatest extent feasible (i) preference and opportunities for training and employment shall be given to Indians, and (ii) preference in the award of contracts and subcontracts shall be given to Indian organizations and Indian-owned Economic Enterprises. Parties to this contract that are subject to the provisions of section 3 and section 7(b) agree to comply with section 3 to the maximum extent feasible, but not in derogation of compliance with section 7(b).

CDBG REQUIREMENTS Insert Current Applicable Federal Wage Rate

"General Decision Number: CA20230018 06/23/2023

Superseded General Decision Number: CA20220018

State: California

Construction Types: Building, Heavy (Heavy and Dredging) and Highway

Counties: Alameda, Calaveras, Contra Costa, Fresno, Kings, Madera, Mariposa, Merced, Monterey, San Benito, San Francisco, San Joaquin, San Mateo, Santa Clara, Santa Cruz, Stanislaus and Tuolumne Counties in California.

BUILDING CONSTRUCTION PROJECTS; DREDGING PROJECTS (does not include hopper dredge work); HEAVY CONSTRUCTION PROJECTS (does not include water well drilling); HIGHWAY CONSTRUCTION PROJECTS

Note: Contracts subject to the Davis-Bacon Act are generally required to pay at least the applicable minimum wage rate required under Executive Order 14026 or Executive Order 13658. Please note that these Executive Orders apply to covered contracts entered into by the federal government that are subject to the Davis-Bacon Act itself, but do not apply to contracts subject only to the Davis-Bacon Related Acts, including those set forth at 29 CFR 5.1(a)(2)-(60).

If the contract is entered into on or after January 30, 2022, or the contract is renewed or extended (e.g., an option is exercised) on or after January 30, 2022:	 Executive Order 14026 generally applies to the contract. The contractor must pay all covered workers at least \$16.20 per hour (or the applicable wage rate listed on this wage determination, if it is higher) for all hours spent performing on the contract in 2023.
If the contract was awarded on or between January 1, 2015 and January 29, 2022, and the contract is not renewed or extended on or after January 30, 2022:	

The applicable Executive Order minimum wage rate will be adjusted annually. If this contract is covered by one of the Executive Orders and a classification considered necessary for performance of work on the contract does not appear on this wage determination, the contractor must still submit a conformance request.

Additional information on contractor requirements and worker

protections under the Executive Orders is available at http://www.dol.gov/whd/govcontracts.

Modification Number	Publication Date
0	01/06/2023
1	01/13/2023
2	02/10/2023
3	02/24/2023
4	03/03/2023
5	03/10/2023
6	03/17/2023
7	03/17/2023
8	03/31/2023
9	04/07/2023
10	04/07/2023
11	04/21/2023
12	06/02/2023

ASBE0016-004 01/01/2021

AREA 1: CALAVERAS, FRESNO, KINGS, MADERA, MARIPOSA, MERCED, MONTEREY, SAN BENITO, SAN JOAQUIN, SANTA CRUZ, STANISLAUS & TOULMNE COUNTIES

AREA 2: ALAMEDA, CONTRA COSTA, SAN FRANCISCO, SAN MATEO & SANTA CLARA COUNTIES

Rates Fringes

Asbestos Removal worker/hazardous material handler (Includes preparation, wetting, stripping, removal, scrapping, vacuuming, bagging and disposing of all insulation materials from mechanical systems, whether they contain asbestos or not) Area 1.....\$ 30.45 10.60 Area 2.....\$ 36.53 9.27

ASBE0016-008 02/01/2023

AREA 1: ALAMEDA, CONTRA COSTA, MONTEREY, SAN BENITO, SAN FRANCISCO, SAN MATEO, SANTA CLARA, & SANTA CRUZ

AREA 2: CALAVERAS, COLUSA, FRESNO, KINGS, MADERA, MARIPOSA, MERCED, SAN JOAQUIN, STANISLAUS, & TUOLUMNE

RatesFringesAsbestos Workers/Insulator
(Includes the application of
all insulating materials,
Protective Coverings,
Coatings, and Finishes to all
types of mechanical systems)
Area 1.....\$ 80.91
Area 2....\$ 62.2623.82
23.82

6/28/23, 1:13 PM SAM.gov -----BOIL0549-001 01/01/2021 AREA 1: ALAMEDA, CONTRA COSTA, SAN FRANCISCO, SAN MATEO & SANTA CLARA COUNTIES AREA 2: REMAINING COUNTIES Rates Fringes BOILERMAKER Area 1.....\$ 49.62 41.27 Area 2.....\$ 45.60 38.99 _____ BRCA0003-001 08/01/2022 Fringes Rates MARBLE FINISHER.....\$ 39.20 18.31 BRCA0003-003 08/01/2022 Rates Fringes MARBLE MASON.....\$ 56.98 28.54 _____ BRCA0003-005 05/01/2022 Rates Fringes BRICKLAYER (1) Fresno, Kings, Madera, Mariposa, Merced....\$ 47.88 23.29 (7) San Francisco, San Mateo.....\$ 53.69 26.03 (8) Alameda, Contra Costa, San Benito, Santa Clara.....\$ 53.61 23.81 (9) Calaveras, San Joaquin, Stanislaus, Toulumne.....\$ 45.12 21.55 (16) Monterey, Santa Cruz...\$ 50.78 25.42 BRCA0003-008 07/01/2022 Rates Fringes TERRAZZO FINISHER.....\$ 41.93 18.98 TERRAZZO WORKER/SETTER.....\$ 56.84 27.53 _____ BRCA0003-011 04/01/2022 AREA 1: Alameda, Contra Costa, Monterey, San Benito, San Francisco, San Mateo, Santa Clara, Santa Cruz

AREA 2: Calaveras, San Joaquin, Stanislaus, Tuolumne

AREA 3: Fresno, Kings, Madera, Mariposa, Merced

Rates Fringes

TILE FI	NISHER		
Ar	ea 1\$	33.86	17.54

0/20/23,	1.13 PM		SAW.gov
Tilo	Area 2 Area 3		17.67 16.80
TIE	Layer Area 1 Area 2 Area 3	\$ 50.66	20.50 20.40 19.92
CARI	20022-001 07/01/2021		
San I	Francisco County		
		Rates	Fringes
Carpe	enters Bridge Builder/Highway		
	Carpenter	\$ 54.85	31.49
	Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold &		
	Steel Shoring Erector, Saw		
	FilerJourneyman Carpenter		31.49 31.49
	Millwright	\$ 54.95	33.08
	20034-001 07/01/2021		
		Rates	Fringes
Dive	n		
	Assistant Tender, ROV		
	Tender/Technician Diver standby		34.69 34.69
	Diver Tender		34.69
	Diver wet		34.69
	Manifold Operator (mixed gas)	¢ 6/ 51	34.69
	Manifold Operator (Standby).		34.69
050 1 101 1 151 1	H PAY (Surface Diving): to 100 ft \$2.00 per foot to 150 ft \$3.00 per foot to 220 ft \$4.00 per foot ftdeeper \$5.00 per foot		
The sat pre cor	RATION DIVING: e standby rate shall apply un turation diving rate applies essure continuously until wor mplete. The diver rate shall urs.	when divers are k task and decor	under npression are
Whe or fo: ent nec	NG IN ENCLOSURES: ere it is necessary for Diver other enclosures where there llowing premium shall be paid trance 26 feet to 300 feet: cessary for a diver to enter closure less than 48"" in hei .00 per foot.	e is no vertical l: Distance trav \$1.00 per foot. any pipe, tunne:	ascent, the veled from When it is l or other
Emp wit are	IN COMBINATION OF CLASSIFICA ployees working in any combin thin the diving crew (except e paid in the classification at shift.	ation of classi dive supervisor) in a shift

6/28/23, 1:13 PM

_____ CARP0034-003 07/01/2021 Rates Fringes Piledriver.....\$ 54.10 34.69 _____ _ _ _ _ _ _ _ _ _ _ _ _ CARP0035-007 07/01/2020 AREA 1: Alameda, Contra Costa, San Francisco, San Mateo, Santa Clara counties AREA 2: Monterey, San Benito, Santa Cruz Counties AREA 3: Calaveras, Fresno, Kings, Madera, Mariposa, Merced, San Joaquin, Stanislaus, Tuolumne Counties Rates Fringes Modular Furniture Installer Area 1 Installer.....\$ 28.76 22.53 Lead Installer.....\$ 32.21 23.03 Master Installer.....\$ 36.43 23.03 Area 2 Installer.....\$ 26.11 22.53 Lead Installer.....\$ 29.08 23.03 Master Installer.....\$ 32.71 23.03 Area 3 Installer.....\$ 25.16 22.53 Lead Installer.....\$ 27.96 23.03 Master Installer.....\$ 31.38 23.03 -----CARP0035-008 08/01/2020 AREA 1: Alameda, Contra Costa, San Francisco, San Mateo, Santa Clara counties AREA 2: Monterey, San Benito, Santa Cruz Counties AREA 3: San Joaquin AREA 4: Calaveras, Fresno, Kings, Madera, Mariposa, Merced, Stanislaus, Tuolumne Counties Rates Fringes Drywall Installers/Lathers: Area 1....\$ 52.65 31.26 Area 2....\$ 46.77 31.26 Area 3.....\$ 47.27 31.26 Area 4.....\$ 45.92 31.26 Drywall Stocker/Scrapper Area 1.....\$ 26.33 18.22 Area 2.....\$ 23.39 18.22 Area 3.....\$ 23.64 18.22 Area 4.....\$ 22.97 18.22 _____

CARP0152-001 07/01/2020

Contra Costa County

28/23, 1:13 PM		SAM.gov
R	ates F	ringes
Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold &	52.65	30.82
Steel Shoring Erector, Saw Filer\$ Journeyman Carpenter\$ Millwright\$	52.65	30.82 30.82 32.41
CARP0152-002 07/01/2020		
San Joaquin County		
R	ates F	ringes
Carpenters Bridge Builder/Highway		
Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold & Steel Shoring Erector, Saw	52.65	30.82
Filer\$		30.82
Journeyman Carpenter		30.82 32.41
CARP0152-004 07/01/2020		
CARP0152-004 07/01/2020 Calaveras, Mariposa, Merced, Stanis		ne Counties ringes
CARP0152-004 07/01/2020 Calaveras, Mariposa, Merced, Stanis		
CARP0152-004 07/01/2020 Calaveras, Mariposa, Merced, Stanis R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold &	ates F	
CARP0152-004 07/01/2020 Calaveras, Mariposa, Merced, Stanis R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold & Steel Shoring Erector, Saw	ates F 52.65	ringes 30.82
CARP0152-004 07/01/2020 Calaveras, Mariposa, Merced, Stanis R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold &	ates F 52.65 45.57	ringes
CARP0152-004 07/01/2020 Calaveras, Mariposa, Merced, Stanis R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold & Steel Shoring Erector, Saw Filer\$	ates F 52.65 45.57 45.42 47.92	ringes 30.82 30.82 30.82 30.82 32.41
CARP0152-004 07/01/2020 Calaveras, Mariposa, Merced, Stanis R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold & Steel Shoring Erector, Saw Filer\$ Journeyman Carpenter\$	ates F 52.65 45.57 45.42 47.92	ringes 30.82 30.82 30.82 30.82 32.41
CARP0152-004 07/01/2020 Calaveras, Mariposa, Merced, Stanis R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold & Steel Shoring Erector, Saw Filer\$ Journeyman Carpenter\$	ates F 52.65 45.57 45.42 47.92	ringes 30.82 30.82 30.82 30.82 32.41
CARP0152-004 07/01/2020 Calaveras, Mariposa, Merced, Stanis R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold & Steel Shoring Erector, Saw Filer\$ Journeyman Carpenter\$ Millwright\$ CARP0217-001 07/01/2021 San Mateo County	ates F 52.65 45.57 45.42 47.92	ringes 30.82 30.82 30.82 30.82 32.41
CARP0152-004 07/01/2020 Calaveras, Mariposa, Merced, Stanis R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold & Steel Shoring Erector, Saw Filer\$ Journeyman Carpenter\$ Millwright\$ CARP0217-001 07/01/2021 San Mateo County R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer,	ates F 52.65 45.57 45.42 47.92 ates F	ringes 30.82 30.82 30.82 30.82 32.41
CARP0152-004 07/01/2020 Calaveras, Mariposa, Merced, Stanis R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold & Steel Shoring Erector, Saw Filer\$ Journeyman Carpenter\$ Millwright\$ CARP0217-001 07/01/2021 San Mateo County R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold &	ates F 52.65 45.57 45.42 47.92 ates F	ringes 30.82 30.82 30.82 30.82 32.41
CARP0152-004 07/01/2020 Calaveras, Mariposa, Merced, Stanis R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold & Steel Shoring Erector, Saw Filer\$ Journeyman Carpenter\$ Millwright\$ CARP0217-001 07/01/2021 San Mateo County R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold & Steel Shoring Erector, Saw Filer\$	ates F 52.65 45.57 45.42 47.92 ates F 54.85 55.00	ringes 30.82 30.82 30.82 30.82 32.41 ringes 31.49 31.49
CARP0152-004 07/01/2020 Calaveras, Mariposa, Merced, Stanis R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold & Steel Shoring Erector, Saw Filer\$ Journeyman Carpenter\$ Millwright\$ CARP0217-001 07/01/2021 San Mateo County R Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold & Steel Shoring Erector, Saw	ates F 52.65 45.57 45.42 47.92 ates F 54.85 55.00 54.85	ringes 30.82 30.82 30.82 30.82 32.41

CARP0405-001 07/01/2021

https://sam.gov/wage-determination/CA20230018/14

Santa Clara County

	Rates	Fringes
Carpenters Bridge Builder/Highway		
Carpenter Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold & Steel Shoring Erector, Saw	\$ 54.85	31.49
Filer	\$ 55.00	31.49
Journeyman Carpenter		31.49
Millwright	\$ 54.95	33.08
CARP0405-002 07/01/2021		
San Benito County		
	Rates	Fringes
Carpenters Bridge Builder/Highway		

BLIDGe BUILDEL/HIGHWAY	
Carpenter\$ 54.85	31.49
Hardwood Floorlayer,	
Shingler, Power Saw	
Operator, Steel Scaffold &	
Steel Shoring Erector, Saw	
Filer\$ 49.12	31.49
Journeyman Carpenter\$ 48.97	31.49
Millwright\$ 51.47	33.08
 -	

CARP0505-001 07/01/2021

Santa Cruz County

	Rates	Fringes
Carpenters Bridge Builder/Highway Carpenter Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold a	&	31.49
Steel Shoring Erector, Sau Filer Journeyman Carpenter Millwright	\$ 49.12 \$ 48.97	31.49 31.49 33.08

CARP0605-001 07/01/2021

Monterey County

	Rates	Fringes
Carpenters Bridge Builder/Highway Carpenter\$ Hardwood Floorlayer, Shingler, Power Saw Operator, Steel Scaffold & Steel Shoring Erector, Saw	54.85	31.49
Filer\$ Journeyman Carpenter\$		31.49 31.49

8/23, 1:13 PM Millwright\$	E1 /7	SAM.gov
	51.47	
CARP0701-001 07/01/2021		
resno and Madera Counties		
R	ates	Fringes
Carpenters		
Bridge Builder/Highway		21 40
Carpenter\$ Hardwood Floorlayer,	54.05	31.49
Shingler, Power Saw		
Operator, Steel Scaffold &		
Steel Shoring Erector, Saw		
Filer\$ Journeyman Carpenter\$		31.49 31.49
Millwright\$		33.08
CARP0713-001 07/01/2021		
Alameda County		
R	ates	Fringes
Carpenters		
Bridge Builder/Highway	- 4 . 0 -	24 40
Carpenter\$ Hardwood Floorlayer,	54.85	31.49
Shingler, Power Saw		
Operator, Steel Scaffold &		
Steel Shoring Erector, Saw		
Filer\$		31.49
Journeyman Carpenter\$ Millwright\$		31.49 33.08
CARP1109-001 07/01/2021		
(ings County		
		Fuinces
ĸ	ates	Fringes
Carpenters Bridge Builder/Highway		
Carpenter\$	54.85	31.49
Hardwood Floorlayer,		
Shingler, Power Saw		
Operator, Steel Scaffold & Steel Shoring Erector, Saw		
Filer\$	47.77	31.49
Journeyman Carpenter\$		31.49
Millwright\$	50.12	33.08
ELEC0006-004 12/01/2021		
SAN FRANCISCO COUNTY		
R	ates	Fringes
Sound & Communications		
	48.43	3%+23.15
Installer¢		
Installer\$ Technician\$		3%+23.15
	55.69	

data systems or multiple systems which include control function or power supply; inclusion or exclusion of terminations and testings of conductors determined by their function; excluding fire alarm work when installed in raceways (including wire and cable pulling) and when performed on new or major remodel building projects or jobs for which the conductors for the fire alarm system are installed in conduit; excluding installation of raceway systems, line voltage work, industrial work, life-safety systems (all buildings having floors located more than 75' above the lowest floor level having building access); excluding energy management systems.

FOOTNOTE: Fire alarm work when installed in raceways (including wire and cable pulling), on projects which involve new or major remodel building construction, for which the conductors for the fire alarm system are installed in the conduit, shall be performed by the inside electrician.

ELEC0006-007 01/01/2023

SAN FRANCISCO COUNTY

	Rates	Fringes
ELECTRICIAN	\$ 84.50	3%+40.315
ELEC0100-002 03/01/2023		

FRESNO, KINGS, AND MADERA COUNTIES

	Rates	Fringes	
ELECTRICIAN	\$ 43.50	27.655	

ELEC0100-005 12/01/2022

FRESNO, KINGS, MADERA

	Rates	Fringes
Communications System		
Installer	\$ 38.24	23.80
Technician	\$ 47.80	24.08

SCOPE OF WORK

Includes the installation testing, service and maintenance, of the following systems which utilize the transmission and/or transference of voice, sound, vision and digital for commercial, education, security and entertainment purposes for the following: TV monitoring and surveillance, background-foreground music, intercom and telephone interconnect, inventory control systems, microwave transmission, multi-media, multiplex, nurse call system, radio page, school intercom and sound, burglar alarms, and low voltage master clock systems.

A. SOUND AND VOICE TRANSMISSION/TRANSFERENCE SYSTEMS Background foreground music, Intercom and telephone interconnect systems, Telephone systems Nurse call systems, Radio page systems, School intercom and sound systems, Burglar alarm systems, Low voltage, master clock systems, Multi-media/multiplex systems, Sound and musical

entertainment systems, RF systems, Antennas and Wave Guide,

B. FIRE ALARM SYSTEMS Installation, wire pulling and testing

C. TELEVISION AND VIDEO SYSTEMS Television monitoring and surveillance systems Video security systems, Video entertainment systems, Video educational systems, Microwave transmission systems, CATV and CCTV

D. SECURITY SYSTEMS Perimeter security systems Vibration sensor systems Card access systems Access control systems, Sonar/infrared monitoring equipment

E. COMMUNICATIONS SYSTEMS THAT TRANSMIT OR RECEIVE INFORMATION AND/OR CONTROL SYSTEMS THAT ARE INTRINSIC TO THE ABOVE LISTED SYSTEMS SCADA (Supervisory Control and Data Acquisition) PCM (Pulse Code Modulation) Inventory Control Systems, Digital Data Systems Broadband and Baseband and Carriers Point of Sale Systems, VSAT Data Systems Data Communication Systems RF and Remote Control Systems, Fiber Optic Data Systems

WORK EXCLUDED Raceway systems are not covered (excluding Ladder-Rack for the purpose of the above listed systems). Chases and/or nipples (not to exceed 10 feet) may be installed on open wiring systems. Energy management systems. SCADA (Supervisory Control and Data Acquisition) when not intrinsic to the above listed systems (in the scope). Fire alarm systems when installed in raceways (including wire and cable pulling) shall be performed at the electrician wage rate, when either of the following two (2) conditions apply:

1. The project involves new or major remodel building trades construction.

2. The conductors for the fire alarm system are installed in conduit.

ELEC0234-001 12/26/2022

MONTEREY, SAN BENITO AND SANTA CRUZ COUNTIES

Rates

Fringes

ELECTRICIAN

Zone	A\$	60.91	29.43
Zone	B\$	67.00	29.61

Zone A: All of Santa Cruz, Monterey, and San Benito Counties within 25 air miles of Highway 1 and Dolan Road in Moss Landing, and an area extending 5 miles east and west of Highway 101 South to the San Luis Obispo County Line

Zone B: Any area outside of Zone A

ELEC0234-003 12/01/2021

MONTEREY, SAN BENITO, AND SANTA CRUZ COUNTIES

Rates Fringes

Sound & Communications	
Installer\$ 47.93	24.09
Technician\$ 55.12	24.30

SCOPE OF WORK: Including any data system whose only function is to transmit or receive information; excluding all other data systems or multiple systems which include control function or power supply; inclusion or exclusion of terminations and testings of conductors determined by their function; excluding fire alarm work when installed in raceways (including wire and cable pulling) and when performed on new or major remodel building projects or jobs for which the conductors for the fire alarm system are installed in conduit; excluding installation of raceway systems, line voltage work, industrial work, life-safety systems (all buildings having floors located more than 75' above the lowest floor level having building access); excluding energy management systems.

FOOTNOTE: Fire alarm work when installed in raceways (including wire and cable pulling), on projects which involve new or major remodel building construction, for which the conductors for the fire alarm system are installed in the conduit, shall be performed by the inside electrician.

ELEC0302-001 02/27/2023

CONTRA COSTA COUNTY

	Rates	Fringes
CABLE SPLICER		32.67
ELECTRICIAN	\$ 61.26	32.44

ELEC0302-003 12/01/2022

CONTRA COSTA COUNTY

	Rates	Fringes
Sound & Communications		
Installer	\$ 46.64	25.30
Technician	\$ 53.64	25.51

SCOPE OF WORK: Including any data system whose only function is to transmit or receive information; excluding all other data systems or multiple systems which include control function or power supply; inclusion or exclusion of terminations and testings of conductors determined by their function; excluding fire alarm work when installed in raceways (including wire and cable pulling) and when performed on new or major remodel building projects or jobs for which the conductors for the fire alarm system are installed in conduit; excluding installation of raceway systems, line voltage work, industrial work, life-safety systems (all buildings having floors located more than 75' above the lowest floor level having building access); excluding energy management systems.

FOOTNOTE: Fire alarm work when installed in raceways (including wire and cable pulling), on projects which involve new or major remodel building construction, for which the conductors for the fire alarm system are installed in the conduit, shall be performed by the inside electrician.

ELEC0332-001 06/01/2022

SANTA CLARA COUNTY

	Rates	Fringes
CABLE SPLICER	\$ 92.20	42.316
ELECTRICIAN	\$ 80.17	41.955

FOOTNOTES: Work under compressed air or where gas masks are required, orwork on ladders, scaffolds, stacks, ""Bosun's chairs,"" or other structures and where the workers are not protected by permanent guard rails at a distance of 40 to 60 ft. from the ground or supporting structures: to be paid one and one-half times the straight-time rate of pay. Work on structures of 60 ft. or over (as described above): to be paid twice the straight-time rate of pay.

ELEC0332-003 12/01/2022

SANTA CLARA COUNTY

Rates Fringes

Sound	&	Со	ommunications
	_		

Installer\$	51.18	25.435
Technician\$	58.86	25.666

SCOPE OF WORK: Including any data system whose only function is to transmit or receive information; excluding all other data systems or multiple systems which include control function or power supply; inclusion or exclusion of terminations and testings of conductors determined by their function; excluding fire alarm work when installed in raceways (including wire and cable pulling) and when performed on new or major remodel building projects or jobs for which the conductors for the fire alarm system are installed in conduit; excluding installation of raceway systems, line voltage work, industrial work, life-safety systems (all buildings having floors located more than 75' above the lowest floor level having building access); excluding energy management systems.

FOOTNOTE: Fire alarm work when installed in raceways (including wire and cable pulling), on projects which involve new or major remodel building construction, for which the conductors for the fire alarm system are installed in the conduit, shall be performed by the inside electrician.

ELEC0595-001 06/01/2021

ALAMEDA COUNTY

	Rates	Fringes	
CABLE SPLICER		3%+39.94 3%+39.94	

ELEC0595-002 12/01/2022

	Rates	Fringes
CABLE SPLICER	\$ 53.34	7.75%+25.88
(1) Tunnel work(2) All other work		7.75%+25.88 7.75%+25.88

ELEC0595-006 12/01/2022

ALAMEDA COUNTY

	Rates	Fringes
Sound & Communications		
Installer	\$ 51.18	3%+23.90
Technician	\$ 58.86	3%+23.90

SCOPE OF WORK: Including any data system whose only function is to transmit or receive information; excluding all other data systems or multiple systems which include control function or power supply; inclusion or exclusion of terminations and testings of conductors determined by their function; excluding fire alarm work when installed in raceways (including wire and cable pulling) and when performed on new or major remodel building projects or jobs for which the conductors for the fire alarm system are installed in conduit; excluding installation of raceway systems, line voltage work, industrial work, life-safety systems (all buildings having floors located more than 75' above the lowest floor level having building access); excluding energy management systems.

FOOTNOTE: Fire alarm work when installed in raceways (including wire and cable pulling), on projects which involve new or major remodel building construction, for which the conductors for the fire alarm system are installed in the conduit, shall be performed by the inside electrician.

ELEC0595-008 12/01/2022

CALAVERAS AND SAN JOAQUIN COUNTIES

Rates	Fringes

Communications System	
Installer\$ 40.88	3%+23.90
Technician\$ 47.01	3%+23.90

SCOPE OF WORK: Including any data system whose only function is to transmit or receive information; excluding all other data systems or multiple systems which include control function or power supply; inclusion or exclusion of terminations and testings of conductors determined by their function; excluding fire alarm work when installed in raceways (including wire and cable pulling) and when performed on new or major remodel building projects or jobs for which the conductors for the fire alarm system are installed in conduit; excluding installation of raceway systems, line voltage work, industrial work, life-safety systems (all buildings having floors located more than 75' above the lowest floor level having building access); excluding energy management systems. FOOTNOTE: Fire alarm work when installed in raceways (including wire and cable pulling), on projects which involve new or major remodel building construction, for which the conductors for the fire alarm system are installed in the conduit, shall be performed by the inside electrician.

ELEC0617-001 06/01/2023

SAN MATEO COUNTY

	Rates	Fringes
ELECTRICIAN	.\$ 78.00	45.88
ELEC0617-003 12/01/2022		

SAN MATEO COUNTY

	Rates	Fringes
Sound & Communications		
Installer	\$ 51.18	25.44
Technician	\$ 58.86	25.67

SCOPE OF WORK: Including any data system whose only function is to transmit or receive information; excluding all other data systems or multiple systems which include control function or power supply; inclusion or exclusion of terminations and testings of conductors determined by their function; excluding fire alarm work when installed in raceways (including wire and cable pulling) and when performed on new or major remodel building projects or jobs for which the conductors for the fire alarm system are installed in conduit; excluding installation of raceway systems, line voltage work, industrial work, life-safety systems (all buildings having floors located more than 75' above the lowest floor level having building access); excluding energy management systems.

FOOTNOTE: Fire alarm work when installed in raceways (including wire and cable pulling), on projects which involve new or major remodel building construction, for which the conductors for the fire alarm system are installed in the conduit, shall be performed by the inside electrician.

ELEC0684-001 06/01/2023

MARIPOSA, MERCED, STANISLAUS AND TUOLUMNE COUNTIES

RatesFringesELECTRICIAN......\$ 46.003%+27.68CABLE SPLICER = 110% of Journeyman ElectricianELEC0684-004 12/01/2021

MARIPOSA, MERCED, STANISLAUS AND TUOLUMNE COUNTIES

Rates

Communications System		
Installer	38.24	23.80
Technician	43.98	23.97

SCOPE OF WORK: Including any data system whose only function is to transmit or receive information; excluding all other data systems or multiple systems which include control function or power supply; inclusion or exclusion of terminations and testings of conductors determined by their function; excluding fire alarm work when installed in raceways (including wire and cable pulling) and when performed on new or major remodel building projects or jobs for which the conductors for the fire alarm system are installed in conduit; excluding installation of raceway systems, line voltage work, industrial work, life-safety systems (all buildings having floors located more than 75' above the lowest floor level having building access); excluding energy management systems.

FOOTNOTE: Fire alarm work when installed in raceways (including wire and cable pulling), on projects which involve new or major remodel building construction, for which the conductors for the fire alarm system are installed in the conduit, shall be performed by the inside electrician.

```
.....
```

ELEC1245-001 06/01/2022

	Rates	Fringes
LINE CONSTRUCTION (1) Lineman; Cable splicer. (2) Equipment specialist (operates crawler tractors, commercial motor vehicles, backhoes, trenchers, cranes (50 tons and below), overhead & underground distribution	.\$ 64.40	22.58
line equipment)		21.30
(3) Groundman		20.89
(4) Powderman	.\$ 51.87	18.79
HOLIDAYS: New Year's Day, M.L. Independence Day, Labor Day, V and day after Thanksgiving, Ch	eterans Day, Th	anksgiving Day
ELEV0008-001 01/01/2023		
	Rates	Fringes
ELEVATOR MECHANIC	.\$ 77.61	37.335+a+b
FOOTNOTE: a. PAID VACATION: Employer co rate as vacation pay credit fo years of service, and 6% for 6 b. PAID HOLIDAYS: New Year's D	r employees with months to 5 yea ay, Memorial Day	h more than 5 ars of service. y, Independence

Day, Labor Day, Veterans' Day, Thanksgiving Day, Friday after Thanksgiving, and Christmas Day.

https://sam.gov/wage-determination/CA20230018/14

ENGI0003-001 06/28/2021

""AREA 1"" WAGE RATES ARE LISTED BELOW

""AREA 2"" RECEIVES AN ADDITIONAL $2.00\ \mbox{per}$ Hour Above AREA 1 RATES.

SEE AREA DEFINITIONS BELOW

F	Rates	Fringes
OPERATOR: Power Equipment		
(AREA 1:) GROUP 1\$	53 92	31.54
GROUP 2\$		31.54
GROUP 3\$		31.54
GROUP 4\$		31.54
GROUP 5\$		31.54
GROUP 6\$		31.54
GROUP 7\$		31.54
GROUP 8\$	44.66	31.54
GROUP 8-A\$	42.45	31.54
OPERATOR: Power Equipment		
(Cranes and Attachments -		
AREA 1:)		
GROUP 1		
Cranes\$		31.15
Oiler\$		31.15
Truck crane oiler\$	46.08	31.15
GROUP 2		
Cranes\$		31.15
0iler\$		31.15
Truck crane oiler\$	45.07	31.15
GROUP 3	40.00	24.45
Cranes\$		31.15 31.15
Hydraulic\$ Oiler\$		31.15
Truck crane oiler\$		31.15
GROUP 4	44.03	51.15
Cranes\$	45 76	31.15
OPERATOR: Power Equipment	45.70	51.15
(Piledriving - AREA 1:)		
GROUP 1		
Lifting devices\$	52.64	31.15
0iler\$		31.15
Truck Crane Oiler\$	45.66	31.15
GROUP 2		
Lifting devices\$	50.82	31.15
0iler\$		31.15
Truck Crane Oiler\$	45.41	31.15
GROUP 3		
Lifting devices\$		31.15
Oiler\$		31.15
Truck Crane Oiler\$	45.12	31.15
GROUP 4	47 37	24.45
Lifting devices\$	47.37	31.15
GROUP 5	44 70	21 15
Lifting devices\$ GROUP 6	44.73	31.15
Lifting devices\$	12 50	31.15
OPERATOR: Power Equipment	42.50	21,12
(Steel Erection - AREA 1:)		
GROUP 1		
Cranes\$	53.27	31.15
https://sam.gov/wage-determination/CA20230018/14		
mos //sam nov//wane-neiermination/LA/U/30018/14		

/28/23, 1:13 PM	SAM.gov
Oiler\$ 43.72	31.15
Truck Crane Oiler\$ 45.95	31.15
GROUP 2	
Cranes\$ 51.50	31.15
Oiler\$ 43.45	31.15
Truck Crane Oiler\$ 45.73	31.15
GROUP 3	
Cranes\$ 50.02	31.15
Hydraulic\$ 45.07	31.15
Oiler\$ 43.23	31.15
Truck Crane Oiler\$ 45.46	31.15
GROUP 4	
Cranes\$ 48.00	31.15
GROUP 5	
Cranes\$ 46.70	31.15
OPERATOR: Power Equipment	
(Tunnel and Underground Work	
- AREA 1:)	
SHAFTS, STOPES, RAISES:	24.45
GROUP 1\$ 47.52	31.15
GROUP 1-A\$ 49.99	31.15
GROUP 2\$ 46.26	31.15
GROUP 3\$ 44.93	31.15
GROUP 4\$ 43.79	31.15
GROUP 5\$ 42.65	31.15
UNDERGROUND: GROUP 1\$ 47.42	31.15
GROUP 1	31.15
GROUP 1-A	31.15
GROUP 2	31.15
GROUP 5	31.15
GROUP 4	31.15
UNUUF J	21.12

FOOTNOTE: Work suspended by ropes or cables, or work on a Yo-Yo Cat: \$.60 per hour additional.

POWER EQUIPMENT OPERATOR CLASSIFICATIONS

GROUP 1: Operator of helicopter (when used in erection work); Hydraulic excavator, 7 cu. yds. and over; Power shovels, over 7 cu. yds.

GROUP 2: Highline cableway; Hydraulic excavator, 3-1/2 cu. yds. up to 7 cu. yds.; Licensed construction work boat operator, on site; Power blade operator (finish); Power shovels, over 1 cu. yd. up to and including 7 cu. yds. m.r.c.

GROUP 3: Asphalt milling machine; Cable backhoe; Combination backhoe and loader over 3/4 cu. yds.; Continuous flight tie back machine assistant to engineer or mechanic; Crane mounted continuous flight tie back machine, tonnage to apply; Crane mounted drill attachment, tonnage to apply; Dozer, slope brd; Gradall; Hydraulic excavator, up to 3 1/2 cu. yds.; Loader 4 cu. yds. and over; Long reach excavator; Multiple engine scraper (when used as push pull); Power shovels, up to and including 1 cu. yd.; Pre-stress wire wrapping machine; Side boom cat, 572 or larger; Track loader 4 cu. yds. and over; Wheel excavator (up to and including 750 cu. yds. per hour)

GROUP 4: Asphalt plant engineer/box person; Chicago boom; Combination backhoe and loader up to and including 3/4 cu. yd.; Concrete batch plant (wet or dry); Dozer and/or push

cat; Pull- type elevating loader; Gradesetter, grade checker (GPS, mechanical or otherwise); Grooving and grinding machine; Heading shield operator; Heavy-duty drilling equipment, Hughes, LDH, Watson 3000 or similar; Heavy-duty repairperson and/or welder; Lime spreader; Loader under 4 cu. yds.; Lubrication and service engineer (mobile and grease rack); Mechanical finishers or spreader machine (asphalt, Barber-Greene and similar); Miller Formless M-9000 slope paver or similar; Portable crushing and screening plants; Power blade support; Roller operator, asphalt; Rubber-tired scraper, self-loading (paddle-wheels, etc.); Rubber- tired earthmoving equipment (scrapers); Slip form paver (concrete); Small tractor with drag; Soil stabilizer (P & H or equal); Spider plow and spider puller; Tubex pile rig; Unlicensed constuction work boat operator, on site; Timber skidder; Track loader up to 4 yds.; Tractor-drawn scraper; Tractor, compressor drill combination; Welder; Woods-Mixer (and other similar Pugmill equipment)

GROUP 5: Cast-in-place pipe laying machine; Combination slusher and motor operator; Concrete conveyor or concrete pump, truck or equipment mounted; Concrete conveyor, building site; Concrete pump or pumpcrete gun; Drilling equipment, Watson 2000, Texoma 700 or similar; Drilling and boring machinery, horizontal (not to apply to waterliners, wagon drills or jackhammers); Concrete mixer/all; Person and/or material hoist; Mechanical finishers (concrete) (Clary, Johnson, Bidwell Bridge Deck or similar types); Mechanical burm, curb and/or curb and gutter machine, concrete or asphalt); Mine or shaft hoist; Portable crusher; Power jumbo operator (setting slip-forms, etc., in tunnels); Screed (automatic or manual); Self-propelled compactor with dozer; Tractor with boom D6 or smaller; Trenching machine, maximum digging capacity over 5 ft. depth; Vermeer T-600B rock cutter or similar

GROUP 6: Armor-Coater (or similar); Ballast jack tamper; Boom- type backfilling machine; Assistant plant engineer; Bridge and/or gantry crane; Chemical grouting machine, truck-mounted; Chip spreading machine operator; Concrete saw (self-propelled unit on streets, highways, airports and canals); Deck engineer; Drilling equipment Texoma 600, Hughes 200 Series or similar up to and including 30 ft. m.r.c.; Drill doctor; Helicopter radio operator; Hydro-hammer or similar; Line master; Skidsteer loader, Bobcat larger than 743 series or similar (with attachments); Locomotive; Lull hi-lift or similar; Oiler, truck mounted equipment; Pavement breaker, truck-mounted, with compressor combination; Paving fabric installation and/or laying machine; Pipe bending machine (pipelines only); Pipe wrapping machine (tractor propelled and supported); Screed (except asphaltic concrete paving); Self- propelled pipeline wrapping machine; Tractor; Self-loading chipper; Concrete barrier moving machine

GROUP 7: Ballast regulator; Boom truck or dual-purpose A-frame truck, non-rotating - under 15 tons; Cary lift or similar; Combination slurry mixer and/or cleaner; Drilling equipment, 20 ft. and under m.r.c.; Firetender (hot plant); Grouting machine operator; Highline cableway signalperson; Stationary belt loader (Kolman or similar); Lift slab machine (Vagtborg and similar types); Maginnes internal full slab vibrator; Material hoist (1 drum); Mechanical trench shield; Pavement breaker with or without compressor 6/28/23, 1:13 PM

SAM.gov

combination); Pipe cleaning machine (tractor propelled and supported); Post driver; Roller (except asphalt); Chip Seal; Self-propelled automatically applied concrete curing mahcine (on streets, highways, airports and canals); Self-propelled compactor (without dozer); Signalperson; Slip-form pumps (lifting device for concrete forms); Tie spacer; Tower mobile; Trenching machine, maximum digging capacity up to and including 5 ft. depth; Truck- type loader

GROUP 8: Bit sharpener; Boiler tender; Box operator; Brakeperson; Combination mixer and compressor (shotcrete/gunite); Compressor operator; Deckhand; Fire tender; Forklift (under 20 ft.); Generator; Gunite/shotcrete equipment operator; Hydraulic monitor; Ken seal machine (or similar); Mixermobile; Oiler; Pump operator; Refrigeration plant; Reservoir-debris tug (selfpropelled floating); Ross Carrier (construction site); Rotomist operator; Self-propelled tape machine; Shuttlecar; Self-propelled power sweeper operator (includes vacuum sweeper); Slusher operator; Surface heater; Switchperson; Tar pot firetender; Tugger hoist, single drum; Vacuum cooling plant; Welding machine (powered other than by electricity)

GROUP 8-A: Elevator operator; Skidsteer loader-Bobcat 743 series or smaller, and similar (without attachments); Mini excavator under 25 H.P. (backhoe-trencher); Tub grinder wood chipper

ALL CRANES AND ATTACHMENTS

GROUP 1: Clamshell and dragline over 7 cu. yds.; Crane, over 100 tons; Derrick, over 100 tons; Derrick barge pedestal-mounted, over 100 tons; Self-propelled boom-type lifting device, over 100 tons

GROUP 2: Clamshell and dragline over 1 cu. yd. up to and including 7 cu. yds.; Crane, over 45 tons up to and including 100 tons; Derrick barge, 100 tons and under; Self-propelled boom-type lifting device, over 45 tons; Tower crane

GROUP 3: Clamshell and dragline up to and including 1 cu. yd.; Cranes 45 tons and under; Self-propelled boom-type lifting device 45 tons and under;

GROUP 4: Boom Truck or dual purpose A-frame truck, non-rotating over 15 tons; Truck-mounted rotating telescopic boom type lifting device, Manitex or similar (boom truck) over 15 tons; Truck-mounted rotating telescopic boom type lifting device, Manitex or similar (boom truck) - under 15 tons;

PILEDRIVERS

GROUP 1: Derrick barge pedestal mounted over 100 tons; Clamshell over 7 cu. yds.; Self-propelled boom-type lifting device over 100 tons; Truck crane or crawler, land or barge mounted over 100 tons

GROUP 2: Derrick barge pedestal mounted 45 tons to and

including 100 tons; Clamshell up to and including 7 cu. yds.; Self-propelled boom-type lifting device over 45 tons; Truck crane or crawler, land or barge mounted, over 45 tons up to and including 100 tons; Fundex F-12 hydraulic pile rig GROUP 3: Derrick barge pedestal mounted under 45 tons; Selfpropelled boom-type lifting device 45 tons and under; Skid/scow piledriver, any tonnage; Truck crane or crawler, land or barge mounted 45 tons and under GROUP 4: Assistant operator in lieu of assistant to engineer; Forklift, 10 tons and over; Heavy-duty repairperson/welder

GROUP 5: Deck engineer

GROUP 6: Deckhand; Fire tender

STEEL ERECTORS

GROUP 1: Crane over 100 tons; Derrick over 100 tons; Selfpropelled boom-type lifting device over 100 tons

GROUP 2: Crane over 45 tons to 100 tons; Derrick under 100 tons; Self-propelled boom-type lifting device over 45 tons to 100 tons; Tower crane

GROUP 3: Crane, 45 tons and under; Self-propelled boom-type lifting device, 45 tons and under

GROUP 4: Chicago boom; Forklift, 10 tons and over; Heavy-duty repair person/welder

GROUP 5: Boom cat

- -

TUNNEL AND UNDERGROUND WORK

GROUP 1-A: Tunnel bore machine operator, 20' diameter or more

GROUP 1: Heading shield operator; Heavy-duty repairperson; Mucking machine (rubber tired, rail or track type); Raised bore operator (tunnels); Tunnel mole bore operator

GROUP 2: Combination slusher and motor operator; Concrete pump or pumpcrete gun; Power jumbo operator

GROUP 3: Drill doctor; Mine or shaft hoist

GROUP 4: Combination slurry mixer cleaner; Grouting Machine operator; Motorman

GROUP 5: Bit Sharpener; Brakeman; Combination mixer and compressor (gunite); Compressor operator; Oiler; Pump operator; Slusher operator

AREA DESCRIPTIONS:

POWER EQUIPMENT OPERATORS, CRANES AND ATTACHMENTS, TUNNEL AND UNDERGROUND [These areas do not apply to Piledrivers and

Steel Erectors] AREA 1: ALAMEDA, CALAVERAS, CONTRA COSTA, FRESNO, KINGS, MADERA, MARIPOSA, MERCED, MONTEREY, SAN BENITO, SAN FRANCISCO, SAN JOAQUIN, SAN MATEO, SANTA CLARA, SANTA CRUZ, STANISLAUS, TUOLUMNE AREA 2 -NOTED BELOW THE REMAINING COUNTIES ARE SPLIT BETWEEN AREA 1 AND AREA 2 AS NOTED BELOW: CALAVERAS COUNTY: Area 1: Remainder Area 2: Eastern Part FRESNO COUNTY: Area 1: Remainder Area 2: Eastern Part MADERA COUNTY: Area 1: Remainder Area 2: Eastern Part MARIPOSA COUNTY: Area 1: Remainder Area 2: Eastern Part MONTEREY COUNTY: Area 1: Remainder Area 2: Southwestern part TUOLUMNE COUNTY: Area 1: Remainder Area 2: Eastern Part ENGI0003-008 08/01/2022 Rates Fringes Dredging: (DREDGING: CLAMSHELL & DIPPER DREDGING; HYDRAULIC SUCTION DREDGING:) AREA 1: (1) Leverman.....\$ 55.15 35.46 (2) Dredge Dozer; Heavy duty repairman.....\$ 50.19 35.46 (3) Booster Pump Operator; Deck Engineer; Deck mate; Dredge Tender; Winch Operator....\$ 49.07 35.46 (4) Bargeman; Deckhand; Fireman; Leveehand; Oiler..\$ 45.77 35.46 AREA 2: (1) Leverman.....\$ 57.15 35.46 (2) Dredge Dozer; Heavy duty repairman.....\$ 52.19 35.46 (3) Booster Pump Operator; Deck Engineer; Deck mate; Dredge Tender; Winch Operator....\$ 51.07 35.46 (4) Bargeman; Deckhand; Fireman; Leveehand; Oiler..\$ 47.77 35.46

6/28/23, 1:13 PM

AREA DESCRIPTIONS

AREA 1: ALAMEDA, BUTTE, CONTRA COSTA, KINGS, MARIN, MERCED, NAPA, SACRAMENTO, SAN BENITO, SAN FRANCISCO, SAN JOAQUIN, SAN MATEO, SANTA CLARA, SANTA CRUZ, SOLANO, STANISLAUS, SUTTER, YOLO, AND YUBA COUNTIES AREA 2: MODOC COUNTY THE REMAINGING COUNTIES ARE SPLIT BETWEEN AREA 1 AND AREA 2 AS NOTED BELOW: ALPINE COUNTY: Area 1: Northernmost part Area 2: Remainder CALAVERAS COUNTY: Area 1: Remainder Area 2: Eastern part COLUSA COUNTY: Area 1: Eastern part Area 2: Remainder ELDORADO COUNTY: Area 1: North Central part Area 2: Remainder FRESNO COUNTY: Area 1: Remainder Area 2: Eastern part GLENN COUNTY: Area 1: Eastern part Area 2: Remainder LASSEN COUNTY: Area 1: Western part along the Southern portion of border with Shasta County Area 2: Remainder MADERA COUNTY: Area 1: Except Eastern part Area 2: Eastern part MARIPOSA COUNTY Area 1: Except Eastern part Area 2: Eastern part MONTERREY COUNTY Area 1: Except Southwestern part Area 2: Southwestern part **NEVADA COUNTY:** Area 1: All but the Northern portion along the border of Sierra County Area 2: Remainder PLACER COUNTY: Area 1: Al but the Central portion Area 2: Remainder PLUMAS COUNTY: Area 1: Western portion

Area 2: Remainder SHASTA COUNTY: Area 1: All but the Northeastern corner Area 2: Remainder SIERRA COUNTY: Area 1: Western part Area 2: Remainder SISKIYOU COUNTY: Area 1: Central part Area 2: Remainder SONOMA COUNTY: Area 1: All but the Northwestern corner Area 2: Remainder **TEHAMA COUNTY:** Area 1: All but the Western border with Mendocino & Trinity Counties Area 2: Remainder TRINITY COUNTY: Area 1: East Central part and the Northeastern border with Shasta County Area 2: Remainder TUOLUMNE COUNTY: Area 1: Except Eastern part Area 2: Eastern part _____ ENGI0003-019 06/29/2020 SEE AREA DESCRIPTIONS BELOW Rates Fringes OPERATOR: Power Equipment (LANDSCAPE WORK ONLY) GROUP 1 AREA 1....\$ 39.95 30.28 AREA 2....\$ 41.95 30.28 GROUP 2 AREA 1....\$ 36.35 30.28 AREA 2....\$ 38.35 30.28 GROUP 3 AREA 1.....\$ 31.74 30.28

GROUP DESCRIPTIONS:

GROUP 1: Landscape Finish Grade Operator: All finish grade work regardless of equipment used, and all equipment with a rating more than 65 HP.

30.28

AREA 2....\$ 33.74

GROUP 2: Landscape Operator up to 65 HP: All equipment with a manufacturer's rating of 65 HP or less except equipment covered by Group 1 or Group 3. The following equipment shall be included except when used for finish work as long as manufacturer's rating is 65 HP or less: A-Frame and Winch Truck, Backhoe, Forklift, Hydragraphic Seeder Machine, Roller, Rubber-Tired and Track Earthmoving Equipment, Skiploader, Straw Blowers, and Trencher 31 HP up to 65 HP.

GROUP 3: Landscae Utility Operator: Small Rubber-Tired Tractor, Trencher Under 31 HP.

AREA DESCRIPTIONS:

AREA 1: ALAMEDA, BUTTE, CONTRA COSTA, KINGS, MARIN, MERCED, NAPA, SACRAMENTO, SAN BENITO, SAN FRANCISCO, SAN JOAQUIN, SAN MATEO, SANTA CLARA, SANTA CRUZ, SOLANO, STANISLAUS, SUTTER, YOLO, AND YUBA COUNTIES

AREA 2 - MODOC COUNTY

THE REMAINING COUNTIES ARE SPLIT BETWEEN AREA 1 AND AREA 2 AS NOTED BELOW:

ALPINE COUNTY: Area 1: Northernmost part Area 2: Remainder

CALAVERAS COUNTY: Area 1: Except Eastern part Area 2: Eastern part

COLUSA COUNTY: Area 1: Eastern part Area 2: Remainder

DEL NORTE COUNTY: Area 1: Extreme Southwestern corner Area 2: Remainder

ELDORADO COUNTY: Area 1: North Central part Area 2: Remainder

FRESNO COUNTY Area 1: Except Eastern part Area 2: Eastern part

GLENN COUNTY: Area 1: Eastern part Area 2: Remainder

HUMBOLDT COUNTY: Area 1: Except Eastern and Southwestern parts Area 2: Remainder

LAKE COUNTY: Area 1: Southern part Area 2: Remainder

LASSEN COUNTY: Area 1: Western part along the Southern portion of border with Shasta County Area 2: Remainder

MADERA COUNTY Area 1: Remainder Area 2: Eastern part

MARIPOSA COUNTY Area 1: Remainder Area 2: Eastern part

MENDOCINO COUNTY: Area 1: Central and Southeastern parts Area 2: Remainder MONTEREY COUNTY Area 1: Remainder Area 2: Southwestern part NEVADA COUNTY: Area 1: All but the Northern portion along the border of Sierra County Area 2: Remainder PLACER COUNTY: Area 1: All but the Central portion Area 2: Remainder PLUMAS COUNTY: Area 1: Western portion Area 2: Remainder SHASTA COUNTY: Area 1: All but the Northeastern corner Area 2: Remainder SIERRA COUNTY: Area 1: Western part Area 2: Remainder SISKIYOU COUNTY: Area 1: Central part Area 2: Remainder SONOMA COUNTY: Area 1: All but the Northwestern corner Area 2: Reaminder TEHAMA COUNTY: Area 1: All but the Western border with mendocino & Trinity Counties Area 2: Remainder TRINITY COUNTY: Area 1: East Central part and the Northeaster border with Shasta County Area 2: Remainder TULARE COUNTY; Area 1: Remainder Area 2: Eastern part TUOLUMNE COUNTY: Area 1: Remainder Area 2: Eastern Part _____ IRON0377-001 01/01/2023 ALAMEDA, CONTRA COSTA, SAN MATEO, SANTA CLARA COUNTIES Rates Fringes Ironworkers: Fence Erector.....\$ 41.28 25.66

6/28/23, 1:13 PM SAM.gov Ornamental, Reinforcing and Structural.....\$ 49.88 34.30 PREMIUM PAY: \$6.00 additional per hour at the following locations: China Lake Naval Test Station, Chocolate Mountains Naval Reserve-Niland, Edwards AFB, Fort Irwin Military Station, Fort Irwin Training Center-Goldstone, San Clemente Island, San Nicholas Island, Susanville Federal Prison, 29 Palms - Marine Corps, U.S. Marine Base - Barstow, U.S. Naval Air Facility - Sealey, Vandenberg AFB \$4.00 additional per hour at the following locations: Army Defense Language Institute - Monterey, Fallon Air Base, Naval Post Graduate School - Monterey, Yermo Marine Corps Logistics Center \$2.00 additional per hour at the following locations: Port Hueneme, Port Mugu, U.S. Coast Guard Station - Two Rock _____ IRON0377-003 01/01/2023 SAN FRANCISCO CITY and COUNTY Rates Fringes Ironworkers: Fence Erector.....\$ 41.28 25.66 Ornamental, Reinforcing 34.30 and Structural.....\$ 50.38 PREMIUM PAY: \$6.00 additional per hour at the following locations: China Lake Naval Test Station, Chocolate Mountains Naval Reserve-Niland, Edwards AFB, Fort Irwin Military Station, Fort Irwin Training Center-Goldstone, San Clemente Island, San Nicholas Island, Susanville Federal Prison, 29 Palms - Marine Corps, U.S. Marine Base - Barstow, U.S. Naval Air Facility - Sealey, Vandenberg AFB \$4.00 additional per hour at the following locations: Army Defense Language Institute - Monterey, Fallon Air Base, Naval Post Graduate School - Monterey, Yermo Marine Corps Logistics Center \$2.00 additional per hour at the following locations: Port Hueneme, Port Mugu, U.S. Coast Guard Station - Two Rock IRON0433-005 01/01/2023 **REMAINING COUNTIES**

Rates

IRONWORKER		
Fence Erector\$	41.28	25.66
Ornamental, Reinforcing		
and Structural\$	46.20	34.30

PREMIUM PAY:

\$6.00 additional per hour at the following locations:

China Lake Naval Test Station, Chocolate Mountains Naval Reserve-Niland, Edwards AFB, Fort Irwin Military Station, Fort Irwin Training Center-Goldstone, San Clemente Island, San Nicholas Island, Susanville Federal Prison, 29 Palms - Marine Corps, U.S. Marine Base - Barstow, U.S. Naval Air Facility - Sealey, Vandenberg AFB

\$4.00 additional per hour at the following locations:

Army Defense Language Institute - Monterey, Fallon Air Base, Naval Post Graduate School - Monterey, Yermo Marine Corps Logistics Center

\$2.00 additional per hour at the following locations:

Port Hueneme, Port Mugu, U.S. Coast Guard Station - Two Rock

LAB00067-006 03/04/2023

AREA ""1"" - ALAMEDA, CONTRA COSTA, SAN FRANCISCO, SAN MATEO AND SANTA CLARA COUNTIES

AREA ""2"" - CALAVERAS, FRESNO, KINGS, MADERA, MARIPOSA, MERCED, MONTEREY, SAN BENITO, SAN JOAQUIN, STANISLAUS, AND TUOLUMNE COUNTIES

Rates Fringes

LABORER (ASBESTOS/MOLD/LEAD	
LABORER)	
Area 1\$ 35.25	27.09
Area 2\$ 34.25	27.09

ASBESTOS REMOVAL-SCOPE OF WORK: Site mobilization; initial site clean-up; site preparation; removal of asbestos-containing materials from walls and ceilings; or from pipes, boilers and mechanical systems only if they are being scrapped; encapsulation, enclosure and disposal of asbestos-containing materials by hand or with equipment or machinery; scaffolding; fabrication of temporary wooden barriers; and assembly of decontamination stations.

LAB00073-002 06/27/2022

CALAVERAS AND SAN JOAQUIN COUNTIES

Rates Fringes

LABORER (TRAFFIC CONTROL/LANE CLOSURE) Escort Driver, Flag Person..\$ 35.01

25.49

Traffic	Control	Person	I\$	35.31	25.49
Traffic	Control	Person	II\$	32.81	25.49

TRAFFIC CONTROL PERSON I: Layout of traffic control, crash cushions, construction area and roadside signage.

TRAFFIC CONTROL PERSON II: Installation and removal of temporary/permanent signs, markers, delineators and crash cushions.

LAB00073-003 07/01/2022

SAN JOAQUIN COUNTY

	Rates	Fringes
LABORER		
Mason Tender-Brick	\$ 35.29	25.21
LABO0073-005 07/01/2021		

	I	Rates	Fringes
Tunnel and	Shaft Laborers:		
GROUP	1\$	42.00	25.71
GROUP	2\$	41.77	25.71
GROUP	3\$	41.52	25.71
GROUP	4\$	41.07	25.71
GROUP	5\$	40.53	25.71
Shotcı	rete Specialist\$	42.52	25.71

TUNNEL AND SHAFT CLASSIFICATIONS

GROUP 1: Diamond driller; Groundmen; Gunite and shotcrete nozzlemen

GROUP 2: Rodmen; Shaft work & raise (below actual or excavated ground level)

GROUP 3: Bit grinder; Blaster, driller, powdermen, heading; Cherry pickermen - where car is lifted; Concrete finisher in tunnel; Concrete screedman; Grout pumpman and potman; Gunite & shotcrete gunman & potman; Headermen; High pressure nozzleman; Miner - tunnel, including top and bottom man on shaft and raise work; Nipper; Nozzleman on slick line; Sandblaster - potman, Robotic Shotcrete Placer, Segment Erector, Tunnel Muck Hauler, Steel Form raiser and setter; Timberman, retimberman (wood or steel or substitute materials therefore); Tugger (for tunnel laborer work); Cable tender; Chuck tender; Powderman - primer house

GROUP 4: Vibrator operator, pavement breaker; Bull gang muckers, trackmen; Concrete crew - includes rodding and spreading, Dumpmen (any method)

GROUP 5: Grout crew; Reboundman; Swamper/ Brakeman

.....

LAB00073-007 06/25/2018

CALAVERAS AND SAN JOAQUIN COUNTIES

Rates

Fringes

LABORER (CONSTRUCTION CRAFT LABORERS)				
Construction Specialist				
Group\$ 30.49	23.20			
GROUP 1\$ 29.79	23.20			
GROUP 1-a\$ 30.01	23.20			
GROUP 1-c\$ 29.84	23.20			
GROUP 1-e\$ 30.34	23.20			
GROUP 1-f\$ 30.37	23.20			
GROUP 2\$ 29.64	23.20			
GROUP 3\$ 29.54	23.20			
GROUP 4\$ 23.23	23.20			
See groups 1-b and 1-d under laborer classifications.				
LABORER (GARDENERS,				
HORTICULTURAL & LANDSCAPE				
LABORERS)				
<pre>(1) New Construction\$ 29.54</pre>	23.20			
(2) Establishment Warranty				
Period\$ 23.23	23.20			
LABORER (GUNITE)				
GROUP 1\$ 29.75	22.31			
GROUP 2\$ 29.25	22.31			
GROUP 3\$ 28.66	22.31			
GROUP 4\$ 28.54	22.31			
LABORER (WRECKING)				
GROUP 1\$ 29.79	23.20			
GROUP 2\$ 29.64	23.20			

FOOTNOTES:

Laborers working off or with or from bos'n chairs, swinging scaffolds, belts shall receive \$0.25 per hour above the applicable wage rate. This shall not apply to workers entitled to receive the wage rate set forth in Group 1-a below.

LABORER CLASSIFICATIONS

CONSTRUCTION SPECIALIST GROUP: Asphalt ironer and raker; Chainsaw; Laser beam in connection with laborers' work; Cast-in- place manhole form setter; Pressure pipelayer; Davis trencher - 300 or similar type (and all small trenchers); Blaster; Diamond driller; Multiple unit drill; Hydraulic drill

GROUP 1: Asphalt spreader boxes (all types); Barko, Wacker and similar type tampers; Buggymobile; Caulker, bander, pipewrapper, conduit layer, plastic pipelayer; Certified hazardous waste worker including Leade Abatement; Compactors of all types; Concrete and magnesite mixer, 1/2 yd. and under; Concrete pan work; Concrete sander; Concrete saw; Cribber and/or shoring; Cut granite curb setter; Dri-pak-it machine; Faller, logloader and bucker; Form raiser, slip forms; Green cutter; Headerboard, Hubsetter, aligner, by any method; High pressure blow pipe (1-1/2"" or over, 100 lbs. pressure/over); Hydro seeder and similar type; Jackhammer operator; Jacking of pipe over 12 inches; Jackson and similar type compactor; Kettle tender, pot and worker applying asphalt, lay-kold, creosote, lime, caustic and similar type materials (applying means applying, dipping or handling of such materials); Lagging, sheeting, whaling, bracing, trenchjacking, lagging hammer; Magnesite, epoxyresin, fiberglass, mastic worker (wet or dry); No

joint pipe and stripping of same, including repair of voids; Pavement breaker and spader, including tool grinder; Perma curb; Pipelayer (including grade checking in connection with pipelaying); Precast-manhole setter; Pressure pipe tester; Post hole digger, air, gas and electric; Power broom sweeper; Power tampers of all types (except as shown in Group 2); Ram set gun and stud gun; Riprap stonepaver and rock-slinger, including placing of sacked concrete and/or sand (wet or dry) and gabions and similar type; Rotary scarifier or multiple head concrete chipping scarifier; Roto and Ditch Witch; Rototiller; Sandblaster, pot, gun, nozzle operators; Signalling and rigging; Tank cleaner; Tree climber; Turbo blaster; Vibrascreed, bull float in connection with laborers' work; Vibrator; Hazardous waste worker (lead removal); Asbestos and mold removal worker

GROUP 1-a: Joy drill model TWM-2A; Gardner-Denver model DH143 and similar type drills; Track driller; Jack leg driller; Wagon driller; Mechanical drillers, all types regardless of type or method of power; Mechanical pipe layers, all types regardless of type or method of power; Blaster and powder; All work of loading, placing and blasting of all powder and explosives of whatever type regardless of method used for such loading and placing; High scalers (including drilling of same); Tree topper; Bit grinder

GROUP 1-b: Sewer cleaners shall receive \$4.00 per day above Group 1 wage rates. ""Sewer cleaner"" means any worker who handles or comes in contact with raw sewage in small diameter sewers. Those who work inside recently active, large diameter sewers, and all recently active sewer manholes shal receive \$5.00 per day above Group 1 wage rates.

GROUP 1-c: Burning and welding in connection with laborers' work; Synthetic thermoplastics and similar type welding

GROUP 1-d: Maintenance and repair track and road beds. All employees performing work covered herein shall receive \$.25 per hour above their regular rate for all work performed on underground structures not specifically covered herein. This paragraph shall not be construed to apply to work below ground level in open cut. It shall apply to cut and cover work of subway construction after the temporary cover has been placed.

GROUP 1-e: Work on and/or in bell hole footings and shafts thereof, and work on and in deep footings. (A deep footing is a hole 15 feet or more in depth.) In the event the depth of the footing is unknown at the commencement of excavation, and the final depth exceeds 15 feet, the deep footing wage rate would apply to all employees for each and every day worked on or in the excavation of the footing from the date of inception.

GROUP 1-f: Wire winding machine in connection with guniting or shot crete

GROUP 2: Asphalt shoveler; Cement dumper and handling dry cement or gypsum; Choke-setter and rigger (clearing work); Concrete bucket dumper and chute; Concrete chipping and grinding; Concrete laborer (wet or dry); Driller tender, chuck tender, nipper; Guinea chaser (stake), grout crew; High pressure nozzle, adductor; Hydraulic monitor (over 100 6/28/23, 1:13 PM

SAM.gov

lbs. pressure); Loading and unloading, carrying and hauling of all rods and materials for use in reinforcing concrete construction; Pittsburgh chipper and similar type brush shredders; Sloper; Single foot, hand-held, pneumatic tamper; All pneumatic, air, gas and electric tools not listed in Groups 1 through 1-f; Jacking of pipe - under 12 inches

GROUP 3: Construction laborers, including bridge and general laborer; Dump, load spotter; Flag person; Fire watcher; Fence erector; Guardrail erector; Gardener, horticultural and landscape laborer; Jetting; Limber, brush loader and piler; Pavement marker (button setter); Maintenance, repair track and road beds; Streetcar and railroad construction track laborer; Temporary air and water lines, Victaulic or similar; Tool room attendant (jobsite only)

GROUP 4: Final clean-up work of debris, grounds and building including but not limited to: street cleaner; cleaning and washing windows; brick cleaner (jobsite only); material cleaner (jobsite only). The classification ""material cleaner"" is to be utilized under the following conditions:A: at demolition site for the salvage of the material.B: at the conclusion of a job where the material is to be salvaged and stocked to be reused on another job.

C: for the cleaning of salvage material at the jobsite or temporary jobsite yard.

The material cleaner classification should not be used in the performance of ""form stripping, cleaning and oiling and moving to the next point of erection"".

.....

GUNITE LABORER CLASSIFICATIONS

GROUP 1: Structural Nozzleman

GROUP 2: Nozzleman, Gunman, Potman, Groundman

GROUP 3: Reboundman

GROUP 4: Gunite laborer

WRECKING WORK LABORER CLASSIFICATIONS

GROUP 1: Skilled wrecker (removing and salvaging of sash, windows and materials)

GROUP 2: Semi-skilled wrecker (salvaging of other building materials)

LAB00073-009 07/01/2022

CALAVERAS AND SAN JOAQUIN COUNTIES

Rates	Fringes

LABORER (Plaster Tender)......\$ 38.02 28.25

Work on a swing stage scaffold: \$1.00 per hour additional.

LAB00261-003 06/28/2021

SAN FRANCISCO AND SAN MATEO COUNTIES

Rates	Fringes
-------	---------

LABORER (TRAFFIC CONTROL/LANE

CLOSURE)

Escort Driver, Flag Person\$ 34.48	26.21
Traffic Control Person I\$ 34.78	26.21
Traffic Control Person II\$ 32.28	26.21

TRAFFIC CONTROL PERSON I: Layout of traffic control, crash cushions, construction area and roadside signage.

TRAFFIC CONTROL PERSON II: Installation and removal of temporary/permanent signs, markers, delineators and crash cushions.

LAB00261-005 07/01/2021

SAN FRANCISCO AND SAN MATEO COUNTIES

Rates	Fringes

Tunnel and Shaft Laborers:	5.
----------------------------	----

GF	OUP 1	42.00	25.71
GF	OUP 2	41.77	25.71
GF	OUP 3	41.52	25.71
GF	OUP 4	41.07	25.71
GF	OUP 5	40.53	25.71
Sł	otcrete Specialist	42.52	25.71

TUNNEL AND SHAFT CLASSIFICATIONS

GROUP 1: Diamond driller; Groundmen; Gunite and shotcrete nozzlemen

GROUP 2: Rodmen; Shaft work & raise (below actual or excavated ground level)

GROUP 3: Bit grinder; Blaster, driller, powdermen, heading; Cherry pickermen - where car is lifted; Concrete finisher in tunnel; Concrete screedman; Grout pumpman and potman; Gunite & shotcrete gunman & potman; Headermen; High pressure nozzleman; Miner - tunnel, including top and bottom man on shaft and raise work; Nipper; Nozzleman on slick line; Sandblaster - potman, Robotic Shotcrete Placer, Segment Erector, Tunnel Muck Hauler, Steel Form raiser and setter; Timberman, retimberman (wood or steel or substitute materials therefore); Tugger (for tunnel laborer work); Cable tender; Chuck tender; Powderman - primer house

GROUP 4: Vibrator operator, pavement breaker; Bull gang muckers, trackmen; Concrete crew - includes rodding and spreading, Dumpmen (any method)

GROUP 5: Grout crew; Reboundman; Swamper/ Brakeman

LAB00261-009 06/25/2018

SAN FRANCISCO, AND SAN MATEO COUNTIES

Rates

Fringes

LABORER (CONSTRUCTION CRAFT LABORERS - AREA A:) Construction Specialist	
Group\$ 31.49	23.20
GROUP 1\$ 30.79	23.20
GROUP 1-a\$ 31.01	23.20
GROUP 1-c\$ 30.84	23.20
GROUP 1-e\$ 31.34	23.20
GROUP 1-f\$ 31.37	23.20
GROUP 2\$ 30.64	23.20
GROUP 3\$ 30.54	23.20
GROUP 4\$ 24.23	23.20
See groups 1-b and 1-d under laborer classificati	ons.
LABORER (GARDENERS,	
HORTICULTURAL & LANDSCAPE	
LABORERS - AREA A:)	
<pre>(1) New Construction\$ 30.54</pre>	23.20
(2) Establishment Warranty	
Period\$ 24.23	23.20
LABORER (WRECKING - AREA A:)	
GROUP 1\$ 30.79	23.20
GROUP 2\$ 30.64	23.20
Laborers: (GUNITE - AREA A:)	
GROUP 1\$ 30.75	22.31
GROUP 2\$ 30.25	22.31
GROUP 3\$ 29.66	22.31
GROUP 4\$ 29.54	22.31

FOOTNOTES:

Laborers working off or with or from bos'n chairs, swinging scaffolds, belts shall receive \$0.25 per hour above the applicable wage rate. This shall not apply to workers entitled to receive the wage rate set forth in Group 1-a below.

LABORER CLASSIFICATIONS

CONSTRUCTION SPECIALIST GROUP: Asphalt ironer and raker; Chainsaw; Laser beam in connection with laborers' work; Cast-in- place manhole form setter; Pressure pipelayer; Davis trencher - 300 or similar type (and all small trenchers); Blaster; Diamond driller; Multiple unit drill; Hydraulic drill

GROUP 1: Asphalt spreader boxes (all types); Barko, Wacker and similar type tampers; Buggymobile; Caulker, bander, pipewrapper, conduit layer, plastic pipelayer; Certified hazardous waste worker including Leade Abatement; Compactors of all types; Concrete and magnesite mixer, 1/2 yd. and under; Concrete pan work; Concrete sander; Concrete saw; Cribber and/or shoring; Cut granite curb setter; Dri-pak-it machine; Faller, logloader and bucker; Form raiser, slip forms; Green cutter; Headerboard, Hubsetter, aligner, by any method; High pressure blow pipe (1-1/2"" or over, 100 lbs. pressure/over); Hydro seeder and similar type; Jackhammer operator; Jacking of pipe over 12 inches; Jackson and similar type compactor; Kettle tender, pot and worker applying asphalt, lay-kold, creosote, lime, caustic and similar type materials (applying means applying, dipping or handling of such materials); Lagging, sheeting,

SAM.gov

whaling, bracing, trenchjacking, lagging hammer; Magnesite, epoxyresin, fiberglass, mastic worker (wet or dry); No joint pipe and stripping of same, including repair of voids; Pavement breaker and spader, including tool grinder; Perma curb; Pipelayer (including grade checking in connection with pipelaying); Precast-manhole setter; Pressure pipe tester; Post hole digger, air, gas and electric; Power broom sweeper; Power tampers of all types (except as shown in Group 2); Ram set gun and stud gun; Riprap stonepaver and rock-slinger, including placing of sacked concrete and/or sand (wet or dry) and gabions and similar type; Rotary scarifier or multiple head concrete chipping scarifier; Roto and Ditch Witch; Rototiller; Sandblaster, pot, gun, nozzle operators; Signalling and rigging; Tank cleaner; Tree climber; Turbo blaster; Vibrascreed, bull float in connection with laborers' work; Vibrator; Hazardous waste worker (lead removal); Asbestos and mold removal worker

GROUP 1-a: Joy drill model TWM-2A; Gardner-Denver model DH143 and similar type drills; Track driller; Jack leg driller; Wagon driller; Mechanical drillers, all types regardless of type or method of power; Mechanical pipe layers, all types regardless of type or method of power; Blaster and powder; All work of loading, placing and blasting of all powder and explosives of whatever type regardless of method used for such loading and placing; High scalers (including drilling of same); Tree topper; Bit grinder

GROUP 1-b: Sewer cleaners shall receive \$4.00 per day above Group 1 wage rates. ""Sewer cleaner"" means any worker who handles or comes in contact with raw sewage in small diameter sewers. Those who work inside recently active, large diameter sewers, and all recently active sewer manholes shal receive \$5.00 per day above Group 1 wage rates.

GROUP 1-c: Burning and welding in connection with laborers' work; Synthetic thermoplastics and similar type welding

GROUP 1-d: Maintenance and repair track and road beds. All employees performing work covered herein shall receive \$.25 per hour above their regular rate for all work performed on underground structures not specifically covered herein. This paragraph shall not be construed to apply to work below ground level in open cut. It shall apply to cut and cover work of subway construction after the temporary cover has been placed.

GROUP 1-e: Work on and/or in bell hole footings and shafts thereof, and work on and in deep footings. (A deep footing is a hole 15 feet or more in depth.) In the event the depth of the footing is unknown at the commencement of excavation, and the final depth exceeds 15 feet, the deep footing wage rate would apply to all employees for each and every day worked on or in the excavation of the footing from the date of inception.

GROUP 1-f: Wire winding machine in connection with guniting or shot crete

GROUP 2: Asphalt shoveler; Cement dumper and handling dry cement or gypsum; Choke-setter and rigger (clearing work); Concrete bucket dumper and chute; Concrete chipping and grinding; Concrete laborer (wet or dry); Driller tender,

SAM.gov

chuck tender, nipper; Guinea chaser (stake), grout crew; High pressure nozzle, adductor; Hydraulic monitor (over 100 lbs. pressure); Loading and unloading, carrying and hauling of all rods and materials for use in reinforcing concrete construction; Pittsburgh chipper and similar type brush shredders; Sloper; Single foot, hand-held, pneumatic tamper; All pneumatic, air, gas and electric tools not listed in Groups 1 through 1-f; Jacking of pipe - under 12 inches

GROUP 3: Construction laborers, including bridge and general laborer; Dump, load spotter; Flag person; Fire watcher; Fence erector; Guardrail erector; Gardener, horticultural and landscape laborer; Jetting; Limber, brush loader and piler; Pavement marker (button setter); Maintenance, repair track and road beds; Streetcar and railroad construction track laborer; Temporary air and water lines, Victaulic or similar; Tool room attendant (jobsite only)

GROUP 4: Final clean-up work of debris, grounds and building including but not limited to: street cleaner; cleaning and washing windows; brick cleaner (jobsite only); material cleaner (jobsite only). The classification ""material cleaner"" is to be utilized under the following conditions: A: at demolition site for the salvage of the material.

B: at the conclusion of a job where the material is to be salvaged and stocked to be reused on another job. C: for the cleaning of salvage material at the jobsite or temporary jobsite yard.

The material cleaner classification should not be used in the performance of ""form stripping, cleaning and oiling and moving to the next point of erection"".

GUNITE LABORER CLASSIFICATIONS

GROUP 1: Structural Nozzleman

GROUP 2: Nozzleman, Gunman, Potman, Groundman

GROUP 3: Reboundman

GROUP 4: Gunite laborer

WRECKING WORK LABORER CLASSIFICATIONS

GROUP 1: Skilled wrecker (removing and salvaging of sash, windows and materials)

GROUP 2: Semi-skilled wrecker (salvaging of other building materials)

LAB00261-011 07/01/2022

SAN FRANCISCO AND SAN MATEO COUNTIES:

Rates	Fringes

MASON TENDER, B	3RICK\$	36.05	27.12
-----------------	---------	-------	-------

6/28/23, 1:13 PM SAM.gov FOOTNOTES: Underground work such as sewers, manholes, catch basins, sewer pipes, telephone conduits, tunnels and cut trenches: \$5.00 per day additional. Work in live sewage: \$2.50 per day additional. _____ LAB00261-014 07/01/2022 SAN FRANCISCO AND SAN MATEO COUNTIES: Rates Fringes PLASTER TENDER.....\$ 40.48 30.23 Work on a swing stage scaffold: \$1.00 per hour additional. LAB00270-003 06/28/2021 AREA A: SANTA CLARA AREA B: MONTEREY, SAN BENITO AND SANTA CRUZ COUNTIES Rates Fringes LABORER (TRAFFIC CONTROL/LANE CLOSURE) Escort Driver, Flag Person Area A.....\$ 34.48 26.21 Area B.....\$ 33.48 26.21 Traffic Control Person I Area A....\$ 34.78 26.21 Area B....\$ 33.78 26.21 Traffic Control Person II 26.21 Area A....\$ 32.28

TRAFFIC CONTROL PERSON I: Layout of traffic control, crash cushions, construction area and roadside signage.

26.21

Fringes

Area B.....\$ 31.28

TRAFFIC CONTROL PERSON II: Installation and removal of temporary/permanent signs, markers, delineators and crash cushions.

Rates

LAB00270-004 07/01/2021

Tunn

MONTEREY, SAN BENITO, SANTA CLARA, AND SANTA CRUZ COUNTIES

el and Shaf [.]	t Laborers:		
GROUP 1	\$	42.00	25.71
GROUP 2	\$	41.77	25.71
GROUP 3	\$	41.52	25.71
GROUP 4	\$	41.07	25.71
GROUP 5	\$	40.53	25.71
Shotcrete S	Specialist\$	42.52	25.71

TUNNEL AND SHAFT CLASSIFICATIONS

GROUP 1: Diamond driller; Groundmen; Gunite and shotcrete nozzlemen

GROUP 2: Rodmen; Shaft work & raise (below actual or

SAM.gov

excavated ground level)

GROUP 3: Bit grinder; Blaster, driller, powdermen, heading; Cherry pickermen - where car is lifted; Concrete finisher in tunnel; Concrete screedman; Grout pumpman and potman; Gunite & shotcrete gunman & potman; Headermen; High pressure nozzleman; Miner - tunnel, including top and bottom man on shaft and raise work; Nipper; Nozzleman on slick line; Sandblaster - potman, Robotic Shotcrete Placer, Segment Erector, Tunnel Muck Hauler, Steel Form raiser and setter; Timberman, retimberman (wood or steel or substitute materials therefore); Tugger (for tunnel laborer work); Cable tender; Chuck tender; Powderman - primer house

GROUP 4: Vibrator operator, pavement breaker; Bull gang muckers, trackmen; Concrete crew - includes rodding and spreading, Dumpmen (any method)

GROUP 5: Grout crew; Reboundman; Swamper/ Brakeman

LAB00270-005 07/01/2022

MONTEREY AND SAN BENITO COUNTIES

Rates Fringes

LABORER

Mason Tender-Brick	\$ 35.29	25.21
1 4000000 007 00/00 /0040		

LAB00270-007 06/25/2018

MONTEREY, SAN BENITO, AND SANTA CRUZ, COUNTIES

	Rates	Fringes
LABORER (CONSTRUCTION CRAFT LABORERS - AREA B) Construction Specialist		
Group	.\$ 30.40	23.20
GROUP 1		23.20
GROUP 1-a		23.20
GROUP 1-c		23.20
GROUP 1-e	.\$ 30.34	23.20
GROUP 1-f	.\$ 30.37	23.20
GROUP 2	.\$ 29.64	23.20
GROUP 3	.\$ 29.54	23.20
GROUP 4	.\$ 23.23	23.20
See groups 1-b and 1-d under la	aborer class	ifications.
LABORER (GARDENERS,		
HORTICULTURAL & LANDSCAPE		
LABORERS - AREA B)		
<pre>(1) New Construction</pre>	.\$ 29.54	23.20
(2) Establishment Warranty		
Period	.\$ 23.23	23.20
LABORER (GUNITE - AREA B)		
GROUP 1		22.31
GROUP 2	.\$ 29.25	22.31
GROUP 3	.\$ 28.66	22.31
GROUP 4	.\$ 28.54	22.31
LABORER (WRECKING - AREA B)		
GROUP 1		23.20
GROUP 2	.\$ 29.64	23.20

Laborers working off or with or from bos'n chairs, swinging scaffolds, belts shall receive \$0.25 per hour above the applicable wage rate. This shall not apply to workers entitled to receive the wage rate set forth in Group 1-a below.

LABORER CLASSIFICATIONS

CONSTRUCTION SPECIALIST GROUP: Asphalt ironer and raker; Chainsaw; Laser beam in connection with laborers' work; Cast-in- place manhole form setter; Pressure pipelayer; Davis trencher - 300 or similar type (and all small trenchers); Blaster; Diamond driller; Multiple unit drill; Hydraulic drill

GROUP 1: Asphalt spreader boxes (all types); Barko, Wacker and similar type tampers; Buggymobile; Caulker, bander, pipewrapper, conduit layer, plastic pipelayer; Certified hazardous waste worker including Leade Abatement; Compactors of all types; Concrete and magnesite mixer, 1/2 yd. and under; Concrete pan work; Concrete sander; Concrete saw; Cribber and/or shoring; Cut granite curb setter; Dri-pak-it machine; Faller, logloader and bucker; Form raiser, slip forms; Green cutter; Headerboard, Hubsetter, aligner, by any method; High pressure blow pipe (1-1/2"" or over, 100 lbs. pressure/over); Hydro seeder and similar type; Jackhammer operator; Jacking of pipe over 12 inches; Jackson and similar type compactor; Kettle tender, pot and worker applying asphalt, lay-kold, creosote, lime, caustic and similar type materials (applying means applying, dipping or handling of such materials); Lagging, sheeting, whaling, bracing, trenchjacking, lagging hammer; Magnesite, epoxyresin, fiberglass, mastic worker (wet or dry); No joint pipe and stripping of same, including repair of voids; Pavement breaker and spader, including tool grinder; Perma curb; Pipelayer (including grade checking in connection with pipelaying); Precast-manhole setter; Pressure pipe tester; Post hole digger, air, gas and electric; Power broom sweeper; Power tampers of all types (except as shown in Group 2); Ram set gun and stud gun; Riprap stonepaver and rock-slinger, including placing of sacked concrete and/or sand (wet or dry) and gabions and similar type; Rotary scarifier or multiple head concrete chipping scarifier; Roto and Ditch Witch; Rototiller; Sandblaster, pot, gun, nozzle operators; Signalling and rigging; Tank cleaner; Tree climber; Turbo blaster; Vibrascreed, bull float in connection with laborers' work; Vibrator; Hazardous waste worker (lead removal); Asbestos and mold removal worker

GROUP 1-a: Joy drill model TWM-2A; Gardner-Denver model DH143 and similar type drills; Track driller; Jack leg driller; Wagon driller; Mechanical drillers, all types regardless of type or method of power; Mechanical pipe layers, all types regardless of type or method of power; Blaster and powder; All work of loading, placing and blasting of all powder and explosives of whatever type regardless of method used for such loading and placing; High scalers (including drilling of same); Tree topper; Bit grinder

GROUP 1-b: Sewer cleaners shall receive \$4.00 per day above Group 1 wage rates. ""Sewer cleaner"" means any worker who

handles or comes in contact with raw sewage in small diameter sewers. Those who work inside recently active, large diameter sewers, and all recently active sewer manholes shal receive \$5.00 per day above Group 1 wage rates.

GROUP 1-c: Burning and welding in connection with laborers' work; Synthetic thermoplastics and similar type welding

GROUP 1-d: Maintenance and repair track and road beds. All employees performing work covered herein shall receive \$.25 per hour above their regular rate for all work performed on underground structures not specifically covered herein. This paragraph shall not be construed to apply to work below ground level in open cut. It shall apply to cut and cover work of subway construction after the temporary cover has been placed.

GROUP 1-e: Work on and/or in bell hole footings and shafts thereof, and work on and in deep footings. (A deep footing is a hole 15 feet or more in depth.) In the event the depth of the footing is unknown at the commencement of excavation, and the final depth exceeds 15 feet, the deep footing wage rate would apply to all employees for each and every day worked on or in the excavation of the footing from the date of inception.

GROUP 1-f: Wire winding machine in connection with guniting or shot crete

GROUP 2: Asphalt shoveler; Cement dumper and handling dry cement or gypsum; Choke-setter and rigger (clearing work); Concrete bucket dumper and chute; Concrete chipping and grinding; Concrete laborer (wet or dry); Driller tender, chuck tender, nipper; Guinea chaser (stake), grout crew; High pressure nozzle, adductor; Hydraulic monitor (over 100 lbs. pressure); Loading and unloading, carrying and hauling of all rods and materials for use in reinforcing concrete construction; Pittsburgh chipper and similar type brush shredders; Sloper; Single foot, hand-held, pneumatic tamper; All pneumatic, air, gas and electric tools not listed in Groups 1 through 1-f; Jacking of pipe - under 12 inches

GROUP 3: Construction laborers, including bridge and general laborer; Dump, load spotter; Flag person; Fire watcher; Fence erector; Guardrail erector; Gardener, horticultural and landscape laborer; Jetting; Limber, brush loader and piler; Pavement marker (button setter); Maintenance, repair track and road beds; Streetcar and railroad construction track laborer; Temporary air and water lines, Victaulic or similar; Tool room attendant (jobsite only)

GROUP 4: Final clean-up work of debris, grounds and building including but not limited to: street cleaner; cleaning and washing windows; brick cleaner (jobsite only); material cleaner (jobsite only). The classification ""material cleaner"" is to be utilized under the following conditions:A: at demolition site for the salvage of the material.B: at the conclusion of a job where the material is to be salvaged and stocked to be reused on another job.C: for the cleaning of salvage material at the jobsite or temporary jobsite yard.

The material cleaner classification should not be used in

the performance of ""form stripping, cleaning and oiling and moving to the next point of erection"". _____ GUNITE LABORER CLASSIFICATIONS GROUP 1: Structural Nozzleman GROUP 2: Nozzleman, Gunman, Potman, Groundman GROUP 3: Reboundman GROUP 4: Gunite laborer -----WRECKING WORK LABORER CLASSIFICATIONS GROUP 1: Skilled wrecker (removing and salvaging of sash, windows and materials) GROUP 2: Semi-skilled wrecker (salvaging of other building materials) LAB00270-010 06/25/2018 SANTA CLARA COUNTY Rates Fringes LABORER (CONSTRUCTION CRAFT LABORERS - AREA A:) Construction Specialist Group.....\$ 31.49 23.20 GROUP 1.....\$ 30.79 23.20 GROUP 1-a....\$ 31.01 23.20 GROUP 1-c....\$ 30.84 23.20 GROUP 1-e....\$ 31.34 23.20 GROUP 1-f.....\$ 30.37 23.20 GROUP 2.....\$ 30.64 23.20 GROUP 3.....\$ 30.54 23.20 GROUP 4.....\$ 24.23 23.20 See groups 1-b and 1-d under laborer classifications. LABORER (GARDENERS, HORTICULTURAL & LANDSCAPE LABORERS - AREA A:) (1) New Construction.....\$ 30.54 23.20 (2) Establishment Warranty Period.....\$ 24.23 23.20 LABORER (GUNITE - AREA A:) GROUP 1.....\$ 30.75 22.31 GROUP 2....\$ 30.25 22.31 GROUP 3.....\$ 29.66 22.31 GROUP 4.....\$ 29.54 22.31 LABORER (WRECKING - AREA A:) GROUP 1.....\$ 30.79 23.20 GROUP 2.....\$ 30.64 23.20

FOOTNOTES:

Laborers working off or with or from bos'n chairs, swinging scaffolds, belts shall receive \$0.25 per hour above the applicable wage rate. This shall not apply to workers entitled to receive the wage rate set forth in Group 1-a

LABORER CLASSIFICATIONS

CONSTRUCTION SPECIALIST GROUP: Asphalt ironer and raker; Chainsaw; Laser beam in connection with laborers' work; Cast-in- place manhole form setter; Pressure pipelayer; Davis trencher - 300 or similar type (and all small trenchers); Blaster; Diamond driller; Multiple unit drill; Hydraulic drill

GROUP 1: Asphalt spreader boxes (all types); Barko, Wacker and similar type tampers; Buggymobile; Caulker, bander, pipewrapper, conduit layer, plastic pipelayer; Certified hazardous waste worker including Leade Abatement; Compactors of all types; Concrete and magnesite mixer, 1/2 yd. and under; Concrete pan work; Concrete sander; Concrete saw; Cribber and/or shoring; Cut granite curb setter; Dri-pak-it machine; Faller, logloader and bucker; Form raiser, slip forms; Green cutter; Headerboard, Hubsetter, aligner, by any method; High pressure blow pipe (1-1/2"" or over, 100 lbs. pressure/over); Hydro seeder and similar type; Jackhammer operator; Jacking of pipe over 12 inches; Jackson and similar type compactor; Kettle tender, pot and worker applying asphalt, lay-kold, creosote, lime, caustic and similar type materials (applying means applying, dipping or handling of such materials); Lagging, sheeting, whaling, bracing, trenchjacking, lagging hammer; Magnesite, epoxyresin, fiberglass, mastic worker (wet or dry); No joint pipe and stripping of same, including repair of voids; Pavement breaker and spader, including tool grinder; Perma curb; Pipelayer (including grade checking in connection with pipelaying); Precast-manhole setter; Pressure pipe tester; Post hole digger, air, gas and electric; Power broom sweeper; Power tampers of all types (except as shown in Group 2); Ram set gun and stud gun; Riprap stonepaver and rock-slinger, including placing of sacked concrete and/or sand (wet or dry) and gabions and similar type; Rotary scarifier or multiple head concrete chipping scarifier; Roto and Ditch Witch; Rototiller; Sandblaster, pot, gun, nozzle operators; Signalling and rigging; Tank cleaner; Tree climber; Turbo blaster; Vibrascreed, bull float in connection with laborers' work; Vibrator; Hazardous waste worker (lead removal); Asbestos and mold removal worker

GROUP 1-a: Joy drill model TWM-2A; Gardner-Denver model DH143 and similar type drills; Track driller; Jack leg driller; Wagon driller; Mechanical drillers, all types regardless of type or method of power; Mechanical pipe layers, all types regardless of type or method of power; Blaster and powder; All work of loading, placing and blasting of all powder and explosives of whatever type regardless of method used for such loading and placing; High scalers (including drilling of same); Tree topper; Bit grinder

GROUP 1-b: Sewer cleaners shall receive \$4.00 per day above Group 1 wage rates. ""Sewer cleaner"" means any worker who handles or comes in contact with raw sewage in small diameter sewers. Those who work inside recently active, large diameter sewers, and all recently active sewer manholes shal receive \$5.00 per day above Group 1 wage

SAM.gov

rates.

GROUP 1-c: Burning and welding in connection with laborers' work; Synthetic thermoplastics and similar type welding

GROUP 1-d: Maintenance and repair track and road beds. All employees performing work covered herein shall receive \$.25 per hour above their regular rate for all work performed on underground structures not specifically covered herein. This paragraph shall not be construed to apply to work below ground level in open cut. It shall apply to cut and cover work of subway construction after the temporary cover has been placed.

GROUP 1-e: Work on and/or in bell hole footings and shafts thereof, and work on and in deep footings. (A deep footing is a hole 15 feet or more in depth.) In the event the depth of the footing is unknown at the commencement of excavation, and the final depth exceeds 15 feet, the deep footing wage rate would apply to all employees for each and every day worked on or in the excavation of the footing from the date of inception.

GROUP 1-f: Wire winding machine in connection with guniting or shot crete

GROUP 2: Asphalt shoveler; Cement dumper and handling dry cement or gypsum; Choke-setter and rigger (clearing work); Concrete bucket dumper and chute; Concrete chipping and grinding; Concrete laborer (wet or dry); Driller tender, chuck tender, nipper; Guinea chaser (stake), grout crew; High pressure nozzle, adductor; Hydraulic monitor (over 100 lbs. pressure); Loading and unloading, carrying and hauling of all rods and materials for use in reinforcing concrete construction; Pittsburgh chipper and similar type brush shredders; Sloper; Single foot, hand-held, pneumatic tamper; All pneumatic, air, gas and electric tools not listed in Groups 1 through 1-f; Jacking of pipe - under 12 inches

GROUP 3: Construction laborers, including bridge and general laborer; Dump, load spotter; Flag person; Fire watcher; Fence erector; Guardrail erector; Gardener, horticultural and landscape laborer; Jetting; Limber, brush loader and piler; Pavement marker (button setter); Maintenance, repair track and road beds; Streetcar and railroad construction track laborer; Temporary air and water lines, Victaulic or similar; Tool room attendant (jobsite only)

GROUP 4: Final clean-up work of debris, grounds and building including but not limited to: street cleaner; cleaning and washing windows; brick cleaner (jobsite only); material cleaner (jobsite only). The classification ""material cleaner"" is to be utilized under the following conditions: A: at demolition site for the salvage of the material.

B: at the conclusion of a job where the material is to be salvaged and stocked to be reused on another job.
C: for the cleaning of salvage material at the jobsite or temporary jobsite yard.

The material cleaner classification should not be used in the performance of ""form stripping, cleaning and oiling and moving to the next point of erection"".

GUNITE LABORER CLASSIFICATIONS
GROUP 1: Structural Nozzleman
GROUP 2: Nozzleman, Gunman, Potman, Groundman
GROUP 3: Reboundman
GROUP 4: Gunite laborer
WRECKING WORK LABORER CLASSIFICATIONS
GROUP 1: Skilled wrecker (removing and salvaging of sash, windows and materials)
GROUP 2: Semi-skilled wrecker (salvaging of other building materials)
LABO0270-011 07/01/2022
MONTEREY, SAN BENITO, SANTA CRUZ, SANTA CLARA COUNTIES
Rates Fringes
LABORER (Plaster Tender)\$ 39.23 29.59
Work on a swing stage scaffold: \$1.00 per hour additional.
LAB00294-001 07/01/2022
FRESNO, KINGS AND MADERA COUNTIES
Rates Fringes
LABORER (Brick) Mason Tender-Brick\$ 35.29 25.21
LAB00294-002 06/28/2021
FRESNO, KINGS, AND MADERA COUNTIES
Rates Fringes
LABORER (TRAFFIC CONTROL/LANE CLOSURE)
Escort Driver, Flag Person\$ 33.48 26.21 Traffic Control Person I\$ 33.78 26.21
Traffic Control Person II\$ 31.28 26.21
TRAFFIC CONTROL PERSON I: Layout of traffic control, crash cushions, construction area and roadside signage.
TRAFFIC CONTROL PERSON II: Installation and removal of temporary/permanent signs, markers, delineators and crash cushions.
LAB00294-005 07/01/2021
FRESNO, KINGS, AND MADERA COUNTIES

Fringes

Tunnel	and	Shaft	Laborers:

GROUP 1	\$ 42.00	25.71
GROUP 2	\$ 41.77	25.71
GROUP 3	\$ 41.52	25.71
GROUP 4	\$ 41.07	25.71
GROUP 5	\$ 40.53	25.71
Shotcrete Specialist	\$ 42.52	25.71

Rates

TUNNEL AND SHAFT CLASSIFICATIONS

GROUP 1: Diamond driller; Groundmen; Gunite and shotcrete
nozzlemen

GROUP 2: Rodmen; Shaft work & raise (below actual or excavated ground level)

GROUP 3: Bit grinder; Blaster, driller, powdermen, heading; Cherry pickermen - where car is lifted; Concrete finisher in tunnel; Concrete screedman; Grout pumpman and potman; Gunite & shotcrete gunman & potman; Headermen; High pressure nozzleman; Miner - tunnel, including top and bottom man on shaft and raise work; Nipper; Nozzleman on slick line; Sandblaster - potman, Robotic Shotcrete Placer, Segment Erector, Tunnel Muck Hauler, Steel Form raiser and setter; Timberman, retimberman (wood or steel or substitute materials therefore); Tugger (for tunnel laborer work); Cable tender; Chuck tender; Powderman - primer house

GROUP 4: Vibrator operator, pavement breaker; Bull gang muckers, trackmen; Concrete crew - includes rodding and spreading, Dumpmen (any method)

GROUP 5: Grout crew; Reboundman; Swamper/ Brakeman

LAB00294-008 06/25/2018

FRESNO, KINGS, AND MADERA COUNTIES

	Rates	Fringes
LABORER (CONSTRUCTION CRAFT LABORERS - AREA B:) Construction Specialist Group GROUP 1 GROUP 1-a GROUP 1-c GROUP 1-c GROUP 1-f GROUP 1-f GROUP 2 GROUP 3 GROUP 4 See groups 1-b and 1-d under 1a	.\$ 29.79 .\$ 30.01 .\$ 29.84 .\$ 30.34 .\$ 30.37 .\$ 29.64 .\$ 29.54 .\$ 23.23	23.20 23.20 23.20 23.20 23.20 23.20 23.20 23.20 23.20 23.20 cations.
LABORER (GARDENERS, HORTICULTURAL & LANDSCAPE		
LABORERS - AREA B:) (1) New Construction (2) Establishment Warranty	.\$ 29.54	23.20
Period LABORER (GUNITE - AREA B:)	.\$ 23.23	23.20
GROUP 1 GROUP 2		22.31 22.31

GROUP 3\$	28.66	22.31
GROUP 4\$	28.54	22.31
LABORER (WRECKING - AREA B:)		
GROUP 1\$	29.79	23.20
GROUP 2\$	29.64	23.20

FOOTNOTES:

Laborers working off or with or from bos'n chairs, swinging scaffolds, belts shall receive \$0.25 per hour above the applicable wage rate. This shall not apply to workers entitled to receive the wage rate set forth in Group 1-a below.

LABORER CLASSIFICATIONS

CONSTRUCTION SPECIALIST GROUP: Asphalt ironer and raker; Chainsaw; Laser beam in connection with laborers' work; Cast-in- place manhole form setter; Pressure pipelayer; Davis trencher - 300 or similar type (and all small trenchers); Blaster; Diamond driller; Multiple unit drill; Hydraulic drill

GROUP 1: Asphalt spreader boxes (all types); Barko, Wacker and similar type tampers; Buggymobile; Caulker, bander, pipewrapper, conduit layer, plastic pipelayer; Certified hazardous waste worker including Leade Abatement; Compactors of all types; Concrete and magnesite mixer, 1/2 yd. and under; Concrete pan work; Concrete sander; Concrete saw; Cribber and/or shoring; Cut granite curb setter; Dri-pak-it machine; Faller, logloader and bucker; Form raiser, slip forms; Green cutter; Headerboard, Hubsetter, aligner, by any method; High pressure blow pipe (1-1/2"" or over, 100 lbs. pressure/over); Hydro seeder and similar type; Jackhammer operator; Jacking of pipe over 12 inches; Jackson and similar type compactor; Kettle tender, pot and worker applying asphalt, lay-kold, creosote, lime, caustic and similar type materials (applying means applying, dipping or handling of such materials); Lagging, sheeting, whaling, bracing, trenchjacking, lagging hammer; Magnesite, epoxyresin, fiberglass, mastic worker (wet or dry); No joint pipe and stripping of same, including repair of voids; Pavement breaker and spader, including tool grinder; Perma curb; Pipelaver (including grade checking in connection with pipelaying); Precast-manhole setter; Pressure pipe tester; Post hole digger, air, gas and electric; Power broom sweeper; Power tampers of all types (except as shown in Group 2); Ram set gun and stud gun; Riprap stonepaver and rock-slinger, including placing of sacked concrete and/or sand (wet or dry) and gabions and similar type; Rotary scarifier or multiple head concrete chipping scarifier; Roto and Ditch Witch; Rototiller; Sandblaster, pot, gun, nozzle operators; Signalling and rigging; Tank cleaner; Tree climber; Turbo blaster; Vibrascreed, bull float in connection with laborers' work; Vibrator; Hazardous waste worker (lead removal); Asbestos and mold removal worker

GROUP 1-a: Joy drill model TWM-2A; Gardner-Denver model DH143 and similar type drills; Track driller; Jack leg driller; Wagon driller; Mechanical drillers, all types regardless of type or method of power; Mechanical pipe layers, all types regardless of type or method of power; Blaster and powder; All work of loading, placing and blasting of all powder and explosives of whatever type regardless of method used for such loading and placing; High scalers (including drilling of same); Tree topper; Bit grinder

GROUP 1-b: Sewer cleaners shall receive \$4.00 per day above Group 1 wage rates. ""Sewer cleaner"" means any worker who handles or comes in contact with raw sewage in small diameter sewers. Those who work inside recently active, large diameter sewers, and all recently active sewer manholes shal receive \$5.00 per day above Group 1 wage rates.

GROUP 1-c: Burning and welding in connection with laborers' work; Synthetic thermoplastics and similar type welding

GROUP 1-d: Maintenance and repair track and road beds. All employees performing work covered herein shall receive \$.25 per hour above their regular rate for all work performed on underground structures not specifically covered herein. This paragraph shall not be construed to apply to work below ground level in open cut. It shall apply to cut and cover work of subway construction after the temporary cover has been placed.

GROUP 1-e: Work on and/or in bell hole footings and shafts thereof, and work on and in deep footings. (A deep footing is a hole 15 feet or more in depth.) In the event the depth of the footing is unknown at the commencement of excavation, and the final depth exceeds 15 feet, the deep footing wage rate would apply to all employees for each and every day worked on or in the excavation of the footing from the date of inception.

GROUP 1-f: Wire winding machine in connection with guniting or shot crete

GROUP 2: Asphalt shoveler; Cement dumper and handling dry cement or gypsum; Choke-setter and rigger (clearing work); Concrete bucket dumper and chute; Concrete chipping and grinding; Concrete laborer (wet or dry); Driller tender, chuck tender, nipper; Guinea chaser (stake), grout crew; High pressure nozzle, adductor; Hydraulic monitor (over 100 lbs. pressure); Loading and unloading, carrying and hauling of all rods and materials for use in reinforcing concrete construction; Pittsburgh chipper and similar type brush shredders; Sloper; Single foot, hand-held, pneumatic tamper; All pneumatic, air, gas and electric tools not listed in Groups 1 through 1-f; Jacking of pipe - under 12 inches

GROUP 3: Construction laborers, including bridge and general laborer; Dump, load spotter; Flag person; Fire watcher; Fence erector; Guardrail erector; Gardener, horticultural and landscape laborer; Jetting; Limber, brush loader and piler; Pavement marker (button setter); Maintenance, repair track and road beds; Streetcar and railroad construction track laborer; Temporary air and water lines, Victaulic or similar; Tool room attendant (jobsite only)

GROUP 4: Final clean-up work of debris, grounds and building including but not limited to: street cleaner; cleaning and washing windows; brick cleaner (jobsite only); material cleaner (jobsite only). The classification ""material cleaner"" is to be utilized under the following conditions:

8/23, 1:13 PM		SAM.gov
A: at demolition site for the sa B: at the conclusion of a job salvaged and stocked to be reu C: for the cleaning of salvage temporary jobsite yard.	where the ma used on anoth	aterial is to be ner job.
The material cleaner classific the performance of ""form stri and moving to the next point o	ipping, clean of erection"	ning and oiling ".
GUNITE LABORER CLASSIFICATIONS		
GROUP 1: Structural Nozzleman		
GROUP 2: Nozzleman, Gunman, Potm	nan, Groundma	an
GROUP 3: Reboundman		
GROUP 4: Gunite laborer		
WRECKING WORK LABORER CLASSIFICA	TIONS	
GROUP 1: Skilled wrecker (remo windows and materials)	oving and sa	lvaging of sash,
GROUP 2: Semi-skilled wrecker materials)	(salvaging o	of other building
LAB00294-010 07/01/2022		
CALAVERAS, FRESNO, KINGS, MADERA JOAQUIN, STANISLAUS & TUOLUMNE	A, MARIPOSA,	MERCED, SAN
	Rates	Fringes
Plasterer tender		
	.\$ 38.02	28.25
Work on a swing stage scaffold: LAB00294-011 07/01/2022		
Work on a swing stage scaffold:	\$1.00 per ho	
Work on a swing stage scaffold: LAB00294-011 07/01/2022	\$1.00 per ho	
Work on a swing stage scaffold: LAB00294-011 07/01/2022	\$1.00 per ho IES Rates	our additional.
Work on a swing stage scaffold: LABO0294-011 07/01/2022 FRESNO, KINGS, AND MADERA COUNTI	\$1.00 per ho IES Rates \$ 38.02	our additional. Fringes 28.25
Work on a swing stage scaffold: LABO0294-011 07/01/2022 FRESNO, KINGS, AND MADERA COUNTI LABORER (Plaster Tender)	\$1.00 per ho IES Rates \$ 38.02	our additional. Fringes 28.25
Work on a swing stage scaffold: LABO0294-011 07/01/2022 FRESNO, KINGS, AND MADERA COUNTI LABORER (Plaster Tender) Work on a swing stage scaffold:	\$1.00 per ho IES Rates \$ 38.02	our additional. Fringes 28.25
Work on a swing stage scaffold: LABO0294-011 07/01/2022 FRESNO, KINGS, AND MADERA COUNTI LABORER (Plaster Tender) Work on a swing stage scaffold: LABO0304-002 06/28/2021	\$1.00 per ho IES Rates \$ 38.02	our additional. Fringes 28.25 our additional.
Work on a swing stage scaffold: LABO0294-011 07/01/2022 FRESNO, KINGS, AND MADERA COUNTI LABORER (Plaster Tender) Work on a swing stage scaffold: LABO0304-002 06/28/2021	\$1.00 per ho IES Rates \$ 38.02 \$1.00 per ho	our additional. Fringes 28.25 our additional.

Traffic Control Person II...\$ 32.28

26.21

TRAFFIC CONTROL PERSON I: Layout of traffic control, crash cushions, construction area and roadside signage.

TRAFFIC CONTROL PERSON II: Installation and removal of temporary/permanent signs, markers, delineators and crash cushions.

LAB00304-003 07/01/2021

ALAMEDA COUNTY

Rates Fringes

Tunnel and Shaft Laborers:

GROUP 1\$	42.00	25.71
GROUP 2\$	41.77	25.71
GROUP 3\$	41.52	25.71
GROUP 4\$	41.07	25.71
GROUP 5\$	40.53	25.71
Shotcrete Specialist\$	42.52	25.71

TUNNEL AND SHAFT CLASSIFICATIONS

GROUP 1: Diamond driller; Groundmen; Gunite and shotcrete nozzlemen

GROUP 2: Rodmen; Shaft work & raise (below actual or excavated ground level)

GROUP 3: Bit grinder; Blaster, driller, powdermen, heading; Cherry pickermen - where car is lifted; Concrete finisher in tunnel; Concrete screedman; Grout pumpman and potman; Gunite & shotcrete gunman & potman; Headermen; High pressure nozzleman; Miner - tunnel, including top and bottom man on shaft and raise work; Nipper; Nozzleman on slick line; Sandblaster - potman, Robotic Shotcrete Placer, Segment Erector, Tunnel Muck Hauler, Steel Form raiser and setter; Timberman, retimberman (wood or steel or substitute materials therefore); Tugger (for tunnel laborer work); Cable tender; Chuck tender; Powderman - primer house

GROUP 4: Vibrator operator, pavement breaker; Bull gang muckers, trackmen; Concrete crew - includes rodding and spreading, Dumpmen (any method)

GROUP 5: Grout crew; Reboundman; Swamper/ Brakeman

LAB00304-004 06/25/2018

ALAMEDA COUNTY

	Rates	Fringes
LABORER (CONSTRUCTION CRAFT LABORERS - AREA A:) Construction Specialist Group	.\$ 31.49	23.20
GROUP 1	-	23.20
GROUP 1-a GROUP 1-c		23.20 23.20
GROUP 1-e		23.20

https://sam.gov/wage-determination/CA20230018/14

GROUP 1-f\$ 30.3723.2GROUP 2\$ 30.6423.2GROUP 3\$ 30.5423.2GROUP 4\$ 24.2323.2	.0 .0
See groups 1-b and 1-d under laborer classifications.	
LABORER (GARDENERS, HORTICULTURAL & LANDSCAPE	
LABORERS - AREA A:)	
<pre>(1) New Construction\$ 30.54 23.2 (2) Establishment Warranty</pre>	:0
Period\$ 24.23 23.2 LABORER (GUNITE - AREA A:)	:0
GROUP 1\$ 30.75 22.3	1
GROUP 2\$ 30.25 22.3	1
GROUP 3\$ 29.66 22.3	1
GROUP 4\$ 29.54 22.3	1
LABORER (WRECKING - AREA A:)	
GROUP 1\$ 30.79 23.2	.0
GROUP 2\$ 30.64 23.2	.0

FOOTNOTES:

Laborers working off or with or from bos'n chairs, swinging scaffolds, belts shall receive \$0.25 per hour above the applicable wage rate. This shall not apply to workers entitled to receive the wage rate set forth in Group 1-a below.

LABORER CLASSIFICATIONS

CONSTRUCTION SPECIALIST GROUP: Asphalt ironer and raker; Chainsaw; Laser beam in connection with laborers' work; Cast-in- place manhole form setter; Pressure pipelayer; Davis trencher - 300 or similar type (and all small trenchers); Blaster; Diamond driller; Multiple unit drill; Hydraulic drill

GROUP 1: Asphalt spreader boxes (all types); Barko, Wacker and similar type tampers; Buggymobile; Caulker, bander, pipewrapper, conduit layer, plastic pipelayer; Certified hazardous waste worker including Leade Abatement; Compactors of all types; Concrete and magnesite mixer, 1/2 yd. and under; Concrete pan work; Concrete sander; Concrete saw; Cribber and/or shoring; Cut granite curb setter; Dri-pak-it machine; Faller, logloader and bucker; Form raiser, slip forms; Green cutter; Headerboard, Hubsetter, aligner, by any method; High pressure blow pipe (1-1/2"" or over, 100 lbs. pressure/over); Hydro seeder and similar type; Jackhammer operator; Jacking of pipe over 12 inches; Jackson and similar type compactor; Kettle tender, pot and worker applying asphalt, lay-kold, creosote, lime, caustic and similar type materials (applying means applying, dipping or handling of such materials); Lagging, sheeting, whaling, bracing, trenchjacking, lagging hammer; Magnesite, epoxyresin, fiberglass, mastic worker (wet or dry); No joint pipe and stripping of same, including repair of voids; Pavement breaker and spader, including tool grinder; Perma curb; Pipelayer (including grade checking in connection with pipelaying); Precast-manhole setter; Pressure pipe tester; Post hole digger, air, gas and electric; Power broom sweeper; Power tampers of all types (except as shown in Group 2); Ram set gun and stud gun; Riprap stonepaver and rock-slinger, including placing of

sacked concrete and/or sand (wet or dry) and gabions and similar type; Rotary scarifier or multiple head concrete chipping scarifier; Roto and Ditch Witch; Rototiller; Sandblaster, pot, gun, nozzle operators; Signalling and rigging; Tank cleaner; Tree climber; Turbo blaster; Vibrascreed, bull float in connection with laborers' work; Vibrator; Hazardous waste worker (lead removal); Asbestos and mold removal worker

GROUP 1-a: Joy drill model TWM-2A; Gardner-Denver model DH143 and similar type drills; Track driller; Jack leg driller; Wagon driller; Mechanical drillers, all types regardless of type or method of power; Mechanical pipe layers, all types regardless of type or method of power; Blaster and powder; All work of loading, placing and blasting of all powder and explosives of whatever type regardless of method used for such loading and placing; High scalers (including drilling of same); Tree topper; Bit grinder

GROUP 1-b: Sewer cleaners shall receive \$4.00 per day above Group 1 wage rates. ""Sewer cleaner"" means any worker who handles or comes in contact with raw sewage in small diameter sewers. Those who work inside recently active, large diameter sewers, and all recently active sewer manholes shal receive \$5.00 per day above Group 1 wage rates.

GROUP 1-c: Burning and welding in connection with laborers' work; Synthetic thermoplastics and similar type welding

GROUP 1-d: Maintenance and repair track and road beds. All employees performing work covered herein shall receive \$.25 per hour above their regular rate for all work performed on underground structures not specifically covered herein. This paragraph shall not be construed to apply to work below ground level in open cut. It shall apply to cut and cover work of subway construction after the temporary cover has been placed.

GROUP 1-e: Work on and/or in bell hole footings and shafts thereof, and work on and in deep footings. (A deep footing is a hole 15 feet or more in depth.) In the event the depth of the footing is unknown at the commencement of excavation, and the final depth exceeds 15 feet, the deep footing wage rate would apply to all employees for each and every day worked on or in the excavation of the footing from the date of inception.

GROUP 1-f: Wire winding machine in connection with guniting or shot crete

GROUP 2: Asphalt shoveler; Cement dumper and handling dry cement or gypsum; Choke-setter and rigger (clearing work); Concrete bucket dumper and chute; Concrete chipping and grinding; Concrete laborer (wet or dry); Driller tender, chuck tender, nipper; Guinea chaser (stake), grout crew; High pressure nozzle, adductor; Hydraulic monitor (over 100 lbs. pressure); Loading and unloading, carrying and hauling of all rods and materials for use in reinforcing concrete construction; Pittsburgh chipper and similar type brush shredders; Sloper; Single foot, hand-held, pneumatic tamper; All pneumatic, air, gas and electric tools not listed in Groups 1 through 1-f; Jacking of pipe - under 12 inches

SAM.gov GROUP 3: Construction laborers, including bridge and general laborer; Dump, load spotter; Flag person; Fire watcher; Fence erector; Guardrail erector; Gardener, horticultural and landscape laborer; Jetting; Limber, brush loader and piler; Pavement marker (button setter); Maintenance, repair track and road beds; Streetcar and railroad construction track laborer; Temporary air and water lines, Victaulic or similar; Tool room attendant (jobsite only) GROUP 4: Final clean-up work of debris, grounds and building including but not limited to: street cleaner; cleaning and washing windows; brick cleaner (jobsite only); material cleaner (jobsite only). The classification ""material cleaner"" is to be utilized under the following conditions: A: at demolition site for the salvage of the material. B: at the conclusion of a job where the material is to be salvaged and stocked to be reused on another job. C: for the cleaning of salvage material at the jobsite or temporary jobsite yard. The material cleaner classification should not be used in the performance of ""form stripping, cleaning and oiling and moving to the next point of erection"". GUNITE LABORER CLASSIFICATIONS GROUP 1: Structural Nozzleman GROUP 2: Nozzleman, Gunman, Potman, Groundman GROUP 3: Reboundman GROUP 4: Gunite laborer _____ WRECKING WORK LABORER CLASSIFICATIONS GROUP 1: Skilled wrecker (removing and salvaging of sash, windows and materials) GROUP 2: Semi-skilled wrecker (salvaging of other building materials) _____ LAB00304-005 07/01/2022

ALAMEDA COUNTY

Rates Fringes

Brick Tender.....\$ 36.05 27.12

FOOTNOTES: Work on jobs where heat-protective clothing is required: \$2.00 per hour additional. Work at grinders: \$.25 per hour additional. Manhole work: \$2.00 per day additional.

LAB00304-008 07/01/2022

ALAMEDA AND CONTRA COSTA COUNTIES:

Rates

Fringes

24.11

24.11

Plasterer	tende	r	• • • • • • • • • • •	\$ 40	.48		30.23	
Work on a	swing	stage	<pre>scaffold:</pre>	\$1.00	per	hour	additional.	
LAB00324	-002 00	6/28/20						

CONTRA COSTA COUNTY

Rates Fr:	inges
-----------	-------

LABORER (TRAFFIC CONTROL/LANE CLOSURE) Escort Driver, Flag Person..\$ 34.48 26.21 Traffic Control Person I....\$ 34.78 26.21 Traffic Control Person II...\$ 32.28 26.21

TRAFFIC CONTROL PERSON I: Layout of traffic control, crash cushions, construction area and roadside signage.

TRAFFIC CONTROL PERSON II: Installation and removal of temporary/permanent signs, markers, delineators and crash cushions.

LAB00324-006 06/25/2018

CONTRA COSTA COUNTY

Tunnel a

	Rates	Fringes
el and	Shaft Laborers:	
GROUP	1\$ 37.82	24.11
GROUP	2\$ 37.59	24.11
GROUP	3\$ 37.34	24.11
GROUP	4\$ 36.89	24.11

TUNNEL AND SHAFT CLASSIFICATIONS

GROUP 5.....\$ 36.35

Shotcrete Specialist.....\$ 38.34

GROUP 1: Diamond driller; Groundmen; Gunite and shotcrete nozzlemen

GROUP 2: Rodmen; Shaft work & raise (below actual or excavated ground level)

GROUP 3: Bit grinder; Blaster, driller, powdermen, heading; Cherry pickermen - where car is lifted; Concrete finisher in tunnel; Concrete screedman; Grout pumpman and potman; Gunite & shotcrete gunman & potman; Headermen; High pressure nozzleman; Miner - tunnel, including top and bottom man on shaft and raise work; Nipper; Nozzleman on slick line; Sandblaster - potman, Robotic Shotcrete Placer, Segment Erector, Tunnel Muck Hauler, Steel Form raiser and setter; Timberman, retimberman (wood or steel or substitute materials therefore); Tugger (for tunnel laborer work); Cable tender; Chuck tender; Powderman - primer house

GROUP 4: Vibrator operator, pavement breaker; Bull gang muckers, trackmen; Concrete crew - includes rodding and spreading, Dumpmen (any method)

GROUP 5: Grout crew; Reboundman; Swamper/ Brakeman

LAB00324-012 06/25/2018

CONTRA COSTA COUNTY

R	ates Fr	ringes
LABORER (CONSTRUCTION CRAFT LABORERS - AREA A:) Construction Specialist		
Group\$	31.49	23.20
GROUP 1\$		23.20
GROUP 1-a\$		23.20
GROUP 1-c\$		23.20
GROUP 1-e\$		23.20
GROUP 1-f\$		23.20
GROUP 1-g\$	30.99	23.20
GROUP 2\$	30.64	23.20
GROUP 3\$	30.54	23.20
GROUP 4\$ 2		23.20
See groups 1-b and 1-d under labo	rer classificat	ions.
LABORER (GARDENERS,		
HORTICULURAL & LANDSCAPE		
LABORERS - AREA A:)		
(1) New Construction\$	30.54	23.20
(2) Establishment Warranty		
Period\$ 2	24.23	23.20
LABORER (GUNITE - AREA A:)		
GROUP 1\$		22.31
GROUP 2\$		22.31
GROUP 3\$ 2		22.31
GROUP 4\$ 2	29.54	22.31
LABORER (WRECKING - AREA A:)		
GROUP 1\$		23.20
GROUP 2\$	30.64	23.20

FOOTNOTES:

Laborers working off or with or from bos'n chairs, swinging scaffolds, belts shall receive \$0.25 per hour above the applicable wage rate. This shall not apply to workers entitled to receive the wage rate set forth in Group 1-a below.

LABORER CLASSIFICATIONS

CONSTRUCTION SPECIALIST GROUP: Asphalt ironer and raker; Chainsaw; Laser beam in connection with laborers' work; Cast-in- place manhole form setter; Pressure pipelayer; Davis trencher - 300 or similar type (and all small trenchers); Blaster; Diamond driller; Multiple unit drill; Hydraulic drill

GROUP 1: Asphalt spreader boxes (all types); Barko, Wacker and similar type tampers; Buggymobile; Caulker, bander, pipewrapper, conduit layer, plastic pipelayer; Certified hazardous waste worker including Leade Abatement; Compactors of all types; Concrete and magnesite mixer, 1/2 yd. and under; Concrete pan work; Concrete sander; Concrete saw; Cribber and/or shoring; Cut granite curb setter; Dri-pak-it machine; Faller, logloader and bucker; Form raiser, slip forms; Green cutter; Headerboard, Hubsetter, aligner, by any method; High pressure blow pipe (1-1/2"" or

SAM.gov

over, 100 lbs. pressure/over); Hydro seeder and similar type; Jackhammer operator; Jacking of pipe over 12 inches; Jackson and similar type compactor; Kettle tender, pot and worker applying asphalt, lay-kold, creosote, lime, caustic and similar type materials (applying means applying, dipping or handling of such materials); Lagging, sheeting, whaling, bracing, trenchjacking, lagging hammer; Magnesite, epoxyresin, fiberglass, mastic worker (wet or dry); No joint pipe and stripping of same, including repair of voids; Pavement breaker and spader, including tool grinder; Perma curb; Pipelayer (including grade checking in connection with pipelaying); Precast-manhole setter; Pressure pipe tester; Post hole digger, air, gas and electric; Power broom sweeper; Power tampers of all types (except as shown in Group 2); Ram set gun and stud gun; Riprap stonepaver and rock-slinger, including placing of sacked concrete and/or sand (wet or dry) and gabions and similar type; Rotary scarifier or multiple head concrete chipping scarifier; Roto and Ditch Witch; Rototiller; Sandblaster, pot, gun, nozzle operators; Signalling and rigging; Tank cleaner; Tree climber; Turbo blaster; Vibrascreed, bull float in connection with laborers' work; Vibrator; Hazardous waste worker (lead removal); Asbestos and mold removal worker

GROUP 1-a: Joy drill model TWM-2A; Gardner-Denver model DH143 and similar type drills; Track driller; Jack leg driller; Wagon driller; Mechanical drillers, all types regardless of type or method of power; Mechanical pipe layers, all types regardless of type or method of power; Blaster and powder; All work of loading, placing and blasting of all powder and explosives of whatever type regardless of method used for such loading and placing; High scalers (including drilling of same); Tree topper; Bit grinder

GROUP 1-b: Sewer cleaners shall receive \$4.00 per day above Group 1 wage rates. ""Sewer cleaner"" means any worker who handles or comes in contact with raw sewage in small diameter sewers. Those who work inside recently active, large diameter sewers, and all recently active sewer manholes shal receive \$5.00 per day above Group 1 wage rates.

GROUP 1-c: Burning and welding in connection with laborers' work; Synthetic thermoplastics and similar type welding

GROUP 1-d: Maintenance and repair track and road beds. All employees performing work covered herein shall receive \$.25 per hour above their regular rate for all work performed on underground structures not specifically covered herein. This paragraph shall not be construed to apply to work below ground level in open cut. It shall apply to cut and cover work of subway construction after the temporary cover has been placed.

GROUP 1-e: Work on and/or in bell hole footings and shafts thereof, and work on and in deep footings. (A deep footing is a hole 15 feet or more in depth.) In the event the depth of the footing is unknown at the commencement of excavation, and the final depth exceeds 15 feet, the deep footing wage rate would apply to all employees for each and every day worked on or in the excavation of the footing from the date of inception.

GROUP 1-f: Wire winding machine in connection with guniting

or shot crete

GROUP 1-g, CONTRA COSTA COUNTY: Pipelayer (including grade checking in connection with pipelaying); Caulker; Bander; Pipewrapper; Conduit layer; Plastic pipe layer; Pressure pipe tester; No joint pipe and stripping of same, including repair of voids; Precast manhole setters, cast in place manhole form setters

GROUP 2: Asphalt shoveler; Cement dumper and handling dry cement or gypsum; Choke-setter and rigger (clearing work); Concrete bucket dumper and chute; Concrete chipping and grinding; Concrete laborer (wet or dry); Driller tender, chuck tender, nipper; Guinea chaser (stake), grout crew; High pressure nozzle, adductor; Hydraulic monitor (over 100 lbs. pressure); Loading and unloading, carrying and hauling of all rods and materials for use in reinforcing concrete construction; Pittsburgh chipper and similar type brush shredders; Sloper; Single foot, hand-held, pneumatic tamper; All pneumatic, air, gas and electric tools not listed in Groups 1 through 1-f; Jacking of pipe - under 12 inches

GROUP 3: Construction laborers, including bridge and general laborer; Dump, load spotter; Flag person; Fire watcher; Fence erector; Guardrail erector; Gardener, horticultural and landscape laborer; Jetting; Limber, brush loader and piler; Pavement marker (button setter); Maintenance, repair track and road beds; Streetcar and railroad construction track laborer; Temporary air and water lines, Victaulic or similar; Tool room attendant (jobsite only)

GROUP 4: Final clean-up work of debris, grounds and building including but not limited to: street cleaner; cleaning and washing windows; brick cleaner (jobsite only); material cleaner (jobsite only). The classification ""material cleaner"" is to be utilized under the following conditions:A: at demolition site for the salvage of the material.B: at the conclusion of a job where the material is to be salvaged and stocked to be reused on another job.

C: for the cleaning of salvage material at the jobsite or temporary jobsite yard.

The material cleaner classification should not be used in the performance of ""form stripping, cleaning and oiling and moving to the next point of erection"".

.....

GUNITE LABORER CLASSIFICATIONS

GROUP 1: Structural Nozzleman

GROUP 2: Nozzleman, Gunman, Potman, Groundman

GROUP 3: Reboundman

GROUP 4: Gunite laborer

WRECKING WORK LABORER CLASSIFICATIONS

GROUP 1: Skilled wrecker (removing and salvaging of sash, windows and materials)

GROUP 2: Semi-skilled wrecker (salvaging of other building materials)

GROUP 1-g, CONTRA COSTA COUNTY: Pipelayer (including grade checking in connection with pipelaying); Caulker; Bander; Pipewrapper; Conduit layer; Plastic pipe layer; Pressure pipe tester; No joint pipe and stripping of same, including repair of voids; Precast manhole setters, cast in place manhole form setters

LAB00324-014 07/01/2022

CONTRA COSTA COUNTY:

Rates	Fringes

Brick Tender.....\$ 36.05 27.12

FOOTNOTES: Work on jobs where heat-protective clothing is required: \$2.00 per hour additional. Work at grinders: \$.25 per hour additional. Manhole work: \$2.00 per day additional.

LAB00324-018 07/01/2022

ALAMEDA AND CONTRA COSTA COUNTIES:

	Rates	Fringes
Plasterer tender	.\$ 40.48	30.23
Work on a swing stage scaffold:	\$1.00 per hour a	additional.
LAB01130-002 06/28/2021		
MARIPOSA, MERCED, STANISLAUS, AN	D TUOLUMNE COUNT	TIES
	Rates	Fringes
LABORER (TRAFFIC CONTROL/LANE CLOSURE)		
Escort Driver, Flag Person.	.\$ 33.48	26.21
Traffic Control Person I	.\$ 33.78	26.21
Traffic Control Person II	.\$ 31.28	26.21
TRAFFIC CONTROL PERSON I: Layo cushions, construction area an TRAFFIC CONTROL PERSON II: Ins temporary/permanent signs, mar cushions.	d roadside signa tallation and re	age. emoval of
LAB01130-003 06/26/2017		
MARIPOSA, MERCED, STANISLAUS, AN	D TUOLUMNE COUNT	TIES
	Rates	Fringes
Tunnel and Shaft Laborers:		
		24 02
GROUP 1		24.83
GROUP 2	.\$ 36.3/	24.83

SAM.gov

GROUP 3\$ 36	12 24.83
GROUP 4\$ 35	67 24.83
GROUP 5\$ 35	13 24.83
Shotcrete Specialist\$ 37	12 24.83

TUNNEL AND SHAFT CLASSIFICATIONS

GROUP 1: Diamond driller; Groundmen; Gunite and shotcrete nozzlemen

GROUP 2: Rodmen; Shaft work & raise (below actual or excavated ground level)

GROUP 3: Bit grinder; Blaster, driller, powdermen, heading; Cherry pickermen - where car is lifted; Concrete finisher in tunnel; Concrete screedman; Grout pumpman and potman; Gunite & shotcrete gunman & potman; Headermen; High pressure nozzleman; Miner - tunnel, including top and bottom man on shaft and raise work; Nipper; Nozzleman on slick line; Sandblaster - potman, Robotic Shotcrete Placer, Segment Erector, Tunnel Muck Hauler, Steel Form raiser and setter; Timberman, retimberman (wood or steel or substitute materials therefore); Tugger (for tunnel laborer work); Cable tender; Chuck tender; Powderman - primer house

GROUP 4: Vibrator operator, pavement breaker; Bull gang muckers, trackmen; Concrete crew - includes rodding and spreading, Dumpmen (any method)

GROUP 5: Grout crew; Reboundman; Swamper/ Brakeman

LAB01130-005 07/01/2022

MARIPOSA, MERCED, STANISLAUS AND TUOLUMNE COUNTIES

Rates Fringes

LABORER Mason Tender-Brick.....\$ 35.29 25.21

LAB01130-007 06/25/2018

MARIPOSA, MERCED, STANISLAUS, AND TUOLUMNE, COUNTIES

Rates Fringes

LABORER (CONSTRUCTION CRAFT LABORERS - AREA B:)

LABORERS - AREA D.)	
Construction Specialist	
Group\$ 30.49	23.20
GROUP 1\$ 29.79	23.20
GROUP 1-a\$ 30.01	23.20
GROUP 1-c\$ 29.84	23.20
GROUP 1-e\$ 30.34	23.20
GROUP 1-f\$ 29.37	23.20
GROUP 2\$ 29.64	23.20
GROUP 3\$ 29.54	23.20
GROUP 4\$ 23.23	23.20
See groups 1-b and 1-d under laborer classificat	ions.
LABORER (GARDENERS,	
HORTICULTURAL & LANDSCAPE	
LABORERS - AREA B:)	
(1) New Construction\$ 29.54	23.20
(2) Establishment Warranty	
\-,	

SAM.gov

Period\$ 2 LABORER (GUNITE - AREA B:)	23.23	23.20
GROUP 1\$ 2	29.75	22.31
GROUP 2\$ 2	29.25	22.31
GROUP 3\$ 2	28.66	22.31
GROUP 4\$ 2	28.54	22.31
LABORER (WRECKING - AREA B:)		
GROUP 1\$ 2	29.79	23.20
GROUP 2\$ 2	29.64	23.20

FOOTNOTES:

Laborers working off or with or from bos'n chairs, swinging scaffolds, belts shall receive \$0.25 per hour above the applicable wage rate. This shall not apply to workers entitled to receive the wage rate set forth in Group 1-a below.

LABORER CLASSIFICATIONS

CONSTRUCTION SPECIALIST GROUP: Asphalt ironer and raker; Chainsaw; Laser beam in connection with laborers' work; Cast-in- place manhole form setter; Pressure pipelayer; Davis trencher - 300 or similar type (and all small trenchers); Blaster; Diamond driller; Multiple unit drill; Hydraulic drill

GROUP 1: Asphalt spreader boxes (all types); Barko, Wacker and similar type tampers; Buggymobile; Caulker, bander, pipewrapper, conduit layer, plastic pipelayer; Certified hazardous waste worker including Leade Abatement; Compactors of all types; Concrete and magnesite mixer, 1/2 yd. and under; Concrete pan work; Concrete sander; Concrete saw; Cribber and/or shoring; Cut granite curb setter; Dri-pak-it machine; Faller, logloader and bucker; Form raiser, slip forms; Green cutter; Headerboard, Hubsetter, aligner, by any method; High pressure blow pipe (1-1/2"" or over, 100 lbs. pressure/over); Hydro seeder and similar type; Jackhammer operator; Jacking of pipe over 12 inches; Jackson and similar type compactor; Kettle tender, pot and worker applying asphalt, lay-kold, creosote, lime, caustic and similar type materials (applying means applying, dipping or handling of such materials); Lagging, sheeting, whaling, bracing, trenchjacking, lagging hammer; Magnesite, epoxyresin, fiberglass, mastic worker (wet or dry); No joint pipe and stripping of same, including repair of voids; Pavement breaker and spader, including tool grinder; Perma curb; Pipelayer (including grade checking in connection with pipelaying); Precast-manhole setter; Pressure pipe tester; Post hole digger, air, gas and electric; Power broom sweeper; Power tampers of all types (except as shown in Group 2); Ram set gun and stud gun; Riprap stonepaver and rock-slinger, including placing of sacked concrete and/or sand (wet or dry) and gabions and similar type; Rotary scarifier or multiple head concrete chipping scarifier; Roto and Ditch Witch; Rototiller; Sandblaster, pot, gun, nozzle operators; Signalling and rigging; Tank cleaner; Tree climber; Turbo blaster; Vibrascreed, bull float in connection with laborers' work; Vibrator; Hazardous waste worker (lead removal); Asbestos and mold removal worker

GROUP 1-a: Joy drill model TWM-2A; Gardner-Denver model DH143

and similar type drills; Track driller; Jack leg driller; Wagon driller; Mechanical drillers, all types regardless of type or method of power; Mechanical pipe layers, all types regardless of type or method of power; Blaster and powder; All work of loading, placing and blasting of all powder and explosives of whatever type regardless of method used for such loading and placing; High scalers (including drilling of same); Tree topper; Bit grinder

GROUP 1-b: Sewer cleaners shall receive \$4.00 per day above Group 1 wage rates. ""Sewer cleaner"" means any worker who handles or comes in contact with raw sewage in small diameter sewers. Those who work inside recently active, large diameter sewers, and all recently active sewer manholes shal receive \$5.00 per day above Group 1 wage rates.

GROUP 1-c: Burning and welding in connection with laborers' work; Synthetic thermoplastics and similar type welding

GROUP 1-d: Maintenance and repair track and road beds. All employees performing work covered herein shall receive \$.25 per hour above their regular rate for all work performed on underground structures not specifically covered herein. This paragraph shall not be construed to apply to work below ground level in open cut. It shall apply to cut and cover work of subway construction after the temporary cover has been placed.

GROUP 1-e: Work on and/or in bell hole footings and shafts thereof, and work on and in deep footings. (A deep footing is a hole 15 feet or more in depth.) In the event the depth of the footing is unknown at the commencement of excavation, and the final depth exceeds 15 feet, the deep footing wage rate would apply to all employees for each and every day worked on or in the excavation of the footing from the date of inception.

GROUP 1-f: Wire winding machine in connection with guniting or shot crete

GROUP 2: Asphalt shoveler; Cement dumper and handling dry cement or gypsum; Choke-setter and rigger (clearing work); Concrete bucket dumper and chute; Concrete chipping and grinding; Concrete laborer (wet or dry); Driller tender, chuck tender, nipper; Guinea chaser (stake), grout crew; High pressure nozzle, adductor; Hydraulic monitor (over 100 lbs. pressure); Loading and unloading, carrying and hauling of all rods and materials for use in reinforcing concrete construction; Pittsburgh chipper and similar type brush shredders; Sloper; Single foot, hand-held, pneumatic tamper; All pneumatic, air, gas and electric tools not listed in Groups 1 through 1-f; Jacking of pipe - under 12 inches

GROUP 3: Construction laborers, including bridge and general laborer; Dump, load spotter; Flag person; Fire watcher; Fence erector; Guardrail erector; Gardener, horticultural and landscape laborer; Jetting; Limber, brush loader and piler; Pavement marker (button setter); Maintenance, repair track and road beds; Streetcar and railroad construction track laborer; Temporary air and water lines, Victaulic or similar; Tool room attendant (jobsite only)

GROUP 4: Final clean-up work of debris, grounds and building

<pre>including but not limited to: street cleaner; cleaning and washing windows; brick cleaner (jobsite only); material cleaner (jobsite only). The classification ""material cleaner"" is to be utilized under the following conditions: A: at demolition site for the salvage of the material. B: at the conclusion of a job where the material is to be salvaged and stocked to be reused on another job. C: for the cleaning of salvage material at the jobsite or temporary jobsite yard. The material cleaner classification should not be used in the performance of ""form stripping, cleaning and oiling and moving to the next point of erection"".</pre>
GUNITE LABORER CLASSIFICATIONS
GROUP 1: Structural Nozzleman
GROUP 2: Nozzleman, Gunman, Potman, Groundman
GROUP 3: Reboundman
GROUP 4: Gunite laborer
WRECKING WORK LABORER CLASSIFICATIONS
GROUP 1: Skilled wrecker (removing and salvaging of sash, windows and materials)
GROUP 2: Semi-skilled wrecker (salvaging of other building materials)
LAB01130-008 07/01/2022
CALAVERAS, FRESNO, KINGS, MADERA, MARIPOSA, MERCED, SAN JOAQUIN, STANISLAUS & TUOLUMNE
Rates Fringes
Plasterer tender
Work on a swing stage scaffold: \$1.00 per hour additional.
LAB01130-009 07/01/2022
MARIPOSA, MERCED, STANISLAUS, AND TUOLUMNE COUNTIES
Rates Fringes
LABORER (Plaster Tender)\$ 38.02 28.25
Work on a swing stage scaffold: \$1.00 per hour additional.
PAIN0016-001 01/01/2023
ALAMEDA, CONTRA COSTA, MONTEREY, SAN BENITO, SAN MATEO, SANTA CLARA, AND SANTA CRUZ COUNTIES

28/23, 1:13 PM		SAM.gov
	Rates	Fringes
Painters:	\$ 47.42	27.28
PREMIUMS:		
EXOTIC MATERIALS - \$1.25 add SPRAY WORK: - \$0.50 addition INDUSTRIAL PAINTING - \$0.25 [Work on industrial buildi processing of goods for sa (bridges), stacks, towers,	al per hour. additional per h ngs used for the le or service; s	our manufacture and teel construction
HIGH WORK: over 50 feet - \$2.00 per hou 100 to 180 feet - \$4.00 per Over 180 feet - \$6.00 per hou	hour additional	
PAIN0016-003 01/01/2023		
AREA 1: ALAMEDA, CONTRA COSTA CLARA COUNTIES	A, SAN FRANCISCO	, SAN MATEO & SANTA
AREA 2: CALAVERAS, MARIPOA, I JOAQUIN, SANTA CRUZ, STANISL		
	Rates	Fringes
Drywall Finisher/Taper AREA 1 AREA 2	\$ 53.03	30.24 28.84
PAIN0016-012 01/01/2023		
ALAMEDA, CONTRA COSTA, MARIP SAN FRANCISCO, SAN MATEO, SA		
	Rates	Fringes
SOFT FLOOR LAYER		32.63
PAIN0016-015 01/01/2023		
CALAVERAS, MARIPOSA, MERCED, COUNTIES	SAN JOAQUIN, ST	ANISLAUS & TUOLUMNE
	Rates	Fringes
PAINTER Brush	\$ 38.23	22.05

SAN FRANCISCO COUNTY

	Rates	Fringes
PAINTER		27.28
PAIN0169-001 01/01/2023		
FRESNO, KINGS, MADERA, MARIPOS	A AND MERCED	COUNTIES:
	Rates	Fringes
GLAZIER	\$ 44.33	28.88
PAIN0169-005 01/01/2023		
ALAMEDA CONTRA COSTA, MONTEREY MATEO, SANTA CLARA & SANTA CRU		SAN FRANCISCO, SAN
	Rates	Fringes
GLAZIER	•	32.45
PAIN0294-004 01/01/2023		
FRESNO, KINGS AND MADERA COUNT	IES	
	Rates	Fringes
PAINTER Brush, Roller Drywall Finisher/Taper		21.30 28.99
FOOTNOTE: Spray Painters & Paperhanger hour. Painters doing Drywall additional per hour. Lead A \$1.50 additional per hour. not include work from a lift	Patching rec baters & Sand High Time - o	eive \$1.25 blasters receive ver 30 feet (does
PAIN0294-005 01/01/2023		
FRESNO, KINGS & MADERA		
	Rates	Fringes
SOFT FLOOR LAYER		23.19
PAIN0767-001 01/01/2023		
CALAVERAS, SAN JOAQUIN, STANIS	LAUS AND TUOL	UMNE COUNTIES:
	Rates	Fringes
GLAZIER	\$ 43.15	33.72
PAID HOLIDAYS: New Year's Da President's Day, Memorial Da Veteran's Day, Thanksgiving	y, Independen	ce Day, Labor Day,
Employee required to wear a	body harness	shall receive \$1.50

Employee required to wear a body harness shall receive \$1.50 per hour above the basic hourly rate at any elevation.

PAIN1176-001 07/01/2022

HIGHWAY IMPRØVEMENT

	Rates	Fringes
Parking Lot Striping/Highway Marking: GROUP 1 GROUP 2 GROUP 3	\$ 34.71	17.62 17.62 17.62
CLASSIFICATIONS		
GROUP 1: Striper: Layout ar stripes and marking; hot th stripes and markings		
GROUP 2: Gamecourt & Playgr	ound Installer	
GROUP 3: Protective Coating	g, Pavement Seal	ing
PAIN1237-003 01/01/2023		
CALAVERAS; SAN JOAQUIN COUNTI COUNTIES:	ES; STANISLAUS	AND TUOLUMNE
	Rates	Fringes
SOFT FLOOR LAYER		25.96
PLAS0066-002 07/01/2019		
ALAMEDA, CONTRA COSTA, SAN MA	TEO AND SAN FRA	NCISCO COUNTIES:
	TEO AND SAN FRA Rates	NCISCO COUNTIES: Fringes
ALAMEDA, CONTRA COSTA, SAN MA PLASTERER	Rates	
ALAMEDA, CONTRA COSTA, SAN MA	Rates	Fringes
ALAMEDA, CONTRA COSTA, SAN MA PLASTERER	Rates	Fringes
ALAMEDA, CONTRA COSTA, SAN MA PLASTERER PLAS0300-001 07/01/2018 PLASTERER AREA 188: Fresno	Rates \$ 42.41 Rates	Fringes 30.73
ALAMEDA, CONTRA COSTA, SAN MA PLASTERER PLAS0300-001 07/01/2018 PLASTERER AREA 188: Fresno AREA 224: San Benito, Santa Clara, Santa Cruz.	Rates \$ 42.41 Rates \$ 32.70 \$ 32.88	Fringes 30.73 Fringes
ALAMEDA, CONTRA COSTA, SAN MA PLASTERER PLAS0300-001 07/01/2018 PLASTERER AREA 188: Fresno AREA 224: San Benito, Santa Clara, Santa Cruz. AREA 295: Calaveras & S Joaquin Couonties AREA 337: Monterey Cour	Rates \$ 42.41 Rates \$ 32.70 \$ 32.88 San \$ 32.70	Fringes 30.73 Fringes 31.68
ALAMEDA, CONTRA COSTA, SAN MA PLASTERER PLAS0300-001 07/01/2018 PLASTERER AREA 188: Fresno AREA 224: San Benito, Santa Clara, Santa Cruz. AREA 295: Calaveras & S Joaquin Couonties AREA 337: Monterey Cour AREA 429: Mariposa, Merced, Stanislaus, Tuolumne Counties	Rates \$ 42.41 \$ 32.70 \$ 32.70 \$ 32.70 \$ 32.70 .ty\$ 32.88 \$ 32.70	Fringes 30.73 Fringes 31.68 31.68 31.68
ALAMEDA, CONTRA COSTA, SAN MA PLASTERER PLAS0300-001 07/01/2018 PLASTERER AREA 188: Fresno AREA 224: San Benito, Santa Clara, Santa Cruz. AREA 295: Calaveras & S Joaquin Couonties AREA 337: Monterey Cour AREA 429: Mariposa, Merced, Stanislaus,	Rates \$ 42.41 \$ 32.70 \$ 32.70 \$ 32.70 \$ 32.70 .ty\$ 32.88 \$ 32.70	Fringes 30.73 Fringes 31.68 31.68 31.68 31.68 31.68
ALAMEDA, CONTRA COSTA, SAN MA PLASTERER PLAS0300-001 07/01/2018 PLASTERER AREA 188: Fresno AREA 224: San Benito, Santa Clara, Santa Cruz. AREA 295: Calaveras & S Joaquin Couonties AREA 337: Monterey Cour AREA 429: Mariposa, Merced, Stanislaus, Tuolumne Counties	Rates \$ 42.41 \$ 32.70 \$ 32.70 \$ 32.70 \$ 32.70 .ty\$ 32.88 \$ 32.70	Fringes 30.73 Fringes 31.68 31.68 31.68 31.68 31.68

https://sam.gov/wage-determination/CA20230018/14

SAN FRANCISCO COUNTY

	Rates	Fringes
PLUMBER (Plumber, Steamfitter, Refrigeration		
Fitter)	\$ 82.00	48.18
PLUM0038-005 07/01/2022		
SAN FRANCISCO COUNTY		
	Rates	Fringes
Landscape/Irrigation Fitter (Underground/Utility Fitter)		33.15
PLUM0062-001 01/01/2023		
MONTEREY AND SANTA CRUZ COUNTIES	5	
	Rates	Fringes
PLUMBER & STEAMFITTER	\$ 50.00	39.35
PLUM0159-001 07/01/2022		
CONTRA COSTA COUNTY		
	Rates	Fringes
Plumber and steamfitter (1) Refrigeration (2) All other work	\$ 56.93 \$ 62.12	41.04 45.24
PLUM0246-001 01/01/2023		
FRESNO, KINGS & MADERA COUNTIES		
	Rates	Fringes
PLUMBER & STEAMFITTER	\$ 46.95	38.59
* PLUM0246-004 01/01/2017		
FRESNO, MERCED & SAN JOAQUIN COU	JNIES	
	Rates	Fringes
PLUMBER (PIPE TRADESMAN)	\$ 13.00 **	10.74
PIPE TRADESMAN SCOPE OF WORK: Installation of corrugated met as installation of corrugated connection with storm sewers a packing and diapering of joint paving over joints, in piping; work for building site prepara pavement breakers, chipping gu to cut holes, chases and chann Digging, grading, backfilling all types of pipe to all point preparation including ground 1 of shrubbery, trees and ground	metal pipin and drains; ts, holes or ; Temporary ation; Opera uns, concret nels for pip and ground ts of the jo leveling, la	g for culverts in Grouting, dry chases including piping for dirt ting jack hammers, se saws and spades ing systems; preparation for obsite; Ground yout and planting

mowing, edging, pruning and fertilizing, the breaking of concrete, digging, backfilling and tamping for the preparation and completion of all work in connection with lawn sprinkler and landscaping; Loading, unloading and distributing materials at jobsite; Putting away materials in storage bins in jobsite secure storage area; Demolition of piping and fixtures for remodeling and additions; Setting up and tearing down work benches, ladders and job shacks; Clean-up and sweeping of jobsite; Pipe wrapping and waterproofing where tar or similar material is applied for protection of buried piping; Flagman

.....

PLUM0342-001 07/01/2022

ALAMEDA & CONTRA COSTA COUNTIES

	Rates	Fringes
PIPEFITTER CONTRA COSTA COUNTY	. \$ 72.00	45.70
PLUMBER, PIPEFITTER,		45.70
STEAMFITTER		
ALAMEDA COUNTY	\$ 72.00	45.70
PLUM0355-004 07/01/2022		

ALAMEDA, CALAVERAS, CONTRA COSTA, FRESNO, KINGS, MADERA, MARIPOSA, MERCED, MONTEREY, SAN BENITO, SAN JOAQUIN, SAN MATEO, SANTA CLARA, SANTA CRUZ, STANISLAUS, AND TUOLUMNE COUNTIES:

Rates Fringes Underground Utility Worker /Landscape Fitter.....\$ 32.22 17.55 PLUM0393-001 07/01/2021

SAN BENITO AND SANTA CLARA COUNTIES

	Rates	Fringes
PLUMBER/PIPEFITTER	\$ 68.76	46.63
PLUM0442-001 01/01/2023		
CALAVERAS, MARIPOSA, MERCED, SA COUNTIES	N JOAQUIN,	STANISLAUS & TUOLUMNE
	Rates	Fringes
PLUMBER & STEAMFITTER	\$ 50.75	35.14
PLUM0467-001 07/01/2021		
SAN MATEO COUNTY		
	Rates	Fringes
Plumber/Pipefitter/Steamfitter.	\$ 73.10	38.61

Plumber/Pipefitter/Steamfitter...\$ 73.10 38.61

ROOF0027-002 01/01/2023

FRESNO, KINGS, AND MADERA COUNTIES

FRESNO, KINGS, AND MADERA COUNTIE	S	
	Rates	Fringes
ROOFER	\$ 41.31	15.31
FOOTNOTE: Work with pitch, pitch base of pitch impregnated products or any material containing coal tar pitch, on any building old or new, where both asphalt and pitchers are used in the application of a built-up roof or tear off: \$2.00 per hour additional.		
ROOF0040-002 08/01/2022		
SAN FRANCISCO & SAN MATEO COUNTIES:		
	Rates	Fringes
ROOFER		21.14
ROOF0081-001 08/01/2022		
ALAMEDA AND CONTRA COSTA COUNTIES	5:	
	Rates	Fringes
Roofer		20.66
ROOF0081-004 08/01/2022		
CALAVERAS, MARIPOSA, MERCED, SAN JOAQUIN, STANISLAUS AND TUOLUMNE COUNTIES:		
	Rates	Fringes
ROOFER		19.71
ROOF0095-002 08/01/2022		
MONTEREY, SAN BENITO, SANTA CLARA, AND SANTA CRUZ COUNTIES:		
	Rates	Fringes
ROOFER Bitumastic, Enameler, Coal Tar, Pitch and Mastic worker Journeyman	\$ 51.16	20.82 20.82
Kettle person (2 kettles)	\$ 53.16	20.82
SFCA0483-001 01/01/2023		
ALAMEDA, CONTRA COSTA, SAN FRANCISCO, SAN MATEO AND SANTA CLARA COUNTIES:		
	Rates	Fringes
SPRINKLER FITTER (FIRE)	\$ 72.59	36.95
SFCA0669-011 04/01/2023		

CALAVERAS, FRESNO, KINGS, MADERA, MARIPOSA, MERCED, MONTEREY,

SAM.gov

SAN BENITO, SAN JOAQUIN, SANTA CRUZ, STANISLAUS AND TUOLUMNE COUNTIES:

Rates Fringes SPRINKLER FITTER.....\$ 44.32 27.25 -----SHEE0104-001 07/01/2020 AREA 1: ALAMEDA, CONTRA COSTA, SAN FRANCISCO, SAN MATEO, SANTA CLARA AREA 2: MONTEREY & SAN BENITO AREA 3: SANTA CRUZ Rates Fringes SHEET METAL WORKER AREA 1: Mechanical Contracts under \$200,000.....\$ 55.92 45.29 All Other Work.....\$ 64.06 46.83 AREA 2.....\$ 52.90 36.44 AREA 3.....\$ 55.16 34.18 _____ SHEE0104-003 07/01/2021 CALAVERAS AND SAN JOAQUIN COUNTIES: Rates Fringes SHEET METAL WORKER.....\$ 44.34 39.22 _____ SHEE0104-005 07/01/2021 MARIPOSA, MERCED, STANISLAUS AND TUOLUMNE COUNTIES: Rates Fringes SHEET METAL WORKER (Excluding metal deck and siding).....\$ 41.28 45.41 SHEE0104-007 07/01/2021 FRESNO, KINGS, AND MADERA COUNTIES: Rates Fringes SHEET METAL WORKER.....\$ 44.07 40.79 SHEE0104-015 07/01/2020 ALAMEDA, CONTRA COSTA, MONTEREY, SAN BENITO, SAN FRANCISCO, SAN MATEO, SANTA CLARA AND SANTA CRUZ COUNTIES: Fringes Rates SHEET METAL WORKER (Metal Decking and Siding only).....\$ 44.45 35.55 -----SHEE0104-018 07/01/2020
CALAVERAS, FRESNO, KINGS, MADERA, MARIPOSA, MERCED, SAN JOAQUIN, STANISLAUS AND TUOLUMNE COUNTIES:

	Rates	Fringes
Sheet metal worker (Metal decking and siding only)	.\$ 44.45	35.55
TEAM0094-001 07/01/2022		
	Rates	Fringes
Truck drivers: GROUP 1 GROUP 2 GROUP 3 GROUP 4 GROUP 5	.\$ 37.25 .\$ 37.55 .\$ 37.90	31.14 31.14 31.14 31.14 31.14 31.14

FOOTNOTES:

Articulated dump truck; Bulk cement spreader (with or without auger); Dumpcrete truck; Skid truck (debris box); Dry pre-batch concrete mix trucks; Dumpster or similar type; Slurry truck: Use dump truck yardage rate. Heater planer; Asphalt burner; Scarifier burner; Industrial lift truck (mechanical tailgate); Utility and clean-up truck: Use appropriate rate for the power unit or the equipment utilized.

TRUCK DRIVER CLASSIFICATIONS

GROUP 1: Dump trucks, under 6 yds.; Single unit flat rack (2axle unit); Nipper truck (when flat rack truck is used appropriate flat rack shall apply); Concrete pump truck (when flat rack truck is used appropriate flat rack shall apply); Concrete pump machine; Fork lift and lift jitneys; Fuel and/or grease truck driver or fuel person; Snow buggy; Steam cleaning; Bus or personhaul driver; Escort or pilot car driver; Pickup truck; Teamster oiler/greaser and/or serviceperson; Hook tender (including loading and unloading); Team driver; Tool room attendant (refineries)

GROUP 2: Dump trucks, 6 yds. and under 8 yds.; Transit mixers, through 10 yds.; Water trucks, under 7,000 gals.; Jetting trucks, under 7,000 gals.; Single-unit flat rack (3-axle unit); Highbed heavy duty transport; Scissor truck; Rubber-tired muck car (not self-loaded); Rubber-tired truck jumbo; Winch truck and ""A"" frame drivers; Combination winch truck with hoist; Road oil truck or bootperson; Buggymobile; Ross, Hyster and similar straddle carriers; Small rubber-tired tractor

GROUP 3: Dump trucks, 8 yds. and including 24 yds.; Transit mixers, over 10 yds.; Water trucks, 7,000 gals. and over; Jetting trucks, 7,000 gals. and over; Vacuum trucks under 7500 gals. Trucks towing tilt bed or flat bed pull trailers; Lowbed heavy duty transport; Heavy duty transport tiller person; Self- propelled street sweeper with self-contained refuse bin; Boom truck - hydro-lift or Swedish type extension or retracting crane; P.B. or similar type self-loading truck; Tire repairperson; Combination bootperson and road oiler; Dry distribution truck (A bootperson when employed on such equipment, shall receive the rate specified for the classification of road oil trucks or bootperson); Ammonia nitrate distributor, driver and mixer; Snow Go and/or plow

GROUP 4: Dump trucks, over 25 yds. and under 65 yds.; Water pulls - DW 10's, 20's, 21's and other similar equipment when pulling Aqua/pak or water tank trailers; Helicopter pilots (when transporting men and materials); Lowbedk Heavy Duty Transport up to including 7 axles; DW10's, 20's, 21's and other similar Cat type, Terra Cobra, LeTourneau Pulls, Tournorocker, Euclid and similar type equipment when pulling fuel and/or grease tank trailers or other miscellaneous trailers; Vacuum Trucks 7500 gals and over and truck repairman

GROUP 5: Dump trucks, 65 yds. and over; Holland hauler; Low bed Heavy Duty Transport over 7 axles

WELDERS - Receive rate prescribed for craft performing operation to which welding is incidental.

** Workers in this classification may be entitled to a higher minimum wage under Executive Order 14026 (\$16.20) or 13658 (\$12.15). Please see the Note at the top of the wage determination for more information.

Note: Executive Order (EO) 13706, Establishing Paid Sick Leave for Federal Contractors applies to all contracts subject to the Davis-Bacon Act for which the contract is awarded (and any solicitation was issued) on or after January 1, 2017. If this contract is covered by the EO, the contractor must provide employees with 1 hour of paid sick leave for every 30 hours they work, up to 56 hours of paid sick leave each year. Employees must be permitted to use paid sick leave for their own illness, injury or other health-related needs, including preventive care; to assist a family member (or person who is like family to the employee) who is ill, injured, or has other health-related needs, including preventive care; or for reasons resulting from, or to assist a family member (or person who is like family to the employee) who is a victim of, domestic violence, sexual assault, or stalking. Additional information on contractor requirements and worker protections under the EO is available at

https://www.dol.gov/agencies/whd/government-contracts.

Unlisted classifications needed for work not included within the scope of the classifications listed may be added after award only as provided in the labor standards contract clauses (29CFR 5.5 (a) (1) (ii)).

The body of each wage determination lists the classification and wage rates that have been found to be prevailing for the cited type(s) of construction in the area covered by the wage determination. The classifications are listed in alphabetical order of ""identifiers"" that indicate whether the particular rate is a union rate (current union negotiated rate for local),

a survey rate (weighted average rate) or a union average rate (weighted union average rate).

Union Rate Identifiers

A four letter classification abbreviation identifier enclosed in dotted lines beginning with characters other than ""SU"" or ""UAVG"" denotes that the union classification and rate were prevailing for that classification in the survey. Example: PLUM0198-005 07/01/2014. PLUM is an abbreviation identifier of the union which prevailed in the survey for this classification, which in this example would be Plumbers. 0198 indicates the local union number or district council number where applicable, i.e., Plumbers Local 0198. The next number, 005 in the example, is an internal number used in processing the wage determination. 07/01/2014 is the effective date of the most current negotiated rate, which in this example is July 1, 2014.

Union prevailing wage rates are updated to reflect all rate changes in the collective bargaining agreement (CBA) governing this classification and rate.

Survey Rate Identifiers

Classifications listed under the ""SU"" identifier indicate that no one rate prevailed for this classification in the survey and the published rate is derived by computing a weighted average rate based on all the rates reported in the survey for that classification. As this weighted average rate includes all rates reported in the survey, it may include both union and non-union rates. Example: SULA2012-007 5/13/2014. SU indicates the rates are survey rates based on a weighted average calculation of rates and are not majority rates. LA indicates the State of Louisiana. 2012 is the year of survey on which these classifications and rates are based. The next number, 007 in the example, is an internal number used in producing the wage determination. 5/13/2014 indicates the survey completion date for the classifications and rates under that identifier.

Survey wage rates are not updated and remain in effect until a new survey is conducted.

Union Average Rate Identifiers

Classification(s) listed under the UAVG identifier indicate that no single majority rate prevailed for those classifications; however, 100% of the data reported for the classifications was union data. EXAMPLE: UAVG-OH-0010 08/29/2014. UAVG indicates that the rate is a weighted union average rate. OH indicates the state. The next number, 0010 in the example, is an internal number used in producing the wage determination. 08/29/2014 indicates the survey completion date for the classifications and rates under that identifier.

A UAVG rate will be updated once a year, usually in January of each year, to reflect a weighted average of the current negotiated/CBA rate of the union locals from which the rate is based.

1.) Has there been an initial decision in the matter? This can be:

- * an existing published wage determination
- * a survey underlying a wage determination
- * a Wage and Hour Division letter setting forth a position on a wage determination matter
- * a conformance (additional classification and rate) ruling

On survey related matters, initial contact, including requests for summaries of surveys, should be with the Wage and Hour National Office because National Office has responsibility for the Davis-Bacon survey program. If the response from this initial contact is not satisfactory, then the process described in 2.) and 3.) should be followed.

With regard to any other matter not yet ripe for the formal process described here, initial contact should be with the Branch of Construction Wage Determinations. Write to:

Branch of Construction Wage Determinations Wage and Hour Division U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

2.) If the answer to the question in 1.) is yes, then an interested party (those affected by the action) can request review and reconsideration from the Wage and Hour Administrator (See 29 CFR Part 1.8 and 29 CFR Part 7). Write to:

Wage and Hour Administrator U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

The request should be accompanied by a full statement of the interested party's position and by any information (wage payment data, project description, area practice material, etc.) that the requestor considers relevant to the issue.

3.) If the decision of the Administrator is not favorable, an interested party may appeal directly to the Administrative Review Board (formerly the Wage Appeals Board). Write to:

Administrative Review Board U.S. Department of Labor 200 Constitution Avenue, N.W. Washington, DC 20210

4.) All decisions by the Administrative Review Board are final.

END OF GENERAL DECISIO"

"REGISTER OF WAGE DETERMINATIONTE SERVICE CONTRACT ACT	DNS UNDER U.S. DEPARTMENT OF LABOR
By direction of the Secretary of Labor	WAGE AND HOUR DIVISION
by direction of the secretary of Labor	
	WASHINGTON D.C. 20210
	Wage Determination No.: 2015-5609
Daniel W. Simms Division of	Revision No.: 20
Director Wage Determinations	Date Of Last Revision: 12/27/2022

Note: Contracts subject to the Service Contract Act are generally required to pay at least the applicable minimum wage rate required under Executive Order 14026 or Executive Order 13658.

If the contract is entered into on or after January 30, 2022, or the contract is renewed or extended (e.g., an option is exercised) on or after January 30, 2022:	Executive Order 14026 generally applies to the contract. The contractor must pay all covered workers at least \$16.20 per hour (or the applicable wage rate listed on this wage determination, if it is higher) for all hours spent performing on the contract in 2023.
<pre>If the contract was awarded on or between January 1, 2015 and January 29, 2022, and the contract is not renewed or extended on or after January 30, 2022:</pre>	Executive Order 13658 generally applies to the contract. The contractor must pay all covered workers at least \$12.15 per hour (or the applicable wage rate listed on this wage determination, if it is higher) for all hours spent performing on the contract in 2023.

The applicable Executive Order minimum wage rate will be adjusted annually. Additional information on contractor requirements and worker protections under the Executive Orders is available at www.dol.gov/whd/govcontracts.

State: California

Area: California County of Fresno

Fringe	Benefits	Required	Follow	the	Occupational	Listing

OCCUPATION CODE - TITLE	FOOTNOTE	RATE
01000 - Administrative Support And Clerical Occupations		
01011 - Accounting Clerk I		17.47
01012 - Accounting Clerk II		19.60
01013 - Accounting Clerk III		21.93
01020 - Administrative Assistant		28.95
01035 - Court Reporter		37.03
01041 - Customer Service Representative I		14.56***
01042 - Customer Service Representative II		15.89***
01043 - Customer Service Representative III		17.83
01051 - Data Entry Operator I		15.25***
01052 - Data Entry Operator II		16.64
01060 - Dispatcher, Motor Vehicle		19.92
01070 - Document Preparation Clerk		17.70
01090 - Duplicating Machine Operator		17.70
01111 - General Clerk I		14.36***
01112 - General Clerk II		15.66***
01113 - General Clerk III		17.58

5/30/23, 11:01 AM	SAM.gov
01120 - Housing Referral Assistant	21.02
01141 - Messenger Courier	16.94
01191 - Order Clerk I	16.27
01192 - Order Clerk II	17.75
01261 - Personnel Assistant (Employment) I	17.43
01262 - Personnel Assistant (Employment) II 01263 - Personnel Assistant (Employment) III	19.50 21.74
01270 - Production Control Clerk	22.26
01290 - Rental Clerk	15.64***
01300 - Scheduler, Maintenance	16.85
01311 - Secretary I	16.85
01312 - Secretary II	18.85
01313 - Secretary III 01320 - Service Orden Dispetator	21.02
01320 - Service Order Dispatcher 01410 - Supply Technician	19.07 28.95
01420 - Survey Worker	20.35
01460 - Switchboard Operator/Receptionist	16.70
01531 - Travel Clerk I	16.89
01532 - Travel Clerk II	17.77
01533 - Travel Clerk III	19.24
01611 - Word Processor I	17.33
01612 - Word Processor II 01613 - Word Processor III	19.45 21.76
05000 - Automotive Service Occupations	21.70
05005 - Automobile Body Repairer, Fiberglass	21.44
05010 - Automotive Electrician	21.34
05040 - Automotive Glass Installer	20.11
05070 - Automotive Worker	20.11
05110 - Mobile Equipment Servicer	17.70
05130 - Motor Equipment Metal Mechanic	22.55
05160 - Motor Equipment Metal Worker 05190 - Motor Vehicle Mechanic	20.12 22.55
05220 - Motor Vehicle Mechanic Helper	16.50
05250 - Motor Vehicle Upholstery Worker	18.92
05280 - Motor Vehicle Wrecker	20.11
05310 - Painter, Automotive	21.34
05340 - Radiator Repair Specialist	20.11
05370 - Tire Repairer 05400 - Transmission Repair Specialist	16.03*** 22.55
07000 - Food Preparation And Service Occupations	22.55
07010 - Baker	15.70***
07041 - Cook I	16.63
07042 - Cook II	18.90
07070 - Dishwasher	14.00***
07130 - Food Service Worker	14.48***
07210 - Meat Cutter 07260 - Waiter/Waitress	17.36 14.00***
09000 - Furniture Maintenance And Repair Occupations	14.00
09010 - Electrostatic Spray Painter	20.92
09040 - Furniture Handler	14.08***
09080 - Furniture Refinisher	19.91
09090 - Furniture Refinisher Helper	16.17***
09110 - Furniture Repairer, Minor	18.55
09130 - Upholsterer 11000 - General Services And Support Occupations	19.91
11030 - Cleaner, Vehicles	14.82***
11060 - Elevator Operator	14.82***
11090 - Gardener	19.58
11122 - Housekeeping Aide	14.87***
11150 - Janitor	14.87***
11210 - Laborer, Grounds Maintenance	16.85
11240 - Maid or Houseman	14.56*** 16.09***
11260 - Pruner 11270 - Tractor Operator	18.38
11330 - Trail Maintenance Worker	16.85
	10.05

5/30/23, 11:01 AM	SAM.gov	
11360 - Window Cleaner	- 15.0	62***
12000 - Health Occupations		
12010 - Ambulance Driver	:	18.28
12011 - Breath Alcohol Technician		24.39
12012 - Certified Occupational Therapist Assistant		34.72
12015 - Certified Physical Therapist Assistant		35.34
12020 - Dental Assistant		18.06
12025 - Dental Hygienist 12030 - EKG Technician		44.44 35.87
12035 - Electroneurodiagnostic Technologist		35.87
12040 - Emergency Medical Technician		18.28
12071 - Licensed Practical Nurse I		21.79
12072 - Licensed Practical Nurse II	:	24.39
12073 - Licensed Practical Nurse III	:	27.19
12100 - Medical Assistant		17.70
12130 - Medical Laboratory Technician		31.31
12160 - Medical Record Clerk		17.61
12190 - Medical Record Technician		19.71
12195 - Medical Transcriptionist 12210 - Nuclear Medicine Technologist		20.13 58.32
12221 - Nursing Assistant I		31***
12222 - Nursing Assistant II		84***
12223 - Nursing Assistant III		10***
12224 - Nursing Assistant IV		16.95
12235 - Optical Dispenser	:	22.14
12236 - Optical Technician		21.79
12250 - Pharmacy Technician		21.32
12280 - Phlebotomist		21.41
12305 - Radiologic Technologist		39.50
12311 - Registered Nurse I		29.00
12312 - Registered Nurse II 12313 - Registered Nurse II, Specialist		35.48 35.48
12314 - Registered Nurse III		45.16
12315 - Registered Nurse III, Anesthetist		45.16
12316 - Registered Nurse IV		51.45
12317 - Scheduler (Drug and Alcohol Testing)		30.21
12320 - Substance Abuse Treatment Counselor	:	23.55
13000 - Information And Arts Occupations		
13011 - Exhibits Specialist I		31.31
13012 - Exhibits Specialist II		38.77
13013 - Exhibits Specialist III 13041 - Illustrator I		47.42 31.31
13042 - Illustrator II		38.77
13043 - Illustrator III		47.42
13047 - Librarian		42.94
13050 - Library Aide/Clerk	:	17.09
13054 - Library Information Technology Systems		39.43
Administrator		
13058 - Library Technician		21.35
13061 - Media Specialist I		27.97
13062 - Media Specialist II 13063 - Media Specialist III		31.31 34.90
13071 - Photographer I		19.08
13072 - Photographer II		21.34
13073 - Photographer III		24.63
13074 - Photographer IV		30.82
13075 - Photographer V		37.28
13090 - Technical Order Library Clerk		21.46
13110 - Video Teleconference Technician	:	23.49
14000 - Information Technology Occupations		
14041 - Computer Operator I		19.00
14042 - Computer Operator II 14042 - Computer Operator III		21.26
14043 - Computer Operator III 14044 - Computer Operator IV		23.70 26.34
14044 - Computer Operator IV 14045 - Computer Operator V		20.34

5/30/23, 11:01 AM		SAM.gov	
14071 - Computer Programmer I		(see 1)	22.87
14072 - Computer Programmer II		(see 1)	
14073 - Computer Programmer II		(see 1)	
14074 - Computer Programmer I∖ 14101 - Computer Systems Analy		(see 1)	
14101 - Computer Systems Analy 14102 - Computer Systems Analy		(see 1) (see 1)	
14103 - Computer Systems Analy		(see 1)	
14150 - Peripheral Equipment C		(555 -)	19.00
14160 - Personal Computer Supp			26.34
14170 - System Support Special			32.94
15000 - Instructional Occupation			
15010 - Aircrew Training Devic			31.05
15020 - Aircrew Training Devic			37.56 45.03
15030 - Air Crew Training Devi 15050 - Computer Based Trainir		`	45.05 31.05
15060 - Educational Technologi			46.80
15070 - Flight Instructor (Pil			45.03
15080 - Graphic Artist	,		24.79
15085 - Maintenance Test Pilot	, Fixed, Jet/Prop		45.01
15086 - Maintenance Test Pilot	, Rotary Wing		45.01
15088 - Non-Maintenance Test/C	o-Pilot		45.01
15090 - Technical Instructor			24.53
15095 - Technical Instructor/C	ourse Developer		30.00
15110 - Test Proctor 15120 - Tutor			19.80 19.80
16000 - Laundry, Dry-Cleaning, P	ressing And Related Occur	ations	19.00
16010 - Assembler	ressing And Related Occup		16.41
16030 - Counter Attendant			16.41
16040 - Dry Cleaner			18.75
16070 - Finisher, Flatwork, Ma	chine		16.41
16090 - Presser, Hand			16.41
16110 - Presser, Machine, Dryc			16.41
16130 - Presser, Machine, Shir			16.41
16160 - Presser, Machine, Wear			16.41
16190 - Sewing Machine Operato 16220 - Tailor	r		19.53 20.31
16250 - Washer, Machine			17.19
19000 - Machine Tool Operation A	nd Repair Occupations		17.125
19010 - Machine-Tool Operator			27.63
19040 - Tool And Die Maker			33.89
21000 - Materials Handling And P	acking Occupations		
21020 - Forklift Operator			17.03
21030 - Material Coordinator			22.26
21040 - Material Expediter 21050 - Material Handling Labo	non		22.26 14.95***
21071 - Order Filler			14.74***
21080 - Production Line Worker	(Food Processing)		17.03
21110 - Shipping Packer	、		17.15
21130 - Shipping/Receiving Cle	rk		17.15
21140 - Store Worker I			15.63***
21150 - Stock Clerk			17.97
21210 - Tools And Parts Attend	ant		17.03
21410 - Warehouse Specialist 23000 - Mechanics And Maintenanc	a And Banain Accurations		17.03
23010 - Aerospace Structural W			34.37
23019 - Aircraft Logs and Reco			27.37
23021 - Aircraft Mechanic I			32.63
23022 - Aircraft Mechanic II			34.37
23023 - Aircraft Mechanic III			36.12
23040 - Aircraft Mechanic Help	er		24.04
23050 - Aircraft, Painter			30.89
23060 - Aircraft Servicer	ht Equipment Technicies		27.37
23070 - Aircraft Survival Flig 23080 - Aircraft Worker	ne equipment reconnectan		30.89 29.10
23091 - Aircrew Life Support E	auipment (ALSE) Mechanic		29.10

Ι

23092 - Aircrew Life Support Equipment (ALSE) Mechanic II	32.63
23110 - Appliance Mechanic	28.34
23120 - Bicycle Repairer	16.36
23125 - Cable Splicer	31.01
23130 - Carpenter, Maintenance	27.42
23140 - Carpet Layer	20.72
23160 - Electrician, Maintenance	29.52
23181 - Electronics Technician Maintenance I	27.92
23182 - Electronics Technician Maintenance II	29.63
23183 - Electronics Technician Maintenance III	31.31
23260 - Fabric Worker	25.48
23290 - Fire Alarm System Mechanic 23310 - Fire Extinguisher Repairer	26.09 23.85
23311 - Fuel Distribution System Mechanic	31.49
23312 - Fuel Distribution System Operator	23.84
23370 - General Maintenance Worker	21.45
23380 - Ground Support Equipment Mechanic	32.63
23381 - Ground Support Equipment Servicer	27.37
23382 - Ground Support Equipment Worker	29.10
23391 - Gunsmith I	23.85
23392 - Gunsmith II	27.09
23393 - Gunsmith III	30.38
23410 - Heating, Ventilation And Air-Conditioning	27.45
Mechanic	
23411 - Heating, Ventilation And Air Contidioning	28.92
Mechanic (Research Facility)	20.95
23430 - Heavy Equipment Mechanic 23440 - Heavy Equipment Operator	29.85 31.79
23460 - Instrument Mechanic	28.75
23465 - Laboratory/Shelter Mechanic	28.75
23470 - Laborer	14.95***
23510 - Locksmith	26.26
23530 - Machinery Maintenance Mechanic	28.40
23550 - Machinist, Maintenance	21.76
23580 - Maintenance Trades Helper	16.72
23591 - Metrology Technician I	28.75
23592 - Metrology Technician II	30.29
23593 - Metrology Technician III	31.82
23640 - Millwright	30.38
23710 - Office Appliance Repairer	18.48
23760 - Painter, Maintenance 23790 - Pipefitter, Maintenance	21.92 28.65
23810 - Plumber, Maintenance	28.03
23820 - Pneudraulic Systems Mechanic	30.38
23850 - Rigger	29.59
23870 - Scale Mechanic	27.09
23890 - Sheet-Metal Worker, Maintenance	26.72
23910 - Small Engine Mechanic	22.87
23931 - Telecommunications Mechanic I	28.99
23932 - Telecommunications Mechanic II	30.55
23950 - Telephone Lineman	41.11
23960 - Welder, Combination, Maintenance	21.19
23965 - Well Driller	24.09
23970 - Woodcraft Worker 23980 - Woodworker	30.38 23.85
23980 - Woodworker 24000 - Personal Needs Occupations	20.00
24550 - Case Manager	18.51
24570 - Child Care Attendant	14.41***
24580 - Child Care Center Clerk	17.97
24610 - Chore Aide	14.27***
24620 - Family Readiness And Support Services	18.51
Coordinator	
24630 - Homemaker	18.51

5/30/23, 11:01 AM	SAM.gov	
25000 - Plant And System Operations Occupations		
25010 - Boiler Tender	40.0	96
25040 - Sewage Plant Operator	29.7	
25070 - Stationary Engineer	40.6	
25190 - Ventilation Equipment Tender	29.3	
25210 - Water Treatment Plant Operator 27000 - Protective Service Occupations	29.7	/5
27004 - Alarm Monitor	23.4	49
27007 - Baggage Inspector	14.87**	
27008 - Corrections Officer	36.5	
27010 - Court Security Officer	33.8	80
27030 - Detection Dog Handler	16.6	54
27040 - Detention Officer	36.5	
27070 - Firefighter	32.6	
27101 - Guard I 27102 - Guard II	14.87**	
27102 - Guard II 27131 - Police Officer I	16.6	
27132 - Police Officer II	46.8	
28000 - Recreation Occupations		
28041 - Carnival Equipment Operator	16.9	92
28042 - Carnival Equipment Repairer	18.1	16
28043 - Carnival Worker	14.00**	**
28210 - Gate Attendant/Gate Tender	17.7	
28310 - Lifeguard	14.19**	
28350 - Park Attendant (Aide) 28510 - Recreation Aide/Health Facility Attendant	19.8 14.51**	
28510 - Recreation Alde/Health Facility Attendant 28515 - Recreation Specialist	24.6	
28630 - Sports Official	15.83*	
28690 - Swimming Pool Operator	20.6	
29000 - Stevedoring/Longshoremen Occupational Services		
29010 - Blocker And Bracer	27.6	99
29020 - Hatch Tender	27.6	
29030 - Line Handler	27.6	
29041 - Stevedore I	25.4	
29042 - Stevedore II 30000 - Technical Occupations	28.7	/5
30010 - Air Traffic Control Specialist, Center (HFO)	(see 2) 41.2	26
30011 - Air Traffic Control Specialist, Station (HFO)	(/	
30012 - Air Traffic Control Specialist, Terminal (HFO)		33
30021 - Archeological Technician I	18.2	
30022 - Archeological Technician II	20.4	
30023 - Archeological Technician III	25.3	
30030 - Cartographic Technician 30040 - Civil Engineering Technician	25.3 34.0	
30051 - Cryogenic Technician I	26.7	
30052 - Cryogenic Technician II	29.5	
30061 - Drafter/CAD Operator I	18.2	
30062 - Drafter/CAD Operator II	20.6	99
30063 - Drafter/CAD Operator III	22.8	
30064 - Drafter/CAD Operator IV	28.0	
30081 - Engineering Technician I	16.01**	
30082 - Engineering Technician II 30083 - Engineering Technician III	17.9 20.6	
30084 - Engineering Technician IV	24.9	
30085 - Engineering Technician V	30.4	
30086 - Engineering Technician VI	36.8	
30090 - Environmental Technician	24.6	63
30095 - Evidence Control Specialist	24.2	
30210 - Laboratory Technician	19.8	
30221 - Latent Fingerprint Technician I	28.2	
30222 - Latent Fingerprint Technician II 30240 - Mathematical Technician	31.0 25.3	
30361 - Paralegal/Legal Assistant I	19.9	
30362 - Paralegal/Legal Assistant II	24.0	
30363 - Paralegal/Legal Assistant III	29.4	

5/30/23, 11:01 AM	SAM.gov	
30364 - Paralegal/Legal Assistant IV		35.56
30375 - Petroleum Supply Specialist		29.55
30390 - Photo-Optics Technician		25.36
30395 - Radiation Control Technician		29.55
30461 - Technical Writer I		24.29
30462 - Technical Writer II		29.71
30463 - Technical Writer III		35.94
30491 - Unexploded Ordnance (UXO) Technician I		26.22
30492 - Unexploded Ordnance (UXO) Technician II		31.73
30493 - Unexploded Ordnance (UXO) Technician III		38.03
30494 - Unexploded (UXO) Safety Escort		26.22
30495 - Unexploded (UXO) Sweep Personnel		26.22
30501 - Weather Forecaster I		26.75
30502 - Weather Forecaster II		32.54
30620 - Weather Observer, Combined Upper Air Or	(see 2)	22.82
Surface Programs	· ·	
30621 - Weather Observer, Senior	(see 2)	25.36
31000 - Transportation/Mobile Equipment Operation Occup		
31010 - Airplane Pilot		31.73
31020 - Bus Aide		21.24
31030 - Bus Driver		28.88
31043 - Driver Courier		16.84
31260 - Parking and Lot Attendant		14.25***
31290 - Shuttle Bus Driver		16.83
31310 - Taxi Driver		14.40***
31361 - Truckdriver, Light		17.94
31362 - Truckdriver, Medium		19.17
31363 - Truckdriver, Heavy		22.56
31364 - Truckdriver, Tractor-Trailer		22.56
99000 - Miscellaneous Occupations		
99020 - Cabin Safety Specialist		15.47***
99030 - Cashier		14.00***
99050 - Desk Clerk		14.26***
99095 - Embalmer		29.37
99130 - Flight Follower		26.22
99251 - Laboratory Animal Caretaker I		15.36***
99252 - Laboratory Animal Caretaker II		16.48
99260 - Marketing Analyst		28.90
99310 - Mortician		29.37
99410 - Pest Controller		18.03
99510 - Photofinishing Worker		13.78***
99710 - Recycling Laborer		24.24
99711 - Recycling Specialist		27.84
99730 - Refuse Collector		23.13
99810 - Sales Clerk		14.44***
99820 - School Crossing Guard		17.02
99830 - Survey Party Chief		34.51
99831 - Surveying Aide		22.95
99832 - Surveying Technician		31.37
99840 - Vending Machine Attendant		18.75
99841 - Vending Machine Repairer		22.70
99842 - Vending Machine Repairer Helper		18.75

***Workers in this classification may be entitled to a higher minimum wage under Executive Order 14026 (\$16.20 per hour) or 13658 (\$12.15 per hour). Please see the Note at the top of the wage determination for more information. Please also note that the minimum wage requirements of Executive Order 14026 and 13658 are not currently being enforced as to contracts or contract-like instruments entered into with the federal government in connection with seasonal recreational services or seasonal recreational equipment rental for the general public on federal lands.

Note: Executive Order (EO) 13706, Establishing Paid Sick Leave for Federal Contractors, applies to all contracts subject to the Service Contract Act for which the contract is awarded (and any solicitation was issued) on or after January 1, 2017. If this contract is covered by the EO, the contractor must provide employees with 1 hour of paid sick leave for every 30 hours they work, up to 56 hours of paid sick leave each year. Employees must be permitted to use paid sick leave for their own illness, injury or other health-related needs, including preventive care; to assist a family member (or person who is like family to the employee) who is ill, injured, or has other health-related needs, including preventive care; or for reasons resulting from, or to assist a family member (or person who is like family to the employee) who is the victim of, domestic violence, sexual assault, or stalking. Additional information on contractor requirements and worker protections under the EO is available at www.dol.gov/whd/govcontracts.

ALL OCCUPATIONS LISTED ABOVE RECEIVE THE FOLLOWING BENEFITS:

HEALTH & WELFARE: \$4.80 per hour, up to 40 hours per week, or \$192.00 per week or \$832.00 per month

HEALTH & WELFARE EO 13706: \$4.41 per hour, up to 40 hours per week, or \$176.40 per week, or \$764.40 per month*

*This rate is to be used only when compensating employees for performance on an SCAcovered contract also covered by EO 13706, Establishing Paid Sick Leave for Federal Contractors. A contractor may not receive credit toward its SCA obligations for any paid sick leave provided pursuant to EO 13706.

VACATION: 2 weeks paid vacation after 1 year of service with a contractor or successor, 3 weeks after 5 years, and 4 weeks after 15 years. Length of service includes the whole span of continuous service with the present contractor or successor, wherever employed, and with the predecessor contractors in the performance of similar work at the same Federal facility. (Reg. 29 CFR 4.173)

HOLIDAYS: A minimum of eleven paid holidays per year: New Year's Day, Martin Luther King Jr.'s Birthday, Washington's Birthday, Memorial Day, Juneteenth National Independence Day, Independence Day, Labor Day, Columbus Day, Veterans' Day, Thanksgiving Day, and Christmas Day. (A contractor may substitute for any of the named holidays another day off with pay in accordance with a plan communicated to the employees involved.) (See 29 CFR 4.174)

THE OCCUPATIONS WHICH HAVE NUMBERED FOOTNOTES IN PARENTHESES RECEIVE THE FOLLOWING:

1) COMPUTER EMPLOYEES: This wage determination does not apply to any individual employed in a bona fide executive, administrative, or professional capacity, as defined in 29 C.F.R. Part 541. (See 41 C.F.R. 6701(3)). Because most Computer Systems Analysts and Computer Programmers who are paid at least \$27.63 per hour (or at least \$684 per week if paid on a salary or fee basis) likely qualify as exempt computer professionals under 29 U.S.C. 213(a)(1) and 29 U.S.C. 213(a)(17), this wage determination may not include wage rates for all occupations within those job families. In such instances, a conformance will be necessary if there are nonexempt employees in these job families working on the contract.

Job titles vary widely and change quickly in the computer industry, and are not determinative of whether an employee is an exempt computer professional. To be exempt, computer employees who satisfy the compensation requirements must also have a primary duty that consists of:

(1) The application of systems analysis techniques and procedures, including consulting with users, to determine hardware, software or system functional specifications;

(2) The design, development, documentation, analysis, creation, testing or modification of computer systems or programs, including prototypes, based on and

related to user or system design specifications;

(3) The design, documentation, testing, creation or modification of computer programs related to machine operating systems; or

(4) A combination of the aforementioned duties, the performance of which requires the same level of skills. (29 C.F.R. 541.400).

Any computer employee who meets the applicable compensation requirements and the above duties test qualifies as an exempt computer professional under both section 13(a)(1) and section 13(a)(17) of the Fair Labor Standards Act. (Field Assistance Bulletin No. 2006-3 (Dec. 14, 2006)). Accordingly, this wage determination will not apply to any exempt computer employee regardless of which of these two exemptions is utilized.

2) AIR TRAFFIC CONTROLLERS AND WEATHER OBSERVERS - NIGHT PAY & SUNDAY PAY: If you work at night as part of a regular tour of duty, you will earn a night differential and receive an additional 10% of basic pay for any hours worked between 6pm and 6am. If you are a full-time employed (40 hours a week) and Sunday is part of your regularly scheduled workweek, you are paid at your rate of basic pay plus a Sunday premium of 25% of your basic rate for each hour of Sunday work which is not overtime (i.e. occasional work on Sunday outside the normal tour of duty is considered overtime work).

** HAZARDOUS PAY DIFFERENTIAL **

An 8 percent differential is applicable to employees employed in a position that represents a high degree of hazard when working with or in close proximity to ordnance, explosives, and incendiary materials. This includes work such as screening, blending, dying, mixing, and pressing of sensitive ordnance, explosives, and pyrotechnic compositions such as lead azide, black powder and photoflash powder. All dry-house activities involving propellants or explosives. Demilitarization, modification, renovation, demolition, and maintenance operations on sensitive ordnance, explosives and incendiary materials. All operations involving re-grading and cleaning of artillery ranges.

A 4 percent differential is applicable to employees employed in a position that represents a low degree of hazard when working with, or in close proximity to ordnance, (or employees possibly adjacent to) explosives and incendiary materials which involves potential injury such as laceration of hands, face, or arms of the employee engaged in the operation, irritation of the skin, minor burns and the like; minimal damage to immediate or adjacent work area or equipment being used. All operations involving, unloading, storage, and hauling of ordnance, explosive, and incendiary ordnance material other than small arms ammunition. These differentials are only applicable to work that has been specifically designated by the agency for ordnance, explosives, and incendiary material differential pay.

** UNIFORM ALLOWANCE **

If employees are required to wear uniforms in the performance of this contract (either by the terms of the Government contract, by the employer, by the state or local law, etc.), the cost of furnishing such uniforms and maintaining (by laundering or dry cleaning) such uniforms is an expense that may not be borne by an employee where such cost reduces the hourly rate below that required by the wage determination. The Department of Labor will accept payment in accordance with the following standards as compliance:

The contractor or subcontractor is required to furnish all employees with an adequate number of uniforms without cost or to reimburse employees for the actual cost of the uniforms. In addition, where uniform cleaning and maintenance is made the responsibility of the employee, all contractors and subcontractors subject to this wage determination shall (in the absence of a bona fide collective bargaining agreement providing for a different amount, or the furnishing of contrary affirmative proof as to the actual cost), reimburse all employees for such cleaning

and maintenance at a rate of \$3.35 per week (or \$.67 cents per day). However, in those instances where the uniforms furnished are made of ""wash and wear"" materials, may be routinely washed and dried with other personal garments, and do not require any special treatment such as dry cleaning, daily washing, or commercial laundering in order to meet the cleanliness or appearance standards set by the terms of the Government contract, by the contractor, by law, or by the nature of the work, there is no requirement that employees be reimbursed for uniform maintenance costs.

** SERVICE CONTRACT ACT DIRECTORY OF OCCUPATIONS **

The duties of employees under job titles listed are those described in the ""Service Contract Act Directory of Occupations"", Fifth Edition (Revision 1), dated September 2015, unless otherwise indicated.

** REQUEST FOR AUTHORIZATION OF ADDITIONAL CLASSIFICATION AND WAGE RATE, Standard Form 1444 (SF-1444) **

Conformance Process:

The contracting officer shall require that any class of service employee which is not listed herein and which is to be employed under the contract (i.e., the work to be performed is not performed by any classification listed in the wage determination), be classified by the contractor so as to provide a reasonable relationship (i.e., appropriate level of skill comparison) between such unlisted classifications and the classifications listed in the wage determination (See 29 CFR 4.6(b)(2)(i)). Such conforming procedures shall be initiated by the contractor prior to the performance of contract work by such unlisted class(es) of employees (See 29 CFR 4.6(b)(2)(ii)). The Wage and Hour Division shall make a final determination of conformed classification, wage rate, and/or fringe benefits which shall be paid to all employees performing in the classification from the first day of work on which contract work is performed by them in the classification. Failure to pay such unlisted employees the compensation agreed upon by the interested parties and/or fully determined by the Wage and Hour Division retroactive to the date such class of employees commenced contract work shall be a violation of the Act and this contract. (See 29 CFR 4.6(b)(2)(v)). When multiple wage determinations are included in a contract, a separate SF-1444 should be prepared for each wage determination to which a class(es) is to be conformed.

The process for preparing a conformance request is as follows:

1) When preparing the bid, the contractor identifies the need for a conformed occupation(s) and computes a proposed rate(s).

2) After contract award, the contractor prepares a written report listing in order the proposed classification title(s), a Federal grade equivalency (FGE) for each proposed classification(s), job description(s), and rationale for proposed wage rate(s), including information regarding the agreement or disagreement of the authorized representative of the employees involved, or where there is no authorized representative, the employees themselves. This report should be submitted to the contracting officer no later than 30 days after such unlisted class(es) of employees performs any contract work.

3) The contracting officer reviews the proposed action and promptly submits a report of the action, together with the agency's recommendations and pertinent information including the position of the contractor and the employees, to the U.S. Department of Labor, Wage and Hour Division, for review (See 29 CFR 4.6(b)(2)(ii)).

4) Within 30 days of receipt, the Wage and Hour Division approves, modifies, or disapproves the action via transmittal to the agency contracting officer, or notifies the contracting officer that additional time will be required to process the request.

5) The contracting officer transmits the Wage and Hour Division's decision to the contractor.

6) Each affected employee shall be furnished by the contractor with a written copy of such determination or it shall be posted as a part of the wage determination (See 29 CFR 4.6(b)(2)(iii)).

Information required by the Regulations must be submitted on SF-1444 or bond paper.

When preparing a conformance request, the ""Service Contract Act Directory of Occupations"" should be used to compare job definitions to ensure that duties requested are not performed by a classification already listed in the wage determination. Remember, it is not the job title, but the required tasks that determine whether a class is included in an established wage determination. Conformances may not be used to artificially split, combine, or subdivide classifications listed in the wage determination (See 29 CFR 4.152(c)(1))."

EXHIBIT E DISCLOSURE OF CONFLICT OF INTEREST

Senior Activity Center

		YES*	NO
1	Are you currently in litigation with the City of Fresno or any of its agents?		X
2	Do you represent any firm, organization, or person who is in litigation with the City of Fresno?		×
3	Do you currently represent or perform work for any clients who do business with the City of Fresno?		X
4	Are you or any of your principals, managers, or professionals, owners or investors in a business which does business with the City of Fresno, or in a business which is in litigation with the City of Fresno? (No)	ø	
5	Are you or any of your principals, managers, or professionals, related by blood or marriage to any City of Fresno employee who has any significant role in the subject matter of this service?		X
6	Do you or any of your subcontractors have, or expect to have, any interest, direct or indirect, in any other contract in connection with this Project?	×	
* If t	he answer to any question is yes, please explain in full below.		

Explanation: BLH Construction

and ICD do work in

Fusno.

Date Charles Brunbaugh
Charles From Downeh
Charles Wan val
Name
BLA, ICD
Company
20750 Ventura Blud #155
Address
Woodland Hills, CA 91364
City, State, Zip

□ Additional page(s) attached.

DPW-S Eng. CSA, Long Form Total Fee – Contingency (11-2022) Exhibit E Page 1 of 1

Standard Form of General Conditions of Contract Between Owner and Design-Builder

Document No. 535 Third Edition, 2022

© Design-Build Institute of America Washington, D.C.





Design-Build Institute of America - Contract Documents LICENSE AGREEMENT

By using the DBIA Contract Documents, you agree to and are bound by the terms of this License Agreement.

- License. The Design-Build Institute of America ("DBIA") provides DBIA Contract Documents and licenses their use worldwide. You acknowledge that DBIA Contract Documents are protected by the copyright laws of the United States. You have a limited nonexclusive license to: (a) Use DBIA Contract Documents on any number of machines owned, leased or rented by your company or organization; (b) Use DBIA Contract Documents in printed form for bona fide contract purposes; and (c) Copy DBIA Contract Documents into any machine-readable or printed form for backup or modification purposes in support of your permitted use.
- 2. User Responsibility. You assume sole responsibility for the selection of specific documents or portions thereof to achieve your intended results, and for the installation, use, and results obtained from the DBIA Contract Documents. You acknowledge that you understand that the text of the DBIA Contract Documents has important legal consequences and that consultation with an attorney is recommended with respect to use or modification of the text. You will not represent that any of the contract documents you generate from DBIA Contract Documents are DBIA documents unless (a) the document text is used without alteration or (b) all additions and changes to, and deletions from, the text are clearly shown.
- 3. Copies. You may not use, copy, modify, or transfer DBIA Contract Documents, or any copy, modification or merged portion, in whole or in part, except as expressly provided for in this license. Reproduction of DBIA Contract Documents in printed or machine-readable format for resale or educational purposes is expressly prohibited. You will reproduce and include DBIA's copyright notice on any printed or machine-readable copy, modification, or portion merged into another document or program.
- 4. Transfers. You may not transfer possession of any copy, modification or merged portion of DBIA Contract Documents to another party, except that a party with whom you are contracting may receive and use such transferred material solely for purposes of its contract with you. You may not sublicense, assign, or transfer this license except as expressly provided in this Agreement, and any attempt to do so is void.
- 5. Term. The license is effective for one year from the date of purchase. DBIA may elect to terminate it earlier, by written notice to you, if you fail to comply with any term or condition of this Agreement.
- 6. Limited Warranty. DBIA warrants the electronic files or other media by which DBIA Contract Documents are furnished to be free from defects in materials and workmanship under normal use during the Term. There is no other warranty of any kind, expressed or implied, including, but not limited to the implied warranties of merchantability and fitness for a particular purpose. Some states do not allow the exclusion of implied warranties, so the above exclusion may not apply to you. This warranty gives you specific legal rights and you may also have other rights which vary from state to state. DBIA does not warrant that the DBIA Contract Documents will meet your requirements or that the operation of DBIA Contract Documents will be uninterrupted or error free.
- 7. Limitations of Remedies. DBIA's entire liability and your exclusive remedy shall be: the replacement of any document not meeting DBIA's "Limited Warranty" which is returned to DBIA with a copy of your receipt, or at DBIA's election, your money will be refunded. In no event will DBIA be liable to you for any damages, including any lost profits, lost savings or other incidental or consequential damages arising out of the use or inability to use DBIA Contract Documents even if DBIA has been advised of the possibility of such damages, or for any claim by any other party. Some states do not allow the limitation or exclusion of liability for incidental or consequential damages, so the above limitation or exclusion may not apply to you.
- 8. Acknowledgment. You acknowledge that you have read this agreement, understand it and agree to be bound by its terms and conditions and that it will be governed by the laws of the District of Columbia. You further agree that it is the complete and exclusive statement of your agreement with DBIA which supersedes any proposal or prior agreement, oral or written, and any other communications between the parties relating to the subject matter of this agreement.

INSTRUCTIONS

For DBIA Document No. 535 Standard Form of General Conditions of Contract Between Owner and Design-Builder (2022 Edition)

General Instructions

No.	Subject	Instruction
1.	Standard Forms	Standard form contracts have long served an important function in the United States and international construction markets. The common purpose of these forms is to provide an economical and convenient way for parties to contract for design and construction services. As standard forms gain acceptance and are used with increased frequency, parties are able to enter into contracts with greater certainty as to their rights and responsibilities.
2.	DBIA Standard Form Contract Documents	Since its formation in 1993, the Design-Build Institute of America (DBIA) has regularly evaluated the needs of Owners, Design-Builders, and other parties to the design-build process in preparation for developing its own contract forms. Consistent with DBIA's mission of promulgating best design-build practices, DBIA believes that the design-build contract should reflect a balanced approach to risk that considers the legitimate interests of all parties to the design-build process. DBIA's Standard Form Contract Documents reflect a modern risk allocation approach, allocating each risk to the party best equipped to manage and minimize that risk, with the goal of promoting best design-build practices.
3.	Use of Non- DBIA Documents	To avoid inconsistencies among documents used for the same project, DBIA's Standard Form Contract Documents should not be used in conjunction with non-DBIA documents unless the non-DBIA documents are appropriately modified on the advice of legal counsel. Moreover, care should also be taken when using different editions of the DBIA Standard Form Document on the same project to ensure consistency.
4.	Legal Consequences	DBIA Standard Form Contract Documents are legally binding contracts with important legal consequences. Contracting parties are advised and encouraged to seek legal counsel in completing or modifying these Documents.
5.	Reproduction	DBIA hereby grants to purchasers a limited license to reproduce its Documents consistent with the License Agreement accompanying these Documents. At least two original versions of the Agreement should be signed by the parties. Any other reproduction of DBIA Documents is strictly prohibited.
6.	Modifications	Effective contracting is accomplished when the parties give specific thought to their contracting goals and then tailor the contract to meet the unique needs of the project and the design-build team. For that reason, these Documents may require modification for various purposes including, for example, to comply with local codes and laws, or to add special terms. DBIA's latest revisions to its Documents provide the parties an opportunity to customize their contractual relationship by selecting various optional contract clauses that may better reflect the unique needs and risks associated with the project. Any modifications to these Documents should be initialed by the parties. At no time should a document be re-typed in its entirety. Re-creating the document violates copyright laws and destroys one of the advantages of standard forms – familiarity with the terms.
7.	Execution	It is good practice to execute two original copies of the Agreement. Only persons authorized to sign for the contracting parties may execute the Agreement.

Specific Instructions

Section	Title	Instruction
General	Purpose of This Document	The General Conditions of Contract provide the terms and conditions under which the Work of the Project will be performed. This document accompanies DBIA Document No. 525 and DBIA Document No. 530 (each referred to herein generally as "Agreement"). It may also be incorporated by reference into other related agreements, as between Design-Builder and Design Consultant, and Design-Builder and Subcontractor.
General	Checklist	The following Sections reference documents that are to be attached to the Agreement: Section 3.5.1 Owner's Permit List Article 5 Insurance and Bonds Section 9.4.2 Unit Prices
2.1.3	Schedule	The parties are encouraged, if possible, to agree to a schedule for the execution of the Work upon execution of the Agreement or upon establishing the GMP.
2.2.1	Design Professional Services	The parties should be aware that in addition to requiring compliance with state licensing laws for design professionals, some states also require that the design professional have a corporate professional license.
2.3.1	Standard of Care for Design Professional's Services	Design-Builder's obligation is to deliver a design that meets prevailing industry standards. However, DBIA has provided the parties at Article 11 of the Agreement an optional provision whereby if Owner can identify specific performance standards that can be objectively measured, Design-Builder is obligated to design the Project to satisfy these standards if this optional provision is selected. To avoid any confusion and to ensure that the parties fully understand what their obligations are, the specific performance standards should be clearly identified and should be able to be objectively measured. Design-Builder should recognize that this is a heightened standard of care that has insurance ramifications that should be discussed with Design-Builder's insurance advisor.
3.5.1	Government Approvals and Permits	Design-Builder is responsible for obtaining all necessary permits, approvals, and licenses, except to the extent specific permits, approvals, and licenses are set forth in an Owner's Permit List, which must be attached as an exhibit to the Agreement. The parties, prior to execution of the Agreement, should discuss which permits, approvals and licenses need to be obtained for the Project and which party is in the best position to do so.
5.1.1	Design-Builder's Insurance Requirements	Design-Builder is obligated to provide insurance coverage from insurance carriers that meet the criteria set forth in the Insurance Exhibit attached to Section 10.1 of the Agreement.
5.1.2	Exclusions to Design-Build	Parties are advised that their standard insurance policies may contain exclusions for the design-build delivery method. This Section 5.1.2 requires that any such exclusions be deleted from the policy.
5.2	Owner's Insurance Requirements	Owner, in addition to providing the insurance set forth in this Section and Section 5.3, is also obligated to procure the insurance coverages for the amounts and consistent with the terms set forth in the Insurance Exhibit made part of the Agreement.
5.4	Bonds and Other Performance Security	Design-Builder is only obligated to provide bonds or other forms of performance security to the extent called for in Section 10.2 of the Agreement.

Section	Title	Instruction
8.2.2	Compensability for Force Majeure Events	The parties are provided the option in the Agreement of negotiating whether Design-Builder is entitled to compensation for Force Majeure Events.
9.4.1	Contract Price Adjustments	Unit prices, if established, shall be attached pursuant to Article 2 of the Agreement.
9.4.3	Payment/ Performance of Disputed Services	When Owner disputes Design-Builder's entitlement to a change order or disagrees with Design-Builder regarding the scope of Work, and nevertheless expects Design-Builder to perform the services, Design-Builder's cash flow and ability to complete the Work will be hampered if Owner fails to pay Design-Builder for the disputed services. This Section provides a balanced approach whereby Design-Builder is required to perform the services, but Owner is required to pay fifty percent (50%) of Design-Builder's reasonable estimated direct costs of performing such services until the dispute is settled. By so doing, Owner does not forfeit its right to deny total responsibility for payment, and Design-Builder does not give up its right to demand full payment. The dispute shall be resolved according to Article 10.
Article 10	Contract Adjustments and Disputes	DBIA endorses the use of partnering, negotiation, mediation and arbitration for the prevention and resolution of disputes. The General Conditions of Contract provides for the parties' Representatives and Senior Representatives to attempt to negotiate the dispute or disagreement. If this attempt fails, the dispute shall be submitted to mandatory, non-binding mediation. Any dispute that cannot be resolved by mediation shall then be submitted to binding arbitration, unless the parties elect in the Agreement to submit their dispute to a court of competent jurisdiction.
10.3.4	Arbitration	The prevailing party in any arbitration shall receive reasonable attorneys' fees from the other party. DBIA supports this "loser pays" provision to encourage parties to negotiate or mediate their differences and to minimize the number of frivolous disputes.
10.4	Duty to Continue Performance	Pending the resolution of any dispute or disagreement, both Owner and Design-Builder shall continue to perform their respective duties under the Contract Documents, unless the parties provide otherwise in the Contract Documents.
10.5	Consequential Damages	DBIA believes that it is inappropriate for either Owner or Design-Builder to be responsible to the other for consequential damages arising from the Project. This limitation on consequential damages in no way restricts, however, the payment of liquidated damages, if any, under Article 5 of the Agreement.
11.4	Design-Builder's Right to Terminate for Cause	If Design-Builder properly terminates the Agreement for cause, it shall recover from Owner in the same way as if Owner had terminated the Agreement for convenience under Section 11.6 of the General Conditions. Owner shall pay to Design-Builder its costs, reasonable overhead and profit on the costs, and an additional payment based on a percentage of the remaining balance of the Contract Price, all as set forth in Article 8 of the Agreement.
11.6.2	Termination for Convenience: Owner's Use of Work Product	Owner should not use the Termination for Convenience clause to obtain Design-Builder's valuable design concepts and then seek lower bids from another design-builder. If Owner terminates this Agreement for its own convenience, and chooses to proceed with the Project using Design-Builder's Work Product, Owner should pay an additional sum for the use of Design-Builder's Work Product pursuant to Section 4.3 of the Agreement.
Article 12	Electronic Data	Design-Builder and Owner shall agree on the software and format for the transmission of Electronic Data. Ownership of Work Product in electronic form is governed by Article 4 of the Agreement. The transmitting party disclaims all warranties with respect to the media transmitting the Electronic Data, but nothing in this Article is intended to negate duties with respect to the standard of care in creating the Electronic Data.

TABLE OF CONTENTS

Article	Name P	Page
Article 1	General	1
Article 2	Design-Builder's Services and Responsibilities	3
Article 3	Owner's Services and Responsibilities	7
Article 4	Hazardous Conditions and Differing Site Conditions	9
Article 5	Insurance and Bonds	10
Article 6	Payment	12
Article 7	Indemnification	14
Article 8	Time	16
Article 9	Changes to the Contract Price and Time	17
Article 10	0 Contract Adjustments and Disputes	18
Article 12	1 Stop Work and Termination	20
Article 12	2 Electronic Data	23
Article 13	3 Miscellaneous	24

Article 1

General

1.1 Mutual Obligations.

1.1.1 Owner and Design-Builder commit at all times to cooperate fully with each other and proceed on the basis of trust and good faith, to permit each party to realize the benefits afforded under the Contract Documents.

1.2 Basic Definitions.

1.2.1 Agreement refers to the executed contract between Owner and Design-Builder under either DBIA Document No. 525, Standard Form of Agreement Between Owner and Design-Builder – Lump Sum (2022 Edition); DBIA Document No. 530, Standard Form of Agreement Between Owner and Design-Builder – Cost Plus Fee With an Option for a Guaranteed Maximum Price (2022 Edition); DBIA Document No. 544, Standard Form of Progressive Design-Build Agreement (2022 Edition); or DBIA Document No. 545, Standard Form of Progressive Design-Build Agreement for Water and Wastewater Projects (2022 Edition).

1.2.2 Basis of Design Documents are as follows: For DBIA Document No. 530, Standard Form of Agreement Between Owner and Design-Builder – Cost Plus Fee With an Option for a Guaranteed Maximum Price, the Basis of Design Documents are those documents specifically listed in, as applicable, the GMP Exhibit or GMP Proposal as being the "Basis of Design Documents." For DBIA Document No. 525, Standard Form of Agreement Between Owner and Design-Builder – Lump Sum, for DBIA Document No. 544, Standard Form of Progressive Design-Build Agreement, the Basis of Design Documents are Owner's Project Criteria, Design-Builder's Proposal, and the Deviation List, if any. For DBIA Document No. 545, Standard Form of Progressive Design Documents are Owner's Projects, the Basis of Design Documents are Owner's Project Criteria, Design-Builder's Proposal and the Deviation List, if any.

1.2.3 California Building Code (CBC), Latest Edition; California Plumbing Code (CPC), Latest Edition; California Mechanical Code (CMC), Latest Edition; National Electrical Code (NEC), Latest Edition; California Fire Code (CFC), Latest Edition; California Health and Safety Code (as applicable). For purposes of this definition, "Latest Edition" shall mean the edition, and to the extent, adopted by the City through the City of Fresno Municipal Code.

1.2.4 *City, Buyer, Owner, Vendee, City of Fresno* shall each mean and refer to the City of Fresno, California.

1.2.5 *City Standard Specifications* - City of Fresno, Standard Specifications, Department of Public Works, dated September 2010 and as amended from time-to-time.

1.2.6 *Construction Documents* are the documents, consisting of Drawings and Specifications, to be prepared or assembled by Design-Builder consistent with the Basis of Design Documents unless a deviation from the Basis of Design Documents is specifically set forth in a Change Order executed by both Owner and Design-Builder, as part of the design review process contemplated by Section 2.4 of these General Conditions of Contract.

1.2.7 *Construction Manager* shall mean and refer to the Owner's authorized representative at the Job Site, in responsible charge of administering the Contract. The Construction Manager shall be the single point of contact for all correspondence, submittals, progress payment requests, and contacts to and from the Contractor.

1.2.8 *Council, City Council* shall each mean and refer to the Council of the Buyer.

1.2.9 Day or Days shall mean calendar days unless otherwise specifically noted in the Contract

Documents.

1.2.10 *Design-Build Team* is comprised of Design-Builder, Design Consultant, and key Subcontractors identified by Design-Builder.

1.2.11 Design Consultant is a qualified, licensed design professional who is not an employee of Design-Builder, but is retained by Design-Builder, or employed or retained by anyone under contract with Design-Builder, to furnish design services required under the Contract Documents. A Design Sub-Consultant is a qualified, licensed design professional who is not an employee of Design Consultant but is retained by Design Consultant or employed or retained by anyone under contract to Design Consultant, to furnish design services required under the Contract Documents.

1.2.12 *Design Submission* means any and all documents, shop drawings, electronic information, including computer programs and computer generated materials, data, plans, drawings, sketches, illustrations, specifications, descriptions, models, and other information developed, prepared, furnished, delivered or required to be delivered by, or for, Design-Builder.

1.2.13 *Engineer, City Engineer* shall mean and refer to the City Engineer and any duly authorized representative.

1.2.14 *Final Completion* is the date on which all Work is complete in accordance with the Contract Documents, including but not limited to, any items identified in the punch list prepared under Section 6.6.1 and the submission of all documents set forth in Section 6.7.2.

1.2.15 *Force Majeure Events* are those events that are beyond the control of both Design-Builder and Owner, including the events of war, floods, labor disputes, earthquakes, epidemics, adverse weather conditions not reasonably anticipated, and other acts of God.

1.2.16 General Conditions of Contract refer to this DBIA Document No. 535, Standard Form of General Conditions of Contract Between Owner and Design-Builder (2022 Edition).

1.2.17 *GMP Exhibit* means that exhibit attached to DBIA Document No. 530, Standard Form of Agreement Between Owner and Design-Builder – Cost Plus Fee With an Option for a Guaranteed Maximum Price, which exhibit will have been agreed upon by Owner and Design-Builder prior to the execution of the Agreement.

1.2.18 *GMP Proposal* or *Proposal* means that proposal developed by Design-Builder in accordance with Section 6.6 of DBIA Document No. 530, *Standard Form of Agreement Between Owner and Design-Builder, Cost Plus Fee With an Option for a Guaranteed Maximum* Price or with Section 2.3 of DBIA Document No. 544, *Progressive Design-Build Agreement,* or DBIA Document No. 545, *Progressive Design-Build Agreement for Water and Wastewater Projects.*

1.2.19 *Hazardous Conditions* are any materials, wastes, substances and chemicals deemed to be hazardous under applicable Legal Requirements, or the handling, storage, remediation, or disposal of which are regulated by applicable Legal Requirements.

1.2.20 Legal Requirements are all applicable federal, state and local laws, codes, ordinances, rules, regulations, orders and decrees of any government or quasi-government entity having jurisdiction over the Project or Site, the practices involved in the Project or Site, or any Work.

1.2.21 *Owner's Project Criteria* are developed by or for Owner to describe Owner's program requirements and objectives for the Project, including use, space, price, time, site and expandability requirements, as well as submittal requirements and other requirements governing Design-Builder's performance of the Work. Owner's Project Criteria may include conceptual documents, design criteria, design performance specifications, design specifications, and LEED® or other sustainable design criteria and other Project-specific technical materials and requirements.

1.2.22 *Purchasing Manager* shall mean and refer to the Purchasing Manager of the Buyer.

1.2.23 Site is the land or premises on which the Project is located.

1.2.24 State Standard Specifications - State of California, Department of Transportation, Standard Specifications, Latest Edition.

1.2.25 Subcontractor is any person or entity retained by Design-Builder as an independent contractor to perform a portion of the Work and shall include materialmen and suppliers.

1.2.26 *Sub-Subcontractor* is any person or entity retained by a Subcontractor as an independent contractor to perform any portion of a Subcontractor's Work and shall include materialmen and suppliers.

1.2.27 Substantial Completion or Substantially Complete means the date on which the Work, or an agreed upon portion of the Work, is sufficiently complete in accordance with the Contract Documents so that Owner can occupy and use the Project or a portion thereof for its intended purposes.

1.2.28 *Work* is comprised of all Design-Builder's design, construction and other services required by the Contract Documents, including procuring and furnishing all materials, equipment, services and labor reasonably inferable from the Contract Documents.

Article 2

Design-Builder's Services and Responsibilities

2.1 General Services.

2.1.1 Design-Builder's Representative shall be reasonably available to Owner and shall have the necessary expertise and experience required to supervise the Work. Design-Builder's Representative shall communicate regularly with Owner and shall be vested with the authority to act on behalf of Design-Builder. Design-Builder's Representative may be replaced only with the mutual agreement of Owner and Design-Builder.

2.1.2 Unless the parties agree on a different time period for submission of a status report, Design-Builder shall provide Owner with a monthly status report detailing the progress of the Work, including (i) whether the Work is proceeding according to schedule; (ii) whether discrepancies, conflicts, or ambiguities exist in the Contract Documents that require resolution; (iii) whether health and safety issues exist in connection with the Work; (iv) status of the contingency account to the extent provided for in the *Standard Form of Agreement Between Owner and Design-Builder – Cost Plus Fee With an Option for a Guaranteed Maximum Price*; and (v) other items that require resolution so as not to jeopardize Design-Builder's ability to complete the Work for the Contract Price and within the Contract Time(s). Status reports shall be submitted with Design-Builder's draft Payment Applications as a pre-requisite to payment.

2.1.3 Unless a schedule for the execution of the Work has been attached to the Agreement as an exhibit at the time the Agreement is executed, Design-Builder shall prepare and submit, at least three (3) days prior to the meeting contemplated by Section 2.1.4 hereof, a schedule for the execution of the Work for Owner's review and response. The schedule shall indicate the dates for the start and completion of the various stages of Work, including the dates when Owner information and approvals are required to enable Design-Builder to achieve the Contract Time(s). The schedule shall be revised as required by conditions and progress of the Work, but such revisions shall not

relieve Design-Builder of its obligations to complete the Work within the Contract Time(s), as such dates may be adjusted in accordance with the Contract Documents. Owner's review of, and response to, the schedule shall not be construed as relieving Design-Builder of its complete and exclusive control over the means, methods, sequences and techniques for executing the Work.

2.1.4 The parties will meet within seven (7) days after execution of the Agreement to discuss issues affecting the administration of the Work and to implement the necessary procedures, including those relating to submittals and payment, to facilitate the ability of the parties to perform their obligations under the Contract Documents.

2.2 Design Professional Services.

2.2.1 Design-Builder shall, consistent with applicable state licensing laws, provide through qualified, licensed design professionals employed by Design-Builder, or procured from qualified, independent licensed Design Consultants, the necessary design services, including architectural, engineering and other design professional services, for the preparation of the required drawings, specifications and other design submittals to permit Design-Builder to complete the Work consistent with the Contract Documents. Nothing in the Contract Documents is intended or deemed to create any legal or contractual relationship between Owner and any Design Consultant.

Design-Builder shall employ only Design Consultants and/or Design Subconsultants who 2.2.2 are duly licensed and qualified to perform the Work consistent with the Contract Documents. Prior to the date that Design Consultants and/or Design Subconsultants perform Work on the Project, Design-Builder shall identify in writing to Owner all Design Consultants and Design Subconsultants. To the extent that Design-Builder has not selected a Design Consultant or Design Subconsultant prior to performing the Work, Design-Builder shall provide Owner in writing a list of any subsequently added Design Consultants and/or Design Subconsultants and their scope of Work prior to their performing Work on the Project. Owner may reasonably object to Design-Builder's selection of any Design Consultant or Design Subconsultant, provided that the Contract Price and/or Contract Time(s) shall be adjusted to the extent that Owner's decision impacts Design-Builder's cost and/or time of performance. Design-Builder shall not substitute a listed Design Consultant or Subconsultant without obtaining Owner's prior written consent; such consent shall not be unreasonably withheld. Nothing in the Contract Documents is intended or deemed to create any legal or contractual relationship between Owner and any Design Consultant or Design Subconsultant, including but not limited to any third-party beneficiary rights.

2.3 Standard of Care for Design Professional Services.

2.3.1 The standard of care for all design professional services performed to execute the Work shall be the care and skill ordinarily used by members of the applicable profession practicing under similar conditions at the same time and locality of the Project.

2.4 Design Development Services.

2.4.1 Design-Builder and Owner shall, consistent with any applicable provision of the Contract Documents, agree upon any interim Design Submissions that Owner may wish to review, which interim Design Submissions may include design criteria, drawings, diagrams and specifications setting forth the Project requirements.

2.4.1.1 Design Submissions shall be consistent with the Owner's Project Criteria as well as the Basis of Design Documents, as the Basis of Design Documents may have been changed or supplemented through the design process set forth in this Section 2.4.1. By submitting Design Submissions, Design-Builder represents to the Owner that the Work depicted and otherwise shown, contained, or reflected in Design Submissions may be constructed in compliance with the then current Contract Price and Contract Time. Notwithstanding the above, Design-Builder may propose Design Submissions that may alter the Basis of Design Documents, the Contract Price and/or Contract Time; however,

Design-Builder must provide notice thereof in accordance with Article 10 of the General Conditions and obtain a Change Order before such proposed Design Submissions are incorporated into the Construction Documents.

2.4.1.2 On or about the time of the Design Submissions, Design-Builder and Owner shall meet and confer about the Design Submissions, with Design-Builder identifying during such meetings, among other things, the evolution of the design and any changes to the Basis of Design Documents, or, if applicable, previously submitted Design Submissions. Changes to the Basis of Design Documents, including those that are deemed minor changes under Section 9.3.1, shall be processed in accordance with Article 9. Minutes of the meetings, including a full listing of all changes, will be maintained by Design-Builder and provided to all attendees for review. Following the design review meeting, Owner shall review and approve the interim Design Submissions and meeting minutes in a time that is consistent with the turnaround times set forth in Design-Builder's schedule.

2.4.1.3 Owner shall review and respond to Design Submissions, providing any comments and/or concerns about the Design Submissions. Owner shall provide all comments on the Design Submissions within the time provided by the Contract Documents. Design-Builder shall revise the Design Submissions (and any other deliverables) in response to Owner's comments and incorporate said responses into the next submission of Design Submissions.

2.4.1.4 If incorporation of Owner's comments results in a design that is inconsistent with or otherwise gives rise to a change in Owner's Project Criteria, the Basis of Design Documents, the Contract Price and/or the Contract Time, Design-Builder shall provide notice thereof in accordance with Articles 9 and 10 of the General Conditions. Changes to the Basis of Design Documents, the Contract Price and/or the Contract Time, including those that are deemed minor changes, shall be processed in accordance with Article 9 of the General Conditions.

2.4.2 Design-Builder shall submit to Owner Construction Documents setting forth in detail drawings and specifications describing the requirements for construction of the Work. The Construction Documents shall be consistent with the latest set of interim Design Submissions, as such submissions may have been modified in a design review meeting and recorded in the meeting minutes. The parties shall have a design review meeting to discuss, and Owner shall review and approve, the Construction Documents in accordance with the procedures set forth in Section 2.4.1 above. Design-Builder shall proceed with construction in accordance with the approved Construction Documents and shall submit one set of approved Construction Documents to Owner prior to commencement of construction.

2.4.3 Owner's review and approval of interim Design Submissions, meeting minutes, and the Construction Documents is for the purpose of mutually establishing a conformed set of Contract Documents compatible with the requirements of the Work. Neither Owner's review nor approval of any interim Design Submissions, meeting minutes, and Construction Documents shall be deemed to transfer any design liability from Design-Builder to Owner. Design-Builder shall provide Owner with sufficient time in the Project Schedule to review and approve the Design Submissions.

2.4.4 To the extent not prohibited by the Contract Documents or Legal Requirements, Design-Builder may prepare interim Design Submissions and Construction Documents for a portion of the Work to permit construction to proceed on that portion of the Work prior to completion of the Construction Documents for the entire Work.

2.5 Legal Requirements.

2.5.1 Design-Builder shall perform the Work in accordance with all Legal Requirements and shall provide all notices applicable to the Work as required by the Legal Requirements.

2.5.2 The Contract Price and/or Contract Time(s) shall be adjusted to compensate Design-Builder for the effects of any changes in the Legal Requirements enacted after the date of the Agreement affecting the performance of the Work, or if a Guaranteed Maximum Price is established after the date of the Agreement, the date the parties agree upon the Guaranteed Maximum Price. Such effects may include, without limitation, revisions Design-Builder is required to make to the Construction Documents because of changes in Legal Requirements.

2.6 Government Approvals and Permits.

2.6.1 Except as identified in an Owner's Permit List attached as an exhibit to the Agreement, Design-Builder shall obtain and pay for all necessary permits, approvals, licenses, government charges and inspection fees required for the prosecution of the Work by any government or quasi-government entity having jurisdiction over the Project.

2.6.2 Design-Builder shall provide reasonable assistance to Owner in obtaining those permits, approvals and licenses that are Owner's responsibility.

2.7 Design-Builder's Construction Phase Services.

2.7.1 Unless otherwise provided in the Contract Documents to be the responsibility of Owner or a separate contractor, Design-Builder shall provide through itself or Subcontractors the necessary supervision, labor, inspection, testing, start-up, material, equipment, machinery, temporary utilities and other temporary facilities to permit Design-Builder to complete construction of the Project consistent with the Contract Documents.

2.7.2 Design-Builder shall perform all construction activities efficiently and with the requisite expertise, skill and competence to satisfy the requirements of the Contract Documents. Design-Builder shall at all times exercise complete and exclusive control over the means, methods, sequences and techniques of construction.

2.7.3 Design-Builder shall employ only Subcontractors who are duly licensed and qualified to perform the Work consistent with the Contract Documents. Prior to the date that Subcontractors perform Work on the Project, Design-Builder shall identify in writing to Owner all Subcontractors. To the extent that Design-Builder has not selected a Subcontractor prior to performing the Work, Design-Builder shall provide Owner in writing a list of any subsequently added Subcontractors prior to their performing Work on the Project. Owner may reasonably object to Design-Builder's selection of any Subcontractor, provided that the Contract Price and/or Contract Time(s) shall be adjusted to the extent that Owner's decision impacts Design-Builder's cost and/or time of performance. Design-Builder may not substitute listed Subcontractors without Owner's prior written consent; such consent shall not be unreasonably withheld.

2.7.4 Design-Builder assumes responsibility to Owner for the proper performance of the Work of Subcontractors and any acts and omissions in connection with such performance. Nothing in the Contract Documents is intended or deemed to create any legal or contractual relationship between Owner and any Subcontractor or Sub-Subcontractor, including but not limited to any third-party beneficiary rights.

2.7.5 Design-Builder shall coordinate the activities of all Subcontractors. If Owner performs other work on the Project or at the Site with separate contractors under Owner's control, Design-Builder agrees to reasonably cooperate and coordinate its activities with those of such separate contractors so that the Project can be completed in an orderly and coordinated manner without unreasonable disruption.

2.7.6 Design-Builder shall keep the Site reasonably free from debris, trash, graffiti and vandalism, and construction wastes to permit Design-Builder to perform its construction services efficiently, safely and without interfering with the use of adjacent land areas. Upon Substantial Completion of the Work, or a portion of the Work, Design-Builder shall remove all debris, trash,

graffiti, U.S.A. markings, construction wastes, materials, equipment, machinery and tools arising from the Work or applicable portions thereof to permit Owner to occupy the Project or a portion of the Project for its intended use.

2.8 Design-Builder's Responsibility for Project Safety.

2.8.1 Design-Builder recognizes the importance of performing the Work in a safe manner so as to prevent damage, injury or loss to (i) all individuals at the Site, whether working or visiting; (ii) the Work, including materials and equipment incorporated into the Work or stored on-Site or off-Site; and (iii) all other property at the Site or adjacent thereto. Design-Builder assumes responsibility for implementing and monitoring all safety precautions and programs related to the performance of the Work. Design-Builder shall, prior to commencing construction, designate a Safety Representative with the necessary qualifications and experience to supervise the implementation and monitoring of all safety precautions and programs related to the Work. Unless otherwise required by the Contract Documents, Design-Builder's Safety Representative shall be an individual stationed at the Site who may have responsibilities on the Project in addition to safety. The Safety Representative shall make routine daily inspections of the Site and shall hold weekly safety meetings with Design-Builder's personnel, Subcontractors and others as applicable.

2.8.2 Design-Builder and Subcontractors shall comply with all Legal Requirements relating to safety, as well as any Owner-specific safety requirements set forth in the Contract Documents, provided that such Owner-specific requirements do not violate any applicable Legal Requirement. Design-Builder will immediately report in writing any safety-related injury, loss, damage or accident arising from the Work to Owner's Representative and, to the extent mandated by Legal Requirements, to all government or quasi-government authorities having jurisdiction over safety-related matters involving the Project or the Work.

2.8.3 Design-Builder's responsibility for safety under this Section 2.8 is not intended in any way to relieve Subcontractors and Sub-Subcontractors of their own contractual and legal obligations and responsibility for (i) complying with all Legal Requirements, including those related to health and safety matters; and (ii) taking all necessary measures to implement and monitor all safety precautions and programs to guard against injuries, losses, damages or accidents resulting from their performance of the Work.

2.9 Design-Builder's Warranty.

2.9.1 Design-Builder warrants to Owner that the construction, including all materials and equipment furnished as part of the construction, shall be new unless otherwise specified in the Contract Documents, of good quality, in conformance with the Contract Documents and free of defects in materials and workmanship. Design-Builder's warranty obligation excludes defects caused by abuse, alterations, or failure to maintain the Work in a commercially reasonable manner. Nothing in this warranty is intended to limit any manufacturer's warranty which provides Owner with greater warranty rights than set forth in this Section 2.9 or the Contract Documents. Design-Builder will provide Owner with all manufacturers' warranties upon Substantial Completion.

2.10 Correction of Defective Work.

2.10.1 Design-Builder agrees to correct any Work that is found to not be in conformance with the Contract Documents, including that part of the Work subject to Section 2.9 hereof, within a period of one year from the date of Substantial Completion of the Work or any portion of the Work, or within such longer period to the extent required by any specific warranty included in the Contract Documents.

2.10.2 Design-Builder shall, within seven (7) days of receipt of written notice from Owner that the Work is not in conformance with the Contract Documents, take meaningful steps to commence correction of such nonconforming Work, including the correction, removal or replacement of the nonconforming Work and any damage caused to other parts of the Work affected by the

nonconforming Work. If Design-Builder fails to commence the necessary steps within such seven (7) day period, Owner, in addition to any other remedies provided under the Contract Documents, may provide Design-Builder with written notice that Owner will commence correction of such nonconforming Work with its own forces. If Owner does perform such corrective Work, Design-Builder shall be responsible for all reasonable costs incurred by Owner in performing such correction. If the nonconforming Work creates an emergency requiring an immediate response, the seven (7) day period identified herein shall be deemed inapplicable.

2.10.3 The one-year period referenced in Section 2.10.1 above applies only to Design-Builder's obligation to correct nonconforming Work and is not intended to constitute a period of limitations for any other rights or remedies Owner may have regarding Design-Builder's other obligations under the Contract Documents.

Article 3

Owner's Services and Responsibilities

3.1 Duty to Cooperate.

3.1.1 Owner shall, throughout the performance of the Work, cooperate with Design-Builder and perform its responsibilities, obligations and services in a timely manner to facilitate Design-Builder's timely and efficient performance of the Work and so as not to delay or interfere with Design-Builder's performance of its obligations under the Contract Documents.

3.1.2 Owner shall provide timely reviews and approvals of interim Design Submissions and Construction Documents consistent with the turnaround times set forth in Design-Builder's schedule.

3.1.3 Owner shall give Design-Builder timely notice of any Work that Owner notices to be defective or not in compliance with the Contract Documents.

3.2 Furnishing of Services and Information.

3.2.1 Unless expressly stated to the contrary in the Contract Documents, Owner shall provide, at its own cost and expense, for Design-Builder's information and use the following, all of which Design-Builder is entitled to rely upon in performing the Work:

3.2.1.1 Surveys describing the property, boundaries, topography and reference points for use during construction, including existing service and utility lines;

3.2.1.2 Geotechnical studies describing subsurface conditions, and other surveys describing other latent or concealed physical conditions at the Site;

3.2.1.3 Temporary and permanent easements, zoning and other requirements and encumbrances affecting land use, or necessary to permit the proper design and construction of the Project and enable Design-Builder to perform the Work;

3.2.1.4 A legal description of the Site;

3.2.1.5 To the extent available, record drawings of any existing structures at the Site; and

3.2.1.6 To the extent available, environmental studies, reports and impact statements describing the environmental conditions, including Hazardous Conditions, in existence at the Site.

3.2.2 Owner is responsible for securing and executing all necessary agreements with adjacent land or property owners that are necessary to enable Design-Builder to perform the Work. Owner is further responsible for all costs, including attorneys' fees, incurred in securing these necessary agreements.

3.3 Financial Information.

3.3.1 At Design-Builder's written request, Owner shall promptly furnish reasonable evidence satisfactory to Design-Builder that Owner has adequate funds available and committed to fulfill all of Owner's contractual obligations under the Contract Documents. If Owner fails to furnish such financial information in a timely manner, Design-Builder may stop Work under Section 11.3 hereof or exercise any other right permitted under the Contract Documents.

3.3.2 Design-Builder shall cooperate with the reasonable requirements of Owner's lenders or other financial sources. Notwithstanding the preceding sentence, after execution of the Agreement Design-Builder shall have no obligation to execute for Owner or Owner's lenders or other financial sources any documents or agreements that require Design-Builder to assume obligations or responsibilities greater than those existing obligations Design-Builder has under the Contract Documents.

3.4 Owner's Representative.

3.4.1 Owner's Representative shall be responsible for providing Owner-supplied information and approvals in a timely manner to permit Design-Builder to fulfill its obligations under the Contract Documents. Owner's Representative shall also provide Design-Builder with prompt notice if it observes any failure on the part of Design-Builder to fulfill its contractual obligations, including any errors, omissions or defects in the performance of the Work. Owner's Representative shall communicate regularly with Design-Builder and shall be vested with the authority to act on behalf of Owner.

3.5 Government Approvals and Permits.

3.5.1 Owner shall obtain and pay for all necessary permits, approvals, licenses, government charges and inspection fees set forth in Owner's Permit List attached as an exhibit to the Agreement.

3.5.2 Owner shall provide reasonable assistance to Design-Builder in obtaining those permits, approvals and licenses that are Design-Builder's responsibility.

3.6 Owner's Separate Contractors.

3.6.1 Owner is responsible for all work performed on the Project or at the Site by separate contractors under Owner's control. Owner shall contractually require its separate contractors to cooperate with and coordinate their activities so as not to interfere with Design-Builder in order to enable Design-Builder to timely complete the Work consistent with the Contract Documents.

Article 4

Hazardous Conditions and Differing Site Conditions

4.1 Hazardous Conditions.

4.1.1 Unless otherwise expressly provided in the Contract Documents to be part of the Work, Design-Builder is not responsible for any Hazardous Conditions encountered at the Site. Upon

encountering any Hazardous Conditions, Design-Builder will stop Work immediately in the affected area and duly notify Owner and, if required by Legal Requirements, all government or quasi-government entities with jurisdiction over the Project or Site.

4.1.2 Upon receiving notice of the presence of suspected Hazardous Conditions, Owner shall take the necessary measures required to ensure that the Hazardous Conditions are remediated or rendered harmless.

4.1.3 Design-Builder shall be obligated to resume Work at the affected area of the Project only after Owner's expert provides it with written certification that (i) the Hazardous Conditions have been removed or rendered harmless; and (ii) all necessary approvals have been obtained from all government and quasi-government entities having jurisdiction over the Project or Site.

4.1.4 Design-Builder will be entitled, in accordance with these General Conditions of Contract, to an adjustment in its Contract Price and/or Contract Time(s) to the extent Design-Builder's cost and/or time of performance have been adversely impacted by the presence of Hazardous Conditions.

4.1.5 Notwithstanding the preceding provisions of this Section 4.1, Owner is not responsible for Hazardous Conditions introduced to the Site by Design-Builder, Subcontractors or anyone for whose acts they may be liable. To the fullest extent permitted by law, Design-Builder shall indemnify, defend and hold harmless Owner and Owner's officers, directors, employees and agents from and against all claims, losses, damages, liabilities and expenses, including attorneys' fees expenses, and costs to enforce this agreement, arising out of or resulting from those Hazardous Conditions introduced to the Site by Design-Builder, Subcontractors or anyone for whose acts they may be liable.

4.2 Differing Site Conditions.

4.2.1 Concealed or latent physical conditions or subsurface conditions at the Site that (i) materially differ from the conditions indicated in the Contract Documents or (ii) are of an unusual nature, differing materially from the conditions ordinarily encountered and generally recognized as inherent in the Work, are collectively referred to herein as "Differing Site Conditions." If Design-Builder encounters a Differing Site Condition, Design-Builder will be entitled to an adjustment in the Contract Price and/or Contract Time(s) to the extent Design-Builder's cost and/or time of performance are adversely impacted by the Differing Site Condition.

4.2.2 Upon encountering a Differing Site Condition, Design-Builder shall provide prompt written notice to Owner of such condition, which notice shall not be later than fourteen (14) days after such condition has been encountered. Design-Builder shall, to the extent reasonably possible, provide such notice before the Differing Site Condition has been substantially disturbed or altered.

Article 5

Insurance and Bonds

5.1 Design-Builder's Insurance Requirements.

5.1.1 MINIMUM LIMITS DETAILED BELOW APPLY DURING DESIGN PHASE (PHASE 1) ONLY. THE HIGHER LIMITS DETAILED IN 5.1.2 MUST BE IN FORCE PRIOR TO ANY CONSTRUCTION (PHASE 2).

(a) Throughout the life of this Agreement, DESIGN-BUILDER shall pay for and maintain in full force and effect all insurance as required herein with an insurance company(ies) either (i) admitted by the California Insurance Commissioner to do business in the State of California and rated no less than "A-

VII" in the Best's Insurance Rating Guide, or (ii) as may be authorized in writing by OWNER'S Risk Manager or his/her designee at any time and in his/her sole discretion. The required policies of insurance as stated herein shall maintain limits of liability of not less than those amounts stated therein. However, the insurance limits available to OWNER, its officers, officials, employees, agents and volunteers as additional insureds, shall be the greater of the minimum limits specified therein or the full limit of any insurance proceeds to the named insured.

(b) If at any time during the life of the Agreement or any extension, DESIGN-BUILDER or any of its subcontractors fail to maintain any required insurance in full force and effect, all services and work under this Agreement shall be discontinued immediately, and all payments due or that become due to DESIGN-BUILDER shall be withheld until notice is received by OWNER that the required insurance has been restored to full force and effect and that the premiums therefore have been paid for a period satisfactory to OWNER. Any failure to maintain the required insurance shall be sufficient cause for OWNER to terminate this Agreement. No action taken by OWNER pursuant to this section shall in any way relieve DESIGN-BUILDER of its responsibilities under this Agreement. The phrase "fail to maintain any required insurance" shall include, without limitation, notification received by OWNER that an insurer has commenced proceedings, or has had proceedings commenced against it, indicating that the insurer is insolvent.

(c) The fact that insurance is obtained by DESIGN-BUILDER shall not be deemed to release or diminish the liability of DESIGN-BUILDER, including, without limitation, liability under the indemnity provisions of this Agreement. The duty to indemnify OWNER shall apply to all claims and liability regardless of whether any insurance policies are applicable. The policy limits do not act as a limitation upon the amount of indemnification to be provided by DESIGN-BUILDER. Approval or purchase of any insurance contracts or policies shall in no way relieve from liability nor limit the liability of DESIGN-BUILDER, vendors, suppliers, invitees, contractors, sub-contractors, subcontractors, or anyone employed directly or indirectly by any of them.

Coverage shall be at least as broad as:

1. The most current version of Insurance Services Office (ISO) Commercial General Liability Coverage Form CG 00 01, providing liability coverage arising out of your business operations. The Commercial General Liability policy shall be written on an occurrence form and shall provide coverage for "bodily injury," "property damage" and "personal and advertising injury" with coverage for premises and operations (including the use of owned and non-owned equipment), products and completed operations, and contractual liability (including, without limitation, indemnity obligations under the Agreement) with limits of liability not less than those set forth under "Minimum Limits of Insurance."

2. The most current version of ISO *Commercial Auto Coverage Form CA 00 01, providing liability coverage arising out of the ownership, maintenance or use of automobiles in the course of your business operations. The Automobile Policy shall be written on an occurrence form and shall provide coverage for all owned, hired, and non-owned automobiles or other licensed vehicles (Code 1- Any Auto).

3. Workers' Compensation insurance as required by the State of California and Employer's Liability Insurance.

4. Professional Liability (Errors and Omissions) insurance appropriate to DESIGN-BUILDER's profession.

MINIMUM LIMITS OF INSURANCE DESIGN/BUILD DESIGN-BUILDER

DESIGN-BUILDER shall procure and maintain for the duration of the contract, and for 5 years thereafter, insurance with limits of liability not less than those set forth below. However, insurance limits available to OWNER, its officers, officials, employees, agents and volunteers as additional insureds, shall be the greater of the minimum limits specified herein or the full limit of any insurance proceeds available to the named insured:

1. COMMERCIAL GENERAL LIABILITY

- (i) \$1,000,000 per occurrence for bodily injury and property damage;
- (ii) \$1,000,000 per occurrence for personal and advertising injury;
- (iii) \$2,000,000 aggregate for products and completed operations; and,
- (iv) \$2,000,000 general aggregate applying separately to the work performed under the Agreement.

2. COMMERCIAL AUTOMOBILE LIABILITY

\$1,000,000 per accident for bodily injury and property damage.

3. Workers' Compensation Insurance as required by the State of California with statutory limits and EMPLOYER'S LIABILITY with limits of liability not less than:

- (i) \$1,000,000 each accident for bodily injury;
- (ii) \$1,000,000 disease each employee; and,
- (iii) \$1,000,000 disease policy limit.
- 4. PROFESSIONAL LIABILITY (Errors and Omissions):
 - (i) \$2,000,000 per claim/occurrence; and,
 - (ii) \$4,000,000 policy aggregate.

UMBRELLA OR EXCESS INSURANCE

In the event DESIGN-BUILDER purchases an Umbrella or Excess insurance policy(ies) to meet the "Minimum Limits of Insurance," this insurance policy(ies) shall "follow form" and afford no less coverage than the primary insurance policy(ies). In addition, such Umbrella or Excess insurance policy(ies) shall also apply on a primary and non-contributory basis for the benefit of the OWNER, its officers, officials, employees, agents and volunteers.

DEDUCTIBLES AND SELF-INSURED RETENTIONS

DESIGN-BUILDER shall be responsible for payment of any deductibles contained in any insurance policy(ies) required herein and DESIGN-BUILDER shall also be responsible for payment of any self-insured retentions. Any deductibles or self-insured retentions must be declared on the Certificate of Insurance, and approved by, the OWNER'S Risk Manager or his/her designee. At the option of the OWNER'S Risk Manager or his/her designee, either:

(i) The insurer shall reduce or eliminate such deductibles or self-insured retentions as respects OWNER, its officers, officials, employees, agents and volunteers; or

(ii) DESIGN-BUILDER shall provide a financial guarantee, satisfactory to OWNER'S Risk Manager or his/her designee, guaranteeing payment of losses and related investigations, claim administration and defense expenses. At no time shall OWNER be responsible for the payment of any deductibles or self-insured retentions.

OTHER INSURANCE PROVISIONS/ENDORSEMENTS

(i) All policies of insurance required herein shall be endorsed to provide that the coverage shall not be cancelled, non-renewed, reduced in coverage or in limits except after thirty (30) calendar days written notice has been given to OWNER, except ten (10) days for nonpayment of premium. DESIGN-BUILDER is also responsible for providing written notice to the OWNER under the same terms and conditions. Upon issuance by the insurer, broker, or agent of a notice of cancellation, non-renewal, or reduction in coverage or in limits, DESIGN-BUILDER shall furnish OWNER with a new certificate and applicable endorsements for such policy(ies). In the event any policy is due to expire during the work to be performed for OWNER, DESIGN-BUILDER shall provide a new certificate, and applicable endorsements, evidencing renewal of such policy not less than fifteen (15) calendar days prior to the expiration date of the expiring policy.

(ii) The Commercial General and Automobile, Liability insurance policies shall be written on an occurrence form.

(iii) The Commercial General and Automobile Liability insurance policies shall be endorsed to name City of Fresno, its officers, officials, agents, employees and volunteers as an additional insured. DESIGN-BUILDER shall establish additional insured status for the OWNER for all ongoing and completed operations under Commercial General Liability policies by use of ISO Forms or an executed manuscript insurance company endorsement providing additional insured status. The Commercial General endorsements must be as broad as that contained in ISO Forms: GC 20 10 11 85 or both CG 20 10 & CG 20 37.

(iv) All such policies of insurance shall be endorsed so the DESIGN-BUILDERS' insurance shall be primary and no contribution shall be required of OWNER. The coverage shall contain no special limitations on the scope of protection afforded to Owner, its officers, officials, employees, agents and volunteers. If DESIGN-BUILDER maintains higher limits of liability than the minimums shown above, OWNER requires and shall be entitled to coverage for the higher limits of liability maintained by DESIGN-BUILDER.

(v) Should any of these policies provide that the defense costs are paid within the Limits of Liability, thereby reducing the available limits by defense costs, then the requirement for the Limits of Liability of these polices will be twice the above stated limits.

(vi) For any claims related to this Agreement, DESIGN-BUILDER'S insurance coverage shall be primary insurance with respect to the City of Fresno, its officers, officials, agents, employees and volunteers. Any insurance or self-insurance maintained by the OWNER, its officers, officials, agents, employees and volunteers shall be excess of the DESIGN-BUILDER'S insurance and shall not contribute with it.

(vii) The Workers' Compensation insurance policy shall contain, or be endorsed to contain, a waiver of subrogation as to City of Fresno, its officers, officials, agents, employees and volunteers.

If the Professional (Errors and Omissions) policy is written on a claims-made form:

1. The retroactive date must be shown, and must be before the effective date of the Agreement or the commencement of work by DESIGN-BUILDER.

2. Insurance must be maintained and evidence of insurance must be provided for at least five (5) years after completion of the Agreement work or termination of the Agreement, whichever occurs first, or, in the alternative, the policy shall be endorsed to provide not less than a five (5) year discovery period.

3. If coverage is canceled or non-renewed, and not replaced with another claims-made policy form with a retroactive date prior to the effective date of the Agreement or the commencement of work by DESIGN-BUILDER, DESIGN-BUILDER must purchase "extended reporting" coverage for a minimum of five (5) years completion of the Agreement work or termination of the Agreement, whichever occurs first.

- 4. A copy of the claims reporting requirements must be submitted to OWNER for review.
- 5. These requirements shall survive expiration or termination of the Agreement.

PROVIDING OF DOCUMENTS - DESIGN-BUILDER shall furnish OWNER with all certificate(s) and applicable endorsements effecting coverage required herein All certificates and applicable endorsements are to be received and approved by the OWNER'S Risk Manager or his/her designee prior to OWNER'S execution of the Agreement and before work commences. All non-ISO endorsements amending policy coverage shall be executed by a licensed and authorized agent or broker. Upon request of OWNER, DESIGN-BUILDER shall immediately furnish OWNER with a complete copy of any insurance policy required under this Agreement, including all endorsements, with said copy certified by the underwriter to be a true and correct copy of the original policy. This requirement shall survive expiration or termination of this Agreement. All subcontractors working under the direction of DESIGN-BUILDER shall also be required to provide all documents noted herein.

SUBCONTRACTORS -If DESIGN-BUILDER subcontracts any or all of the services to be performed under this Agreement, DESIGN-BUILDER shall require, at the discretion of the OWNER Risk Manager or designee, subcontractor (s) to enter into a separate Side Agreement with the Owner to provide required indemnification and insurance protection. Any required Side Agreement(s) and associated insurance documents for the subcontractors must be reviewed and preapproved by OWNER Risk Manager or designee. If no Side Agreement is required, DESIGN-BUILDER shall require and verify that subcontractors maintain insurance meeting all the requirements stated herein and DESIGN-BUILDER shall ensure that OWNER, its officers, officials, employees, agents and volunteers are additional insureds. The subcontractors' certificates and endorsements shall be on file with DESIGN-BUILDER, and OWNER, prior to commencement of any work by the subcontractors.

5.1.2 PHASE 2 BUILD Design-Builder's insurance shall specifically delete any design-build or similar exclusions that could compromise coverages because of the design-build delivery of the Project.

(a) Throughout the life of this Agreement, DESIGN-BUILDER shall pay for and maintain in full force and effect all insurance as required herein with an insurance company(ies) either (i) admitted by the California Insurance Commissioner to do business in the State of California and rated no less than "A-VII" in the Best's Insurance Rating Guide, or (ii) as may be authorized in writing by OWNER'S Risk Manager or his/her designee at any time and in his/her sole discretion. The required policies of insurance as stated herein shall maintain limits of liability of not less than those
amounts stated therein. However, the insurance limits available to OWNER, its officers, officials, employees, agents and volunteers as additional insureds, shall be the greater of the minimum limits specified therein or the full limit of any insurance proceeds to the named insured.

(b) If at any time during the life of the Agreement or any extension, DESIGN-BUILDER or any of its subcontractors fail to maintain any required insurance in full force and effect, all services and work under this Agreement shall be discontinued immediately, and all payments due or that become due to DESIGN-BUILDER shall be withheld until notice is received by OWNER that the required insurance has been restored to full force and effect and that the premiums therefore have been paid for a period satisfactory to OWNER. Any failure to maintain the required insurance shall be sufficient cause for OWNER to terminate this Agreement. No action taken by OWNER pursuant to this section shall in any way relieve DESIGN-BUILDER of its responsibilities under this Agreement. The phrase "fail to maintain any required insurance" shall include, without limitation, notification received by OWNER that an insurer has commenced proceedings, or has had proceedings commenced against it, indicating that the insurer is insolvent.

(c) The fact that insurance is obtained by DESIGN-BUILDER shall not be deemed to release or diminish the liability of DESIGN-BUILDER, including, without limitation, liability under the indemnity provisions of this Agreement. The duty to indemnify OWNER shall apply to all claims and liability regardless of whether any insurance policies are applicable. The policy limits do not act as a limitation upon the amount of indemnification to be provided by DESIGN-BUILDER. Approval or purchase of any insurance contracts or policies shall in no way relieve from liability nor limit the liability of DESIGN-BUILDER, vendors, suppliers, invitees, contractors, sub-contractors, subcontractors, or anyone employed directly or indirectly by any of them.

Coverage shall be at least as broad as:

1. The most current version of Insurance Services Office (ISO) Commercial General Liability Coverage Form CG 00 01, providing liability coverage arising out of your business operations. The Commercial General Liability policy shall be written on an occurrence form and shall provide coverage for "bodily injury," "property damage" and "personal and advertising injury" with coverage for premises and operations (including the use of owned and non-owned equipment), products and completed operations, and contractual liability (including, without limitation, indemnity obligations under the Agreement) with limits of liability not less than those set forth under "Minimum Limits of Insurance."

2. The most current version of ISO *Commercial Auto Coverage Form CA 00 01, providing liability coverage arising out of the ownership, maintenance or use of automobiles in the course of your business operations. The Automobile Policy shall be written on an occurrence form and shall provide coverage for all owned, hired, and non-owned automobiles or other licensed vehicles (Code 1- Any Auto).

3. Workers' Compensation insurance as required by the State of California and Employer's Liability Insurance.

4. Professional Liability (Errors and Omissions) insurance appropriate to DESIGN-BUILDER's profession.

MINIMUM LIMITS OF INSURANCE DESIGN/BUILD DESIGN-BUILDER

DESIGN-BUILDER shall procure and maintain for the duration of the contract, and for 5 years thereafter, insurance with limits of liability not less than those set forth below. However, insurance limits available to OWNER, its officers, officials, employees, agents and volunteers as additional insureds, shall be the greater of the minimum limits specified herein or the full limit of any insurance proceeds available to the named insured:

1. COMMERCIAL GENERAL LIABILITY

- (i) \$5,000,000 per occurrence for bodily injury and property damage;
- (ii) \$5,000,000 per occurrence for personal and advertising injury;
- (iii) \$10,000,000 aggregate for products and completed operations; and,

(iv) \$10,000,000 general aggregate applying separately to the work performed under the Agreement.

2. COMMERCIAL AUTOMOBILE LIABILITY

\$1,000,000 per accident for bodily injury and property damage.

3. Workers' Compensation Insurance as required by the State of California with statutory limits and EMPLOYER'S LIABILITY with limits of liability not less than:

- (i) \$1,000,000 each accident for bodily injury;
- (ii) \$1,000,000 disease each employee; and,
- (iii) \$1,000,000 disease policy limit.

4. PROFESSIONAL LIABILITY (Errors and Omissions):

- (i) \$2,000,000 per claim/occurrence; and,
- (ii) \$4,000,000 policy aggregate.

5. BUILDERS RISK (Course of Construction) insurance in an amount equal to the completed value of the project with no coinsurance penalty provisions.

6. CONTRACTOR'S POLLUTION LIABILITY with coverage for bodily injury, property damage or pollution clean-up costs that could result from of pollution condition, both sudden and gradual. Including a discharge of pollutants brought to the work site, a release of pre-existing pollutants at the site, or other pollution conditions with limits of liability of not less than the following:

- (i) \$2,000,000 per occurrence or claim; and,
- (ii) \$4,000,000 general aggregate per annual policy period.

(a) In the event this Agreement involves the transportation of hazardous material, either the Commercial Automobile policy or other appropriate insurance policy shall be endorsed to include Transportation Pollution Liability insurance covering materials to be transported by DESIGN-BUILDER pursuant to the Agreement.

UMBRELLA OR EXCESS INSURANCE

In the event DESIGN-BUILDER purchases an Umbrella or Excess insurance policy(ies) to meet the "Minimum Limits of Insurance," this insurance policy(ies) shall "follow form" and afford no less coverage than the primary insurance policy(ies). In addition, such Umbrella or Excess insurance policy(ies) shall also apply on a primary and non-contributory basis for the benefit of the OWNER, its officers, officials, employees, agents and volunteers.

DEDUCTIBLES AND SELF-INSURED RETENTIONS

DESIGN-BUILDER shall be responsible for payment of any deductibles contained in any insurance policy(ies) required herein and DESIGN-BUILDER shall also be responsible for payment of any self-insured retentions. Any deductibles or self-insured retentions must be declared on the Certificate of Insurance, and approved by, the OWNER'S Risk Manager or his/her designee. At the option of the OWNER'S Risk Manager or his/her designee, either:

(i) The insurer shall reduce or eliminate such deductibles or self-insured retentions as respects OWNER, its officients, officials, employees, agents and volunteers; or

(ii) DESIGN-BUILDER shall provide a financial guarantee, satisfactory to OWNER'S Risk Manager or his/her designee, guaranteeing payment of losses and related investigations, claim administration and defense expenses. At no time shall OWNER be responsible for the payment of any deductibles or self-insured retentions.

OTHER INSURANCE PROVISIONS/ENDORSEMENTS

(i) All policies of insurance required herein shall be endorsed to provide that the coverage shall not be cancelled, non-renewed, reduced in coverage or in limits except after thirty (30) calendar days written notice has been given to OWNER, except ten (10) days for nonpayment of premium. DESIGN-BUILDER is also responsible for providing written notice to the OWNER under the same terms and conditions. Upon issuance by the insurer, broker, or agent of a notice of cancellation, non-renewal, or reduction in coverage or in limits, DESIGN-BUILDER shall furnish OWNER with a new certificate and applicable endorsements for such policy(ies). In the event any policy is due to expire during the work to be performed for OWNER, DESIGN-BUILDER shall provide a new certificate, and applicable endorsements, evidencing renewal of such policy not less than fifteen (15) calendar days prior to the expiration date of the expiring policy.

(ii) The Commercial General, Automobile, and Pollution Liability insurance policies shall be written on an occurrence form.

(iii) The Commercial General, Automobile and Contractors Pollution Liability insurance policies shall be endorsed to name City of Fresno, its officers, officials, agents, employees and volunteers as an additional insured. DESIGN-BUILDER shall establish additional insured status for the OWNER for all ongoing and completed operations under both Commercial General and Commercial Pollution Liability policies by use of ISO Forms or an executed manuscript insurance company endorsement providing additional insured status. The Commercial General endorsements must be as broad as that contained in ISO Forms: GC 20 10 11 85 or both CG 20 10 & CG 20 37.

(iv) All such policies of insurance shall be endorsed so the DESIGN-BUILDERS' insurance shall be primary and no contribution shall be required of OWNER. The coverage shall contain no special limitations on the scope of protection afforded to Owner, its officers, officials, employees, agents and volunteers. If DESIGN-BUILDER maintains higher limits of liability than the minimums shown above, OWNER requires and shall be entitled to coverage for the higher limits of liability maintained by DESIGN-BUILDER.

(v) Should any of these policies provide that the defense costs are paid within the Limits of Liability, thereby reducing the available limits by defense costs, then the requirement for the Limits of Liability of these polices will be twice the above stated limits.

(vi) For any claims related to this Agreement, DESIGN-BUILDER'S insurance coverage shall be primary insurance with respect to the City of Fresno, its officers, officials, agents, employees and volunteers. Any insurance or self-insurance maintained by the OWNER, its officers, officials, agents, employees and volunteers shall be excess of the DESIGN-BUILDER'S insurance and shall not contribute with it.

(vii) The Workers' Compensation insurance policy shall contain, or be endorsed to contain, a waiver of subrogation as to City of Fresno, its officers, officials, agents, employees and volunteers.

(viii) The Builder's Risk Insurance shall have the policy endorsed to provide the City of Fresno to be named as a Loss Payee.

If the Professional (Errors and Omissions) policy is written on a claims-made form:

1. The retroactive date must be shown, and must be before the effective date of the Agreement or the commencement of work by DESIGN-BUILDER.

2. Insurance must be maintained and evidence of insurance must be provided for at least five (5) years after completion of the Agreement work or termination of the Agreement, whichever occurs first, or, in the alternative, the policy shall be endorsed to provide not less than a five (5) year discovery period.

3. If coverage is canceled or non-renewed, and not replaced with another claims-made policy form with a retroactive date prior to the effective date of the Agreement or the commencement of work by DESIGN-BUILDER, DESIGN-BUILDER must purchase "extended reporting" coverage for a minimum of five (5) years completion of the Agreement work or termination of the Agreement, whichever occurs first.

- 4. A copy of the claims reporting requirements must be submitted to OWNER for review.
- 5. These requirements shall survive expiration or termination of the Agreement.

PROVIDING OF DOCUMENTS - DESIGN-BUILDER shall furnish OWNER with all certificate(s) and applicable endorsements effecting coverage required herein All certificates and applicable endorsements are to be received and approved by the OWNER'S Risk Manager or his/her designee prior to OWNER'S execution of the Agreement and before work commences. All non-ISO endorsements amending policy coverage shall be executed by a licensed and authorized agent or broker. Upon request of OWNER, DESIGN-BUILDER shall immediately furnish OWNER with a complete copy of any insurance policy required under this Agreement, including all endorsements, with said copy certified by the underwriter to be a true and correct copy of the original policy. This requirement shall survive expiration or termination of this Agreement. All subcontractors working under the direction of DESIGN-BUILDER shall also be required to provide all documents noted herein.

SUBCONTRACTORS -If DESIGN-BUILDER subcontracts any or all of the services to be performed under this Agreement, DESIGN-BUILDER shall require, at the discretion of the OWNER Risk Manager or designee, subcontractor (s) to enter into a separate Side Agreement with the Owner to provide required indemnification and insurance protection. Any required Side Agreement(s) and associated insurance documents for the subcontractors must be reviewed and preapproved by OWNER Risk Manager or designee. If no Side Agreement is required, DESIGN-BUILDER shall require and verify that subcontractors maintain insurance meeting all the requirements stated herein and DESIGN-BUILDER shall ensure that OWNER, its officers, officials, employees, agents and volunteers are additional insureds. The subcontractors' certificates and endorsements shall be on file with DESIGN-BUILDER, and OWNER, prior to commencement of any work by the subcontractors.

5.4 Bonds and Other Performance Security.

5.4.1 If Owner requires Design-Builder to obtain performance and labor and material payment bonds, or other forms of performance security, the amount, form and other conditions of such security shall be as set forth in the Agreement.

5.4.2 All bonds furnished by Design-Builder shall be in a form satisfactory to Owner. The surety shall be a company

Article 6

Payment

6.1 Schedule of Values.

6.1.1 Unless required by Owner upon execution of this Agreement, within ten (10) days of execution of the Agreement, Design-Builder shall submit for Owner's review and approval a schedule of values for all of the Work. The Schedule of Values will (i) subdivide the Work into its respective parts; (ii) include values for all items comprising the Work; and (iii) serve as the basis for monthly progress payments made to Design-Builder throughout the Work.

6.1.2 Owner will timely review and approve the schedule of values so as not to delay the submission of Design-Builder's first application for payment. Owner and Design-Builder shall timely resolve any differences so as not to delay Design-Builder's submission of its first application for payment.

6.2 Monthly Progress Payments.

6.2.1 On or before the date established in the Agreement, Design-Builder shall submit for Owner's review and approval its Application for Payment requesting payment for all Work performed as of the date of the Application for Payment. The Application for Payment shall be accompanied by all supporting documentation required by the Contract Documents and/or established at the meeting required by Section 2.1.4 hereof.

6.2.2 The Application for Payment may request payment for equipment and materials not yet incorporated into the Project, provided that (i) Owner is satisfied that the equipment and materials are suitably stored at either the Site or another acceptable location; (ii) the equipment and materials are protected by suitable insurance; and (iii) upon payment, Owner will receive the equipment and materials free and clear of all liens and encumbrances.

6.2.3 All discounts offered by Subcontractors, Sub-Subcontractors, and suppliers to Design-Builder for early payment shall accrue one hundred percent to Design-Builder to the extent Design-Builder advances payment. Unless Owner advances payment to Design-Builder specifically to receive the discount, Design-Builder may include in its Application for Payment the full undiscounted cost of the item for which payment is sought.

6.2.4 The Application for Payment shall constitute Design-Builder's representation that the Work described therein has been performed consistent with the Contract Documents, has progressed to the point indicated in the Application for Payment, and that title to all Work will pass to Owner free and clear of all claims, liens, encumbrances, and security interests upon the incorporation of the Work into the Project, or upon Design-Builder's receipt of payment, whichever occurs earlier.

6.3 Withholding of Payments.

6.3.1 On or before the date established in the Agreement, Owner shall pay Design-Builder all amounts properly due. If Owner determines that Design-Builder is not entitled to all or part of an

Application for Payment as a result of Design-Builder's failure to meet its obligations hereunder, it will notify Design-Builder in writing at least five (5) days prior to the date payment is due. The notice shall indicate the specific amounts Owner intends to withhold, the reasons and contractual basis for the withholding, and the specific measures Design-Builder must take to rectify Owner's concerns. Design-Builder and Owner will attempt to resolve Owner's concerns prior to the date payment is due. If the parties cannot resolve such concerns, Design-Builder may pursue its rights under the Contract Documents, including those under Article 10 hereof.

6.3.2 Notwithstanding anything to the contrary in the Contract Documents, Owner shall pay Design-Builder all undisputed amounts in an Application for Payment within the times required by the Agreement.

6.4 Design-Builder's Payment Obligations.

6.4.1 Design-Builder will pay Design Consultants and Subcontractors, in accordance with its contractual obligations to such parties, all the amounts Design-Builder has received from Owner on account of their work. Design-Builder will impose similar requirements on Design Consultants and Subcontractors to pay those parties with whom they have contracted. Design-Builder will indemnify and defend Owner against any claims for payment and mechanic's liens as set forth in Section 7.3 hereof.

6.5 Substantial Completion.

6.5.1 Design-Builder shall notify Owner when it believes the Work, or to the extent permitted in the Contract Documents, a portion of the Work, is Substantially Complete. Within five (5) days of Owner's receipt of Design-Builder's notice, Owner and Design-Builder will jointly inspect such Work to verify that it is Substantially Complete in accordance with the requirements of the Contract Documents. If such Work is Substantially Complete, Owner shall prepare and issue a Certificate of Substantial Completion that will set forth (i) the date of Substantial Completion of the Work or portion thereof; (ii) the remaining items of Work that have to be completed before final payment; (iii) provisions (to the extent not already provided in the Contract Documents) establishing Owner's and Design-Builder's responsibility for the Project's security, maintenance, utilities and insurance pending final payment; and (iv) an acknowledgment that warranties commence to run on the date of Substantial Completion, except as may otherwise be noted in the Certificate of Substantial Completion.

6.5.2 Upon Substantial Completion of the entire Work or, if applicable, any portion of the Work, Owner shall release to Design-Builder all retained amounts relating, as applicable, to the entire Work or completed portion of the Work, less an amount equal to the reasonable value of all remaining or incomplete items of Work as noted in the Certificate of Substantial Completion.

6.5.3 Owner, at its option, may use a portion of the Work which has been determined to be Substantially Complete, provided, however, that (i) a Certificate of Substantial Completion has been issued for the portion of Work addressing the items set forth in Section 6.6.1 above; (ii) Design-Builder and Owner have obtained the consent of their sureties and insurers, and to the extent applicable, the appropriate government authorities having jurisdiction over the Project; and (iii) Owner and Design-Builder agree that Owner's use or occupancy will not interfere with Design-Builder's completion of the remaining Work.

6.5.4 On the date that the Owner issues the Certificate of Substantial Completion, the Owner shall provide Design-Builder with the final punch list identifying the remaining minor corrective items to be completed for final completion of the Project.

When the Design-Builder considers the final punch list work to be complete, it shall request Owner to perform a final walk through of the Project to determine if said punch list work is complete and whether Design-Builder has otherwise completed all of its obligations under the Contract Documents.

6.6 Completion.

The Owner shall record the Notice of Completion when the entire Work including, but not limited to Design-Builder's closeout document obligations are fully satisfied, Design-Builder's punch list(s) and work shall have been completed to the satisfaction of the Owner.

However, the Owner, at its sole option, may accept completion of the Contract and have the Notice of Completion recorded when the entire Work including individual portions of the Work shall have been completed to the satisfaction of the Owner, except for minor corrective items, as distinguished from incomplete items.

Regardless of the cause therefore, the Design-Builder may not maintain any claim or cause of action against the Owner for damages incurred as a result of its failure or inability to complete its Work on the Project in a shorter period than established in the Contract Documents, the parties stipulating that the period set forth in the Contract Documents is a reasonable time within which to perform the work on the Project.

6.7 Final Payment.

6.7.1 After receipt of a Final Application for Payment from Design-Builder, Owner shall make final payment by the time required in the Agreement, provided that Design-Builder has achieved Final Completion.

6.7.2 At the time of submission of its Final Application for Payment, Design-Builder shall provide the following information:

6.7.2.1 An affidavit that there are no claims, obligations or liens outstanding or unsatisfied for labor, services, material, equipment, taxes or other items performed, furnished or incurred for or in connection with the Work which will in any way affect Owner's interests;

6.7.2.2 A general release executed by Design-Builder waiving, upon receipt of final payment by Design-Builder, all claims, except those claims previously made in writing to Owner and remaining unsettled at the time of final payment;

6.7.2.3 Consent of Design-Builder's surety, if any, to final payment;

6.7.2.4 All operating manuals, warranties, As-Built drawings, marked up Record documents, and other deliverables required by the Contract Documents; and

6.7.2.5 Certificates of insurance confirming that required coverages will remain in effect consistent with the requirements of the Contract Documents.

6.7.3 Upon making final payment, Owner waives all claims against Design-Builder except claims relating to (i) Design-Builder's failure to satisfy its payment obligations, if such failure affects Owner's interests; (ii) Design-Builder's failure to complete the Work consistent with the Contract Documents, including defects appearing after Substantial Completion; and (iii) the terms of any special warranties required by the Contract Documents.

6.7.4 Deficiencies in the Work discovered after Substantial Completion, whether or not such deficiencies would have been included on the punch list if discovered earlier, shall be deemed warranty Work. Such deficiencies shall be corrected by Design-Builder under Sections 2.9 and 2.10 herein and shall not be a reason to withhold final payment from Design-Builder, provided, however, that Owner shall be entitled to withhold from the Final Payment the reasonable value of completion of such deficient work until such work is completed.

Article 7

Indemnification

7.1 Patent and Copyright Infringement.

7.1.1 Design-Builder shall defend any action or proceeding brought against Owner based on any claim that the Work, or any part thereof, or the operation or use of the Work or any part thereof, constitutes infringement of any United States patent or copyright, now or hereafter issued. Owner shall give prompt written notice to Design-Builder of any such action or proceeding and will reasonably provide authority, information and assistance in the defense of same. Design-Builder shall indemnify and hold harmless Owner from and against all damages and costs, including but not limited to attorneys' fees and expenses awarded against Owner or Design-Builder in any such action or proceeding. Design-Builder agrees to keep Owner informed of all developments in the defense of such actions.

7.1.2 If Owner is enjoined from the operation or use of the Work, or any part thereof, as the result of any patent or copyright suit, claim, or proceeding, Design-Builder shall at its sole expense take reasonable steps to procure the right to operate or use the Work. If Design-Builder cannot so procure such right within a reasonable time, Design-Builder shall promptly, at Design-Builder's option and at Design-Builder's expense, (i) modify the Work so as to avoid infringement of any such patent or copyright; or (ii) replace said Work with Work that does not infringe or violate any such patent or copyright.

7.1.3 Sections 7.1.1 and 7.1.2 above shall not be applicable to any suit, claim or proceeding based on infringement or violation of a patent or copyright (i) relating solely to a particular process or product of a particular manufacturer specified by Owner and not offered or recommended by Design-Builder to Owner; or (ii) arising from modifications to the Work by Owner or its agents after acceptance of the Work. If the suit, claim or proceeding is based upon events set forth in the preceding sentence, Owner shall defend, indemnify and hold harmless Design-Builder to the same extent Design-Builder is obligated to defend, indemnify and hold harmless Owner in Section 7.1.1 above.

7.1.4 The obligations set forth in this Section 7.1 shall constitute the sole agreement between the parties relating to liability for infringement of violation of any patent or copyright.

7.2 Payment Claim Indemnification.

7.2.1 Provided that Owner is not in breach of its contractual obligation to make payments to Design-Builder for the Work, Design-Builder shall indemnify, defend and hold harmless Owner from any claims or mechanic's liens brought against Owner or against the Project as a result of the failure of Design-Builder, or those for whose acts it is responsible, to pay for any services, materials, labor, equipment, taxes or other items or obligations furnished or incurred for or in connection with the Work. Within three (3) days of receiving written notice from Owner that such a claim or mechanic's lien has been filed, Design-Builder shall commence to take the steps necessary to discharge said claim or lien, including, if necessary, the furnishing of a mechanic's lien bond. If Design-Builder fails to do so, Owner will have the right to discharge the claim or lien and hold Design-Builder liable for costs and expenses incurred, including attorneys' fees.

7.3 Design-Builder's General Indemnification.

7.3.1 To the furthest extent allowed by law, including California Civil Code section 2782, DESIGN-BUILDER shall indemnify, defend and hold harmless OWNER and each of its officiers, officials, employees, agents, and volunteers from any and all claims, demands, actions in law or equity, loss, liability, fines, penalties, forfeitures, interest, costs including legal fees, and damages (whether in contract, tort, or strict liability, including but not limited to personal injury, death at any time,

property damage, or loss of any type) arising or alleged to have arisen directly or indirectly out of (1) any voluntary or involuntary act or omission, (2) error, omission or negligence, or (3) the performance or non-performance of this Contract. DESIGN-BUILDER'S obligations as set forth in this section shall apply regardless of whether OWNER or any of its officers, officials, employees, agents, or volunteers are passively negligent, but shall not apply to any loss, liability, fines, penalties, forfeitures, costs or damages caused by the active or sole negligence, or the willful misconduct, of OWNER or any of its officers, officials, employees, agents or volunteers.

To the fullest extent allowed by law, and in addition to the express duty to indemnify, DESIGN-BUILDER, whenever there is any causal connection between the DESIGN-BUILDER's performance or non-performance of the work or services required under this Contract and any claim or loss, injury or damage of any type, DESIGN-BUILDER expressly agrees to undertake a duty to defend OWNER and any of its officers, officials, employees, agents, or volunteers, as a separate duty, independent of and broader that the duty to indemnify. The duty to defend as herein agreed to by DESIGN-BUILDER expressly includes all costs of litigation, attorneys fees, settlement costs and expenses in connection with claims or litigation, whether or not the claims are valid, false or groundless, as long as the claims could be in any manner be causally connected to DESIGN-BUILDER as reasonably determined by OWNER.

Upon the tender by OWNER to DESIGN-BUILDER, DESIGN-BUILDER shall be bound and obligated to assume the defense of OWNER and any of its officers, officials, employees, agents, or volunteers, including the a duty to settle and otherwise pursue settlement negotiations, and shall pay, liquidate, discharge and satisfy any and all settlements, judgments, awards, or expenses resulting from or arising out of the claims without reimbursement from OWNER or any of its officers, officials, employees, agents, or volunteers.

It is further understood and agreed by DESIGN-BUILDER that if OWNER tenders a defense of a claim on behalf of OWNER or any of its officers, officials, employees, agents, or volunteers and DESIGN-BUILDER fails, refuses or neglects to assume the defense thereof, OWNER and its officers, officials, employees, agents, or volunteers may agree to compromise and settle or defend any such claim or action and DESIGN-BUILDER shall be bound and obligated to reimburse OWNER and its officers, officials, employees, agents, or volunteers for the amounts expended by each in defending or settling such claim, or in the amount required to pay any judgment rendered therein.

The defense and indemnity obligations set forth above shall be direct obligations and shall be separate from and shall not be limited in any manner by any insurance procured in accordance with the insurance requirements set forth in this Contract. In addition, such obligations remain in force regardless of whether OWNER provided approval for, or did not review or object to, any insurance DESIGN-BUILDER may have procured in a accordance with the insurance requirements set forth in this Contract. The defense and indemnity obligations shall arise at such time that any claim is made, or loss, injury or damage of any type has been incurred by OWNER, and the entry of judgment, arbitration, or litigation of any claim shall not be a condition precedent to these obligations.

The defense and indemnity obligations set forth in this section shall survive termination or expiration of this Contract.

If DESIGN-BUILDER should subcontract all or any portion of the work to be performed under this Contract, DESIGN-BUILDER shall require each subcontractor to Indemnify, hold harmless and defend OWNER and each of its officers, officials, employees, agents and volunteers in accordance with the terms as set forth above.

Article 8

Time

8.1 Obligation to Achieve the Contract Times.

8.1.1 Design-Builder agrees that it will commence performance of the Work and achieve the Contract Time(s) in accordance with Article 5 of the Agreement.

8.2 Delays to the Work.

8.2.1 If Design-Builder is delayed on the critical path in the performance of the Work due to acts, omissions, conditions, events, or circumstances beyond its control and due to no fault of its own or those for whom Design-Builder is responsible, the Contract Time(s) for performance shall be reasonably extended by Change Order. By way of example, events that will entitle Design-Builder to an extension of the Contract Time(s) include acts or omissions of Owner or anyone under Owner's control (including separate contractors), changes in the Work, Differing Site Conditions, Hazardous Conditions, and Force Majeure Events.

8.2.2 In addition to Design-Builder's right to a time extension for those events set forth in Section 8.2.1 above, Design-Builder shall also be entitled to an appropriate adjustment of the Contract Price provided, however, that the Contract Price shall not be adjusted for Force Majeure Events unless otherwise provided in the Agreement.

Article 9

Changes to the Contract Price and Time

9.1 Change Orders.

9.1.1 A Change Order is a written instrument issued after execution of the Agreement signed by Owner and Design-Builder, stating their agreement upon all of the following:

- **9.1.1.1** The scope of the change in the Work;
- **9.1.1.2** The amount of the adjustment to the Contract Price; and
- **9.1.1.3** The extent of the adjustment to the Contract Time(s).

9.1.2 All changes in the Work authorized by applicable Change Order shall be performed under the applicable conditions of the Contract Documents. Owner and Design-Builder shall negotiate in good faith and as expeditiously as possible the appropriate adjustments for such changes.

9.1.3 If Owner requests a proposal for a change in the Work from Design-Builder and subsequently elects not to proceed with the change, a Change Order shall be issued to reimburse Design-Builder for reasonable costs incurred for estimating services, design services and services involved in the preparation of proposed revisions to the Contract Documents.

9.1.4 Approval of any change order by the Owner is subject to Resolution No. 2017-158 as adopted by the City Council. The approval authority defined in Resolution No. 2017-158 shall be based on the current awarded contract price for each Phase as approved by City Council.

9.2 Work Change Directives.

9.2.1 A Work Change Directive is a written order prepared and signed by Owner directing a change in the Work prior to agreement on an adjustment in the Contract Price and/or the Contract Time(s).

9.2.2 Owner and Design-Builder shall negotiate in good faith and as expeditiously as possible the appropriate adjustments for the Work Change Directive. Upon reaching an agreement, the parties shall prepare and execute an appropriate Change Order reflecting the terms of the agreement.

9.3 Minor Changes in the Work.

9.3.1 Minor changes in the Work do not involve an adjustment in the Contract Price and/or Contract Time(s) and do not materially and adversely affect the Work, including the design, quality, performance and workmanship required by the Contract Documents. Design-Builder may make minor changes in the Work consistent with the intent of the Contract Documents, provided, however, that Design-Builder shall promptly inform Owner, in writing, of any such changes and record such changes on the documents maintained by Design-Builder.

9.4 Contract Price Adjustments.

9.4.1 The increase or decrease in Contract Price resulting from a change in the Work shall be determined by one or more of the following methods:

9.4.1.1 Unit prices set forth in the Agreement or as subsequently agreed to between the parties;

9.4.1.2 A mutually accepted lump sum, properly itemized and supported by sufficient substantiating data to permit evaluation by Owner;

9.4.1.3 Costs, fees and any other markups set forth in the Agreement; or

9.4.1.4 If an increase or decrease cannot be agreed to as set forth in items 9.4.1.1 through 9.4.1.3 above and Owner issues a Work Change Directive, the cost of the change of the Work shall be determined by the reasonable expense and savings in the performance of the Work resulting from the change, including a reasonable overhead and profit, as may be set forth in the Agreement.

9.4.2 If unit prices are set forth in the Contract Documents or are subsequently agreed to by the parties, but application of such unit prices will cause substantial inequity to Owner or Design-Builder because of differences in the character or quantity of such unit items as originally contemplated, such unit prices shall be equitably adjusted.

9.4.3 If Owner and Design-Builder disagree upon whether Design-Builder is entitled to be paid for any services required by Owner, or if there are any other disagreements over the scope of Work or proposed changes to the Work, Owner and Design-Builder shall resolve the disagreement pursuant to Article 10 hereof. As part of the negotiation process, Design-Builder shall furnish Owner with a good faith estimate of the costs to perform the disputed services in accordance with Owner's interpretations. If the parties are unable to agree and Owner expects Design-Builder to perform the services in accordance with Owner's interpretations, Design-Builder shall proceed to perform the disputed services, conditioned upon Owner issuing a written order to Design-Builder (i) directing Design-Builder to proceed; and (ii) specifying Owner's interpretation of the services that are to be performed. If this occurs, Design-Builder shall be entitled to submit in its Applications for Payment an amount equal to fifty percent (50%) of its reasonable estimated direct cost to perform the services, and Owner agrees to pay such amounts, with the express understanding that (i) such payment by Owner does not prejudice Owner's right to argue that it has no responsibility to pay for

such services; and (ii) receipt of such payment by Design-Builder does not prejudice Design-Builder's right to seek full payment of the disputed services if Owner's order is deemed to be a change to the Work.

9.5 Emergencies.

9.5.1 In any emergency affecting the safety of persons and/or property, Design-Builder shall act, at its discretion, to prevent threatened damage, injury or loss. Any change in the Contract Price and/or Contract Time(s) on account of emergency work shall be determined as provided in this Article 9.

Article 10

Contract Adjustments and Disputes

10.1 Requests for Contract Adjustments and Relief.

10.1.1 If either Design-Builder or Owner believes that it is entitled to relief against the other for any event arising out of or related to the Work or Project, such party shall provide written notice to the other party of the basis for its claim for relief. Such notice shall, if possible, be made prior to incurring any cost or expense and in accordance with any specific notice requirements contained in applicable sections of these General Conditions of Contract. In the absence of any specific notice requirement, written notice shall be given within a reasonable time, not to exceed twenty-one (21) days, after the occurrence giving rise to the claim for relief or after the claiming party reasonably should have recognized the event or condition giving rise to the request, whichever is later. The claimant shall provide more complete information with respect to the claim within fourteen (14) days of the initial notice. The more complete information shall include sufficient information to advise the other party of the circumstances giving rise to the claim for relief, the specific contractual adjustment or relief requested and the basis of such request.

10.2 Dispute Avoidance and Resolution.

10.2.1 The parties are fully committed to working with each other throughout the Project and agree to communicate regularly with each other at all times so as to avoid or minimize disputes or disagreements. If disputes or disagreements do arise, Design-Builder and Owner each commit to resolving such disputes or disagreements in an amicable, professional and expeditious manner so as to avoid unnecessary losses, delays and disruptions to the Work.

10.2.2 Claims for Additional Time.

Extension of time, when granted, will be based upon the effect of delays to the Project as a whole and will not be granted for noncontrolling delays to minor included portions of the Work unless it can be shown that such delays did, in fact, delay the progress of the Project as a whole. The Design-Builder shall not be entitled to damages or additional payment due to these delays except to the extent the delay exceeds the original Contract duration for Substantial Completion, and any extension hereunder other than any extension granted due to Owner caused delay, when Owner is responsible for the delay, the delay is unreasonable under the circumstances involved, not within the contemplation of the parties, and such delay causes actual damage to the Design-Builder. The Owner shall not be entitled to liquidated damages for Design-Builder delays unless the delay by Design-Builder exceeds the original Contract duration for Substantial Completion and any extension of time to which the Design-Builder is entitled to under the Specifications.

If delays are caused by unforeseen events beyond the control of the Design-Builder, such delays will entitle the Design-Builder to an extension of time as provided herein. War, governmental regulations, priorities, labor disputes, strikes, fires, floods, adverse weather necessitating cessation of Work, other similar action of the elements, inability to obtain materials, equipment or labor because of Federal Government restrictions arising out of the National Defense or War Program, required Extra Work, action or inaction by the Owner, or other specific reasons as may be further described in the Specifications may constitute such a delay.

If the Design-Builder is delayed by the failure of the Owner to furnish necessary rights of way or materials agreed to be furnished by it, or by failure to supply necessary plans or instructions concerning the Work, after written request therefore, the Design-Builder shall be entitled to an extension of time as provided herein.

10.2.3 Claims and Disputes.

10.2.3.1 General.

The Design-Builder and Owner shall make good faith efforts to resolve any and all Claims and disputes in a timely manner that may from time to time arise during Design-Builder's performance of the Work. Claims, including those alleging an error or omission shall be directed to the Owner's Construction Manager for action as provided in the "Resolution of Claims and Disputes," below.

It shall be a condition precedent to Claims review by the Public Works Director or his or her designated representative and to mediation or litigation between the Design-Builder and Owner as to all such matters arising prior to the date final payment is due, that a formal decision on all Design-Builder Claims be made by the Construction Manager. It shall be a condition precedent that the Design-Builder appeal any disputed Claim to the Public Works Director prior to initiating mediation or litigation. It shall be a condition precedent that the Design-Builder mediate any disputed Claim through non-binding mediation as provided herein, prior to initiating litigation. Unless mutually waived in writing by both parties, these provisions apply regardless of 1) whether such matters relate to execution and progress of the Work, or 2) the extent to which the Work has been completed.

Notice of Intent to Claim by Design-Builder must be made within 72 hours after occurrence of the event giving rise to such Claim, or within 72 hours after the claimant first discovers or should have reasonably discovered the condition giving rise to the Claim, whichever is later. Notice of Intent to Claim and Claims must be made by written notice.

At all times during the course of the dispute resolution process pursuant to the "Resolution of Claims and Disputes," the Design-Builder shall continue with the Work as directed, in a diligent manner and without delay, or shall conform to the Owner's decision or order, and shall be governed by all applicable provisions of the Contract. Records of the Work shall be kept in sufficient detail to enable payment in accordance with applicable provisions in the Contract if this should become necessary. The making of final payment shall not constitute a waiver of Claims by the Owner including, but not limited to, the following:

- (i) liens, Claims, security interests or encumbrances arising out of the Contract and unsettled;
- (ii) failure of the Work to comply with the requirements of the Contract Documents; or
- (iii) terms of special warranties required by the Contract Documents.

Design-Builder shall promptly provide an unconditional waiver and release upon final payment in accordance with California Civil Code section 3262 and these Contract Documents. Except to the extent of any Claim arising from City's sole or active negligence, and except to the extent Design-Builder expressly describes any other disputed Claims for which prior written notice has been given the City and lists the respective dollar amounts in an unconditional waiver and release, the making of final payment shall constitute a waiver of Claims by the Design-Builder pertaining to any and all costs, expenses, changes or other Claims related to Contract Price or Contract Time, including any synergistic effects attributed to multiple Change Orders. In the event of any disputed Claims between the City and Design-Builder, the City may withhold from the final payment an amount not to exceed 150% of the disputed amount.

10.2.3.2 Resolution of Claims and Disputes.

(1) Decision of Construction Manager:

a. If the Design-Builder believes any Work or demand to be outside the requirements of the Contract or believes that omissions, conflicts, errors, or discrepancies will cause or have caused the Design-Builder additional costs or delays in the performance of the Work, he/she shall file a written Notice of Intent to Claim with the Construction Manager within 72 hours after occurrence of the event giving rise to the Claim, or within 72 hours after the Design-Builder or its subcontractor first discovers or should have reasonably discovered the condition giving rise to the Claim, whichever is later. If a written Notice of Intent to Claim is not submitted within this time period, the Design-Builder shall waive his or her right to further Claims on the issue and any synergistic effects related to such Claim.

1. Within 10 working days following the Notice of Intent to Claim, the Design-Builder shall provide a Notice of Claim with complete supporting data for the Claim of the cost and delay related to such omissions, conflicts, errors, discrepancies, or Work alleged to be outside the requirements of the Contract. Notwithstanding the foregoing, if all supporting data cannot reasonably be made available within said 10 working days, then Design-Builder shall provide all then available supporting data along with a request for additional time, stating a time certain, to obtain the remainder of supporting data along with both an explanation of the nature of such supporting data and the reason why additional time is necessary to provide same to the Construction Manager.

2. If a written Notice of Claim, along with complete supporting data or all then available supporting data and reasonable request for additional time with explanation as required above, is not submitted within 10 working days following the Notice of Intent to Claim, the Design-Builder shall waive his or her right to make further Claims on the issue and any synergistic effects related to such Claim.

3. The Design-Builder's request for additional time to provide the remainder of its supporting data shall be deemed acceptable to Owner unless the Construction Manager rejects in writing Design-Builder's request within 5 working days from receipt of Design-Builder's request. If the Construction Manager rejects Design-Builder's request for additional time, then Design-Builder shall either provide complete supporting data immediately upon receipt of such rejection or within any time acceptable to the Construction Manager as stated in his or her written rejection, whichever is later (unless otherwise mutually agreed upon in writing by Design-Builder and Construction Manager).

b. The Construction Manager will review any and all Claims and take one or more of the following preliminary actions in writing within 10 working days of receipt of written Notice of Claim and complete supporting data: 1) request additional supporting data from the Design-Builder; 2) reject the Claim in whole, or in part, stating reasons for rejection; or, 3) recommend approval of the Claim. In the event the Construction Manager has not taken any preliminary action within 10 working days, then the Claim is deemed rejected unless the Design-Builder and the Construction Manager mutually agree in writing to extend the time period for taking preliminary action. The Construction Manager will make his or her decision on the basis of the pertinent Contract provisions, together with the facts and circumstances involved in the dispute. The Construction Manager may also, but is not obligated to, notify the surety on Design-Builder's performance bond of the nature and amount of the Claim.

c. If the Construction Manager requests additional supporting data from the Design-Builder, the Design-Builder shall supply the additional information to the Construction Manager within 10 working days unless the Design-Builder and the Construction Manager mutually agree in writing to extend the time period for supplying such information. The Construction Manager will have 10 working days from the receipt of additional supporting data to provide a written decision unless the Design-Builder and the Construction Manager mutually agree in writing to extend the time period for providing such decision. In the event the Construction Manager has not provided a written decision within 10 working days, or any extended time period agreed to in writing by Construction Manager and Design-Builder, then the Claim is deemed rejected and this shall constitute Design-Builder's automatic request for an appeal meeting with the Public Works Director unless the Design-Builder submits a written withdrawal of its Claim.

d. If the Claim is rejected in whole or in part by a written decision of the Construction Manager, the Design-Builder shall notify the Construction Manager in writing within 10 working days after receiving the written decision that either: 1) the decision is accepted, and the Claim is amended; accordingly, or 2) the Design-Builder requests an appeal meeting with the Public Works Director. Failure to timely request an appeal meeting with the Public Works Director following receipt of the written decision constitutes acceptance by the Design-Builder of the Construction Manager's decision. If the Owner and Design-Builder are able to resolve their dispute, the Owner will promptly process any required Contract changes.

(2) Decision of Public Works Director

a. The Public Works Director, or his or her designee, shall meet with the Design-Builder and the Construction Manager within 15 working days (unless necessary to accommodate the Public Works Director's schedule, or that of his or her designee, but in no event longer than 20 working days; or unless otherwise mutually agreed upon in writing by Design-Builder and Construction Manager) from the Design-Builder's timely submittal of his or her request, or any automatic request hereunder, for a meeting. The Design-Builder may make a presentation in support of his or her Claim. No attorney may take part in the presentation or defense of the Claim in the meeting with the Public Works Director, or his or her designee. Nothing herein shall prevent an attorney from providing advice to a party either before or after the meeting. In the event the meeting with the Public Works Director, or his or her designee, has not been conducted within the time provided herein or as agreed upon in writing by Design-Builder and Construction Manager, then the Claim is deemed rejected and, unless the Design-Builder submits a written withdrawal of its Claim, the parties shall proceed to mediation as provided herein.

b. Within 10 working days (unless otherwise mutually agreed upon in writing by Design-Builder and Construction Manager) from the meeting with the Design-Builder and the Construction Manager, the Public Works Director, or his or her designee, shall render a written decision and a copy thereof shall be personally delivered, or mailed return receipt requested, to the Design-Builder. In the event the Public Works Director, or his or her designee, has not provided a written decision within 10 working days, or any extended time period agreed to in writing by Construction Manager and Design-Builder, then the Claim is deemed rejected and, unless the Design-Builder submits a written withdrawal of its Claim, the parties shall proceed to mediation as provided herein.

c. If the Claim is rejected in whole or in part by a written decision of the Public Works Director, or his or her designee, the Design-Builder shall notify the Construction Manager in writing within 10 working days after receiving the written decision that either: 1) the decision is accepted and the Claim is amended accordingly; or 2) the Design-Builder rejects the decision of the Public Works Director, or his or her designee. Failure to timely notify the Construction Manager of either following receipt of the Public Works Director's written decision, or that of his or her designee, shall constitute acceptance by the Design-Builder of the Public Works Director's decision, or that of his or her designee. If the Owner and Design-Builder are able to resolve their dispute, the Owner will promptly process any required Contract changes. If the Design-Builder rejects the written decision of the Public Works Director or his or her designee, the parties shall proceed to mediation as provided herein.

(3) Mediation

a. In the event that the Claim is not resolved after exhausting all aforementioned administrative measures, then the Design-Builder must participate in non-binding mediation with City before the Design-Builder may initiate litigation.

b. The parties shall mutually select, in writing, a mediator with at least 5 years' experience in the construction industry. In the event that the parties are unable to agree on a mediator within 15 working days of Design-Builder's rejection of the decision of the Public Works Director or his or her designee, the City may select the mediator. Mediation, including at least one session requiring physical attendance by all parties, shall begin within 15 working days of selection of the mediator, unless necessary to accommodate the mediator's schedule. The parties shall share the mediator's fees and any administrative costs of mediation equally. The mediation shall be held in Fresno, California, unless another location is mutually agreed upon by the parties in writing. In the event the parties are unable to reach a mutually acceptable resolution of the Claim within 20 working days of the start of mediation, unless extended or otherwise terminated by written mutual agreement of the parties, mediation shall terminate.

c. If the Owner and Design-Builder are able to resolve their dispute the Owner will promptly process any required Contract changes. Any settlement reached in principle must be in writing and is subject to approval by the City Manager or City Council consistent with City laws and policies. Should the dispute remain unresolved, the parties may resort to other dispute resolution procedures.

d. All statements made during the mediation shall be confidential and subject to sections 703.5, 1119 and 1152 of the California Evidence Code.

(4) Government Claims Act. Nothing herein is intended by the parties to waive any requirements of the Design-Builder to comply with the Government Claims Act including, without limitation, California Government Code section 905; and Design-Builder agrees that it shall remain responsible for complying with said section regarding any Claim. The parties agree, however, that the timeline for the Design-Builder to file a claim under the Government Claims Act is tolled until exhaustion of the Design-Builder of its administrative remedies hereunder (i.e., either upon termination of mediation, or upon written mutual waiver of mediation by the parties, whichever first occurs).

(5) Litigation

a. If the Design-Builder continues to dispute the Work demanded of him/her after exhausting all aforementioned administrative measures, the Design-Builder may institute legal proceedings, but only after final acceptance of the Project by the Owner.

Unless specifically waived by the Owner, in writing, the submission of a dispute for mediation in accordance with the above provisions shall be a condition precedent to the Design-Builder's right to initiate a suit, action or other proceeding against the Owner for damages.

b. In the event Owner initiates suit, action, or other proceeding against the Design-Builder for damages, the prevailing party in such suit or action shall be entitled to recover reasonable attorney's fees and costs of suit.

c. In the event Design-Builder initiates suit, action, or other proceeding against the Owner, the Owner shall be entitled to recover reasonable attorneys' fees and engineering defense costs if the Design-Builder is not awarded, by the arbitrator or court, a dollar amount greater than 50 percent of the Design-Builder's original Claim for damages.

d. The Design-Builder shall include, or cause to be included, a requirement in all subcontracts of all tiers of Subcontractors for this Project that whenever the Subcontractor disputes the Work demanded of him/her, he/she shall cooperate and comply with the Claims and Dispute procedures contained herein including, without limitation, exhausting all administrative measures prior to instituting legal proceedings, and instituting legal proceedings only after final acceptance of the Project by the Owner.

10.3 Mediator

The Owner shall be responsible for fifty percent (50%) of the Mediator costs. If needed, the owner's share of the cost shall be added through a contract change order. The dollar amount listed in the bid item is an estimate only and will be included in each Bidder's Proposal. Invoices of the Mediator shall be paid by the Design-Builder only upon direct written authorization from the Owner.

Final payment to Design-Builder will be based on fifty percent (50%) of the total amount of Owner approved invoices of the Mediator actually billed to Owner by Design-Builder. The Design-Builder shall include the specified lump sum bid item on the Bid Proposal for payment of Owner's share of costs for the Mediator. Payment will be made under this bid item by issuance of a Change Order approved by the Owner and charged against this lump sum allocation.

This bid item may be increased, decreased, or deleted in its entirety and is not to be construed as additional money owed to the Design-Builder. If no Change Order is issued against this bid item, the Contract Price shall be reduced by the full amount of the bid item included in the Bid Proposal for the Mediator.

The Design-Builder shall have no claim for anticipated overhead or profit should the Owner fail to issue any Change Orders against this bid item.

10.4 Duty to Continue Performance.

10..1 Unless provided to the contrary in the Contract Documents, Design-Builder shall continue to perform the Work and Owner shall continue to satisfy its payment obligations for undisputed amounts to Design-Builder as well as any further amounts pursuant to Section 9.4.3, pending the final resolution of any dispute or disagreement between Design-Builder and Owner.

10.5 CONSEQUENTIAL DAMAGES.

10.5.1 NOTWITHSTANDING ANYTHING HEREIN TO THE CONTRARY (EXCEPT AS SET

FORTH IN SECTION 10.5.2 BELOW), NEITHER DESIGN-BUILDER NOR OWNER SHALL BE LIABLE TO THE OTHER FOR ANY CONSEQUENTIAL LOSSES OR DAMAGES, WHETHER ARISING IN CONTRACT, WARRANTY, TORT (INCLUDING NEGLIGENCE), STRICT LIABILITY OR OTHERWISE, INCLUDING BUT NOT LIMITED TO LOSSES OF USE, PROFITS, BUSINESS, REPUTATION OR FINANCING.

10.7.2 The consequential damages limitation set forth in Section 10.5.1 above is not intended to affect the payment of liquidated damages or lost early completion bonus, if any, set forth in Article 5 of the Agreement, which both parties recognize has been established, in part, to reimburse Owner or reward Design-Builder for some damages that might otherwise be deemed to be consequential.

Article 11

Stop Work and Termination

11.1 Owner's Right to Stop Work.

11.1.1 Owner may, without cause and for its convenience, order Design-Builder in writing to stop and suspend the Work.

11.1.2 Design-Builder is entitled to seek an adjustment of the Contract Price and/or Contract Time(s) if its cost or time to perform the Work has been adversely impacted by any suspension or stoppage of the Work by Owner.

11.2 Owner's Right to Perform and Terminate for Cause.

11.2.1 If Design-Builder persistently fails to (i) provide a sufficient number of skilled workers, (ii) supply the materials required by the Contract Documents, (iii) comply with applicable Legal Requirements, (iv) timely pay, without cause, Design Consultants or Subcontractors, (v) prosecute the Work with promptness and diligence to ensure that the Work is completed by the Contract Time(s), as such times may be adjusted, or (vi) perform material obligations under the Contract Documents, then Owner, in addition to any other rights and remedies provided in the Contract Documents or by law, shall have the rights set forth in Sections 11.2.2 and 11.2.3 below.

11.2.2 Upon the occurrence of an event set forth in Section 11.2.1 above, Owner may provide written notice to Design-Builder that it intends to terminate the Agreement unless the problem cited is cured, or commenced to be cured, within seven (7) days of Design-Builder's receipt of such notice. If Design-Builder fails to cure, or reasonably commence to cure, such problem, then Owner may give a second written notice to Design-Builder of its intent to terminate within an additional seven (7) day period. If Design-Builder, within such second seven (7) day period, fails to cure, or reasonably commence to cure, such problem, then Owner may declare the Agreement terminated for default by providing written notice to Design-Builder of such declaration.

11.2.3 Upon declaring the Agreement terminated pursuant to Section 11.2.2 above, Owner may enter upon the premises and take possession, for the purpose of completing the Work, of all materials, equipment, scaffolds, tools, appliances and other items thereon, which have been purchased or provided for the performance of the Work, all of which Design-Builder hereby transfers, assigns and sets over to Owner for such purpose, and to employ any person or persons to complete the Work and provide all of the required labor, services, materials, equipment and other items. In the event of such termination, Design-Builder shall not be entitled to receive any further payments under the Contract Documents until the Work shall be finally completed in accordance with the Contract Documents. At such time, if the unpaid balance of the Contract Price exceeds the cost and expense incurred by Owner in completing the Work, such excess shall be paid by Owner to Design-Builder. Notwithstanding the preceding sentence, if the Agreement establishes a

Guaranteed Maximum Price, Design-Builder will only be entitled to be paid for Work performed prior to its default. If Owner's cost and expense of completing the Work exceeds the unpaid balance of the Contract Price, then Design-Builder shall be obligated to pay the difference to Owner. Such costs and expense shall include not only the cost of completing the Work, but also losses, damages, costs and expense, including attorneys' fees and expenses, incurred by Owner in connection with the reprocurement and defense of claims arising from Design-Builder's default, subject to the waiver of consequential damages set forth in Section 10.5 hereof.

11.3 Termination for Convenience.

11.3.1 Upon ten (10) days' written notice to Design-Builder, Owner may, for its convenience and without cause, elect to terminate this Agreement. In such event, Owner shall pay Design-Builder for the following:

11.3.1.1 All Work executed and for proven loss, cost or expense in connection with the Work;

11.3.1.2 The reasonable costs and expenses attributable to such termination, including demobilization costs and amounts due in settlement of terminated contracts with Subcontractors and Design Consultants; and

11.3.1.3 The amount set forth in Article 8 of the Agreement.

11.3.2 If Owner terminates this Agreement pursuant to Section 11.6.1 above and proceeds to design and construct the Project through its employees, agents or third parties, Owner's rights to use the Work Product shall be as set forth in Section 4.3 of the Agreement. Such rights may not be transferred or assigned to others without Design-Builder's express written consent and such third parties' agreement to the terms of Article 4 of the Agreement.

Article 12

Electronic Data

12.1 Electronic Data.

12.1.1 The parties recognize that Contract Documents, including drawings, specifications and three-dimensional modeling (such as Building Information Models) and other Work Product may be transmitted among Owner, Design-Builder and others in electronic media as an alternative to paper hard copies (collectively "Electronic Data").

12.2 Transmission of Electronic Data.

12.2.1 Owner and Design-Builder shall agree upon the software and the format for the transmission of Electronic Data. Each party shall be responsible for securing the legal rights to access the agreed-upon format, including, if necessary, obtaining appropriately licensed copies of the applicable software or electronic program to display, interpret and/or generate the Electronic Data.

12.2.2 Neither party makes any representations or warranties to the other with respect to the functionality of the software or computer program associated with the electronic transmission of Work Product. Unless specifically set forth in the Agreement, ownership of the Electronic Data does not include ownership of the software or computer program with which it is associated, transmitted, generated or interpreted.

12.2.3 By transmitting Work Product in electronic form, the transmitting party does not transfer or assign its rights in the Work Product. The rights in the Electronic Data shall be as set forth in Article 4 of the Agreement. Under no circumstances shall the transfer of ownership of Electronic Data be deemed to be a sale by the transmitting party of tangible goods.

12.3 Electronic Data Protocol.

12.3.1 The parties acknowledge that Electronic Data may be altered or corrupted, intentionally or otherwise, due to occurrences beyond their reasonable control or knowledge, including but not limited to compatibility issues with user software, manipulation by the recipient, errors in transcription or transmission, machine error, environmental factors, and operator error. Consequently, the parties understand that there is some level of increased risk in the use of Electronic Data for the communication of design and construction information and, in consideration of this, agree, and shall require their independent contractors, Subcontractors and Design Consultants to agree, to the following protocols, terms and conditions set forth in this Section 12.3.

12.3.2 Electronic Data will be transmitted in the format agreed upon in Section 12.2.1 above, including file conventions and document properties, unless prior arrangements are made in advance in writing.

12.3.3 The Electronic Data represents the information at a particular point in time and is subject to change. Therefore, the parties shall agree upon protocols for notification by the author to the recipient of any changes which may thereafter be made to the Electronic Data, which protocol shall also address the duty, if any, to update such information, data or other information contained in the electronic media if such information changes prior to Final Completion of the Project.

12.3.4 The transmitting party specifically disclaims all warranties, expressed or implied, including, but not limited to, implied warranties of merchantability and fitness for a particular purpose, with respect to the media transmitting the Electronic Data. However, transmission of the Electronic Data via electronic means shall not invalidate or negate any duties pursuant to the applicable standard of care with respect to the creation of the Electronic Data, unless such data is materially changed or altered after it is transmitted to the receiving party, and the transmitting party did not participate in such change or alteration.

Article 13

Miscellaneous

13.1 Confidential Information.

13.1.1 Confidential Information is defined as information which is determined by the transmitting party to be of a confidential or proprietary nature and: (i) the transmitting party identifies as either confidential or proprietary; (ii) the transmitting party takes steps to maintain the confidential or proprietary nature of the information; and (iii) the document is not otherwise available in or considered to be in the public domain. The receiving party agrees to maintain the confidentiality of the Confidential Information and agrees to use the Confidential Information solely in connection with the Project.

13.2 Assignment.

13.2.1 Neither Design-Builder nor Owner shall, without the written consent of the other assign, transfer or sublet any portion or part of the Work or the obligations required by the Contract Documents.

13.3 Successorship.

13.3.1 Design-Builder and Owner intend that the provisions of the Contract Documents are binding upon the parties, their employees, agents, heirs, successors and assigns.

13.4 Governing Law.

13.4.1 The Agreement and all Contract Documents shall be governed by the laws of the location of the Project, without giving effect to its conflict of law principles.

13.5 Severability.

13.5.1 If any provision or any part of a provision of the Contract Documents shall be finally determined to be superseded, invalid, illegal, or otherwise unenforceable pursuant to any applicable Legal Requirements, such determination shall not impair or otherwise affect the validity, legality, or enforceability of the remaining provision or parts of the provision of the Contract Documents, which shall remain in full force and effect as if the unenforceable provision or part were deleted.

13.6 No Waiver.

13.6.1 The failure of either Design-Builder or Owner to insist, in any one or more instances, on the performance of any of the obligations required by the other under the Contract Documents shall not be construed as a waiver or relinquishment of such obligation or right with respect to future performance.

13.7 Headings.

13.7.1 The headings used in these General Conditions of Contract, or any other Contract Document, are for ease of reference only and shall not in any way be construed to limit or alter the meaning of any provision.

13.8 Notice.

13.8.1 Whenever the Contract Documents require that notice be provided to the other party, notice will be deemed to have been validly given (i) if delivered in person to the individual intended to receive such notice; (ii) four (4) days after being sent by registered or certified mail, postage prepaid to the address indicated in the Agreement; (iii) if transmitted by facsimile, by the time stated in a machine-generated confirmation that notice was received at the facsimile number of the intended recipient; or (iv) by electronic mail, by the time frame stated in the email-generated confirmation that notice was received at the facsimile number of the intended recipient; or (iv) by the email of the intended recipient.

13.9 Amendments.

13.9.1 The Contract Documents may not be changed, altered, or amended in any way except in writing signed by a duly authorized representative of each party.

13.10 Attorney's Fees.

13.10.1 If either party is required to commence any proceeding or legal action to enforce or interpret any term, covenant of the Contract Documents, the prevailing party in such proceeding shall be entitled to recover from the other party its reasonable attorney's fees and legal expenses.

13.11 Interpretation.

13.11.1 The parties acknowledge that this Contract in its final form is the result of the combined efforts of the parties and that, should any provision of this Contract be found to be

ambiguous in any way, such ambiguity shall not be resolved by construing this Contract in favor of or against either party, but rather by construing the terms in accordance with their generally accepted meaning The license for use of this document expires 1 year from the date of purchase. To renew your license, visit <u>store.dbia.org</u>.

Questions? We're here to help.

Contact us



Design-Build Institute of America 1001 Pennsylvania Ave. NW, Suite 410 Washington, DC 20004

> (202) 682-0110 dbia@dbia.org